Edith Cowan University Research Online

Theses: Doctorates and Masters

Theses

2021

The Coaches' Eye: Exploring coach decision-making during talent identification

Alexandra Hannah Roberts Edith Cowan University

Follow this and additional works at: https://ro.ecu.edu.au/theses

Part of the Sports Sciences Commons, and the Sports Studies Commons

Recommended Citation

Roberts, A. H. (2021). *The Coaches' Eye: Exploring coach decision-making during talent identification*. https://ro.ecu.edu.au/theses/2391

This Thesis is posted at Research Online. https://ro.ecu.edu.au/theses/2391

Edith Cowan University

Copyright Warning

You may print or download ONE copy of this document for the purpose of your own research or study.

The University does not authorize you to copy, communicate or otherwise make available electronically to any other person any copyright material contained on this site.

You are reminded of the following:

- Copyright owners are entitled to take legal action against persons who infringe their copyright.
- A reproduction of material that is protected by copyright may be a copyright infringement. Where the reproduction of such material is done without attribution of authorship, with false attribution of authorship or the authorship is treated in a derogatory manner, this may be a breach of the author's moral rights contained in Part IX of the Copyright Act 1968 (Cth).
- Courts have the power to impose a wide range of civil and criminal sanctions for infringement of copyright, infringement of moral rights and other offences under the Copyright Act 1968 (Cth).
 Higher penalties may apply, and higher damages may be awarded, for offences and infringements involving the conversion of material into digital or electronic form.

The Coaches' Eye:

Exploring coach decision-making during talent identification

This thesis is presented for the degree of

Doctor of Philosophy

Alexandra Hannah Roberts



Edith Cowan University

School of Medical and Health Sciences

2021

THIS PAGE INTENTIONALLY LEFT BLANK

Abstract

For decades, researchers and practitioners interested in talent identification have discussed the *coaches' eye*: the elusive ability that allows some coaches to 'see' qualities in an athlete that point to their talent or future potential. While there is significant anecdotal evidence of coaches who possess this ability, there is little empirical research supporting the validity or reliability of the *coaches' eye*. Guided by ecological dynamics, this thesis employs mixed methodologies to explore the decision-making that underpins how high-level coaches identify talent in Olympic combat sports. These four studies captured the processes of thirtyfour coaches during the talent identification process, exploring and identifying the factors that impact on a coach's ability to perform this integral task.

A systematic review and meta-synthesis revealed that 'instinct' is a primary contributor to coach decision-making during talent identification (TID), allowing coaches to 'know it when they see it'. Semi-structured interviews with international coaches explored this 'instinct' during TID and revealed that coaches require experience, time and knowledge of context in order to identify talent. An instrumental case study corroborated these results, and also found that there is a significant conceptual difference between talent *identification* and talent *selection*, in the eyes of this coach. Both studies indicated that coaches likely select athletes based on their capabilities as a coach, not purely on athlete ability or potential. The final study found that nine national-level coaches did not agree on the rankings of talented youth judo athletes after four days. This finding indicates that the *coaches' eye* is subjective and confirms the novel findings of the prior studies; namely that coaches require time to get to know athletes, their opinions of the athletes' talent changed over time, and coaches vary in who they 'see' as talented. Finally, two new models are presented: the *Coach-Informed Talent Identification Process* and a novel model of the *Coaches' Eye in Talent Identification*.

The experiential coach knowledge gathered in this thesis informed the creation of these models.

This thesis indicates that the *coaches' eye* is the lens through which coaches view athletes, using their expertise and experience to interpret the athlete's raw potential, and the time spent with the athlete and the context of their identification to determine whom they will select into their team. It appears that coaches perceive talent with reference to what they can develop in an athlete; thus, coaches must be involved in the identification and selection of talented athletes. These results indicate that National Sporting Organisations should ensure that coaches are provided with the necessary time, education and guidance to ensure that athlete outcomes are optimised.

This thesis provides an understanding of how the *coaches' eye* works during TID and a new understanding of this term. These findings have implications for the ongoing practice and research of talent identification in combat sports, and this work contains recommendations for both coaches and national sporting organisations to improve the confidence, accuracy and reliability of the *coaches'* eye when forecasting talent.

ii

Signed Declaration

I certify that this thesis does not, to the best of my knowledge and belief:

- i) Incorporate without acknowledgement any material previously submitted for a degree or diploma in any institution of higher education
- ii) Contain any material previously published or written by another person except where due reference is made in the text of this thesis; or
- iii) Contain any defamatory material

Signed:

THIS PAGE INTENTIONALLY LEFT BLANK

Acknowledgements

There are many people to whom I wish to express my gratitude and appreciation for their enduring and unwavering support throughout my PhD journey. I've had the great privilege to work with world-leading coaches, practitioners, academics and everyone in between and I can say with certainty that without every one of you, I would not be where I am now.

Firstly, to the four people who took a chance on me and accepted me into this PhD, my supervisors – Annette Raynor, Daniel Greenwood, Clare Humberstone and Fiona Iredale. Annette, I don't know where to start. Your guidance and patience throughout this process have been invaluable. Despite the time differences you have always made yourself available for whatever I needed, whether it was a kick, career advice, or re-writing a certain set of results for the umpteenth time. Thank you for showing me how to deal with the ups and downs of academia, and for reviewing innumerable grant and job applications - looks like it's all finally paying off! Daniel, you are responsible for me being the skill acquisition practitioner and coach educator that I am today. The experience of working with you has taught me much more than I could have ever hoped to learn in any other setting. Thank you for being there every step of the way, from picking me up from the airport when I first arrived back in Australia with my five suitcases, to teaching me skill acquisition from scratch, to continuing to support me personally and professionally from afar. To think that four years ago I hadn't even heard of 'Skill Acq'! Clare, thank you for your time and guidance throughout my time at the AIS and particularly in the Combat Centre. The support, networking, connections and opportunities that you made available to me are a big part of what has allowed me to be successful in completing this research and, more importantly, provided me with the practical, hands-on experiences that I could not have received in any other environment. Thank you for convincing the rest of the team to take a chance on me and my 'suspicious' CV.

Fi, thank you for your support throughout this project. Your insider knowledge and understanding of the combat sports was invaluable in the planning stages of this work.

To **Mandy Stanley**, my unofficial supervisor. Thank you for stepping in and taking on a novice qualitative researcher and showing me the ropes. Your input has dramatically shaped both my PhD and my research interests. I can't adequately express my gratitude for your guidance throughout this process and the time and effort you have put into my work – the work of a student who was not your responsibility. Thank you.

To the Australian and international **coaches** who participated in, guided and supported my research. Thank you for sharing your time and expertise, both in an 'official' research capacity as well as supporting my development as a practitioner.

To the **AIS crew** – Scholars, Movement Science, Combat Centre, coaches, athletes, support staff, technicians and everyone else in between. I will never forget my first few weeks at the AIS, thrown straight into the ultra-competitive MS Olympics (go Tapirs!) followed shortly by 'team-building' paintball with the Combat Centre. Thanks for the baking, the scoring of the baking, the cricket matches (and training), the raft building, the lunchtime quizzes, the table-tennis staff meetings and listening to countless presentations, workshopping ideas and proof-reading. While I am deliberately avoiding naming names for fear of forgetting anyone, there is one group in particular who deserve a shout out – the Skill Acq team. **Derek, Dan, Mike, Steve, Amy, Courtney** and **Georgia** – thank you for making me one of the team immediately, for journal clubs and writing groups, for team lunches and for being there every step of the way.

Britt –Thank you for hospital visits, Pokémon adventures, butter halloumi, teaching me how to push buttons on VICON, and for motivating me through these last few months (years). Thank you for being with me each step of the way and putting up with my distracting office habits. **Jeremy** –Thank you for always being available for a phone call, any time of day or night. Thank you for your unwavering belief in my ability, for buying a house with a

bedroom for me, and for reminding me that this will all be worth it. **G and CP** – thank you for being you. Words cannot express how much I owe to both of you for getting me to this point. Now it's your turn.

Finally, thank you to my family. This thesis and my achievements to date are a direct result of the opportunities and support you have given me throughout my whole life. **Mum**, thank you for showing me that being passionate about what you do is more important than the hours; and a begrudging thanks for the work ethic that you instilled in me. Thank you for making sure that I always had the opportunity to do what I needed to do to reach my goals, even when it meant moving to the other side of the world. **Dad**, thank you for the early mornings, late evenings and weekends away in support of my rowing, hockey, water polo or whatever other activity I was doing at the time. Sorry that it took moving to Melbourne to gain an appreciation for footy. **Kestin** and **Declan**, thanks for keeping me grounded and reminding me that I'm not all that just because I've (nearly) got a PhD.

THIS PAGE INTENTIONALLY LEFT BLANK

Abstract	i
Signed Declaration	iii
Acknowledgements	v
Table of Contents	ix
List of Tables	xiii
List of Figures	xiv
List of Publications and Conferences	xv
Statement of Author Contributions to Publications	xvii
Abbreviated Terminology	xviii
CHAPTER ONE Introduction	1
Background of Talent Identification	1
Statement of the Problem: The Coach as a Critical Decision-Maker	7
Context of this Thesis	8
Myself as the Researcher	
Thesis Overview	
CHAPTER TWO Literature Review	
Foreword	17
Theoretical Framework: Ecological Dynamics	17
Ecological Psychology	19
Dynamical Systems Theory	20
Decision-Making	22
The Decision-Making Continuum	22
Coach Decision-Making	24
An Ecological Approach to Decision-Making	25
Affordances	29
Talent Identification: A Research Summary	
Talent Identification in Combat Sports	32
Limitations in Talent Identification	
Latest Work in Talent Identification	34
The Coaches' Eye in Talent Identification	35
Expertise and Experiential Knowledge	
Expertise in Coaching	40
Summary	41

Table of Contents

CHAPTER THREE Expert Knowledge in Talent Identification Meta-Synthesis	A Systematic Review and
Foreword	
Abstract	Error! Bookmark not defined.
Introduction	Error! Bookmark not defined.
Methods	Error! Bookmark not defined.
Results	Error! Bookmark not defined.
Coach Instinct – "The Coaches' Eye"	Error! Bookmark not defined.
Drive and Ambition	Error! Bookmark not defined.
Physical and Technical Skills	Error! Bookmark not defined.
Game Intelligence	Error! Bookmark not defined.
Discussion	Error! Bookmark not defined.
Conclusion	Error! Bookmark not defined.
Publication Update	44
CHAPTER FOUR Understanding 'Gut Instinct'	
Foreword	46
Abstract	Error! Bookmark not defined.
Introduction	Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined.	ive Error! Bookmark not
Intuition and Decision-Making – An Ecological Perspect defined. Methods	iveError! Bookmark not
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants	iveError! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection	ive Error! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis	<i>ive</i> Error! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined. Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings	ive Error! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined. Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes	ive Error! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined. Error! Bookmark not defined. Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'?	ive Error! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'? Themes	ive Error! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'? Themes Discussion	iveError! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'? Themes Discussion The Coach as a Performer	iveError! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'? Themes Discussion The Coach as a Performer Appropriate Talent Identification	iveError! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'? Themes Discussion The Coach as a Performer Appropriate Talent Identification Conclusions	iveError! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'? Themes Discussion The Coach as a Performer Appropriate Talent Identification Conclusions CHAPTER FIVE A Case Study in Talent Identification	iveError! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'? Themes Discussion The Coach as a Performer Appropriate Talent Identification Conclusions CHAPTER FIVE A Case Study in Talent Identification	iveError! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined.
Intuition and Decision-Making – An Ecological Perspect defined. Methods Participants Data Collection Data Analysis Findings Participant Attributes What is 'Talent'? Themes Discussion The Coach as a Performer Appropriate Talent Identification Conclusions CHAPTER FIVE A Case Study in Talent Identification Foreword Abstract	iveError! Bookmark not Error! Bookmark not defined. Error! Bookmark not defined. 47 47

Coach Decision-Making	Error! Bookmark not defined.
Methodology and Methods	Error! Bookmark not defined.
Design and Epistemological Approach	Error! Bookmark not defined.
Data Collection	Error! Bookmark not defined.
Data Analysis	Error! Bookmark not defined.
Findings	Error! Bookmark not defined.
Phases and Themes	Error! Bookmark not defined.
Discussion	Error! Bookmark not defined.
Coach's Definition of Talent	Error! Bookmark not defined.
Identification vs Selection	Error! Bookmark not defined.
Conclusions and Practical Implications	Error! Bookmark not defined.
CHAPTER SIX Reliability of Coach Ratings over a Four-Day Tra	ining Camp49
Foreword	49
Abstract	Error! Bookmark not defined.
Introduction	Error! Bookmark not defined.
Methods	Error! Bookmark not defined.
Participants	Error! Bookmark not defined.
Procedure	Error! Bookmark not defined.
Ethics	Error! Bookmark not defined.
Data Analysis	Error! Bookmark not defined.
Results	Error! Bookmark not defined.
Inter-Coach Reliability	Error! Bookmark not defined.
Sensitivity of Coach Judgments	Error! Bookmark not defined.
Discussion	Error! Bookmark not defined.
Future Directions	Error! Bookmark not defined.
Study Limitations	Error! Bookmark not defined.
Conclusion	Error! Bookmark not defined.
CHAPTER SEVEN Discussion: Perceiving Talent and its Identifi Eye	cation Through the Coaches' 50
A Model of Talent Identification in High-Performance Coml	bat Sports51
Talent, Identification, Judgment, and Decision-Making	55
Forecasting and Confirmation	57
Selection	59
Development, Participation and Detection	60
Summary	62
The Coaches' Eye during Talent Identification	63

Coach-Athlete Dyad66
Forecasting, Selection and Attunement67
Practical Implications
Practical Implications for National Sporting Organisations
Practical Implications for Coaches and Coach Educators71
CHAPTER EIGHT Summary and Future Directions72
Summary of Findings73
Limitations of this study77
Future Directions
Conclusion79
References
Appendices114
Appendix A115
Appendix B120
Appendix C
Appendix D126
Appendix E127
Appendix F132
Appendix G133
Appendix H134
Appendix I135
Appendix J

List of Tables

Table 1: Operational Definitions of Thesis Terminology	Error! Bookmark not defined.
Table 2: Traditional Approaches to Identifying Talent	
Table 3: Results of Critical Appraisal	
Table 4: Coach Characteristics	Error! Bookmark not defined.
Table 5: Summary of Data Collection	Error! Bookmark not defined.
Table 6: Participant Details	Error! Bookmark not defined.
Table 7: Number of Coaches Rating >50% of Athletes at Each	Time Point Error! Bookmark
not defined.	

List of Figures

Figure 1: Key Stages in Talent Identification and Development	3
Figure 2: Flow Diagram of Studies	13
Figure 3: Relationship between the Individual and their Environment	20
Figure 4: The Cognitive Continuum	24
Figure 5: PRISMA Flow Diagram Error! Bookmark not def	ined.
Figure 6: Range of Athlete Rankings by Coaches at Time Point 7 Error! Bookmar	k not
defined.	
Figure 7: Number of Athletes Placed into Like Groups Error! Bookmark not def	ined.
Figure 8: Number of Athletes Classified by Group for Each Level of Agreement E	rror!
Bookmark not defined.	
Figure 9: The Coach-Informed Talent Identification Process	54
Figure 10: The Coaches' Eye during Talent Identification	65

List of Publications and Conferences

The following articles and conference presentations arose from this thesis.

Peer-reviewed journal articles

 Roberts, A.H., Greenwood, D., Stanley, M., Humberstone, C., Iredale, F., & Raynor, A.
(2019). Coach knowledge in talent identification: A systematic review and metasynthesis. *Journal of Science and Medicine in Sport*. 22(10): 1163-1172.
https://doi.org/10.1016/j.jsams.2019.05.008

- Roberts, A.H., Greenwood, D., Humberstone, C., Iredale, F., Stanley, M., & Raynor, A. (2020). Understanding the 'gut instinct' of expert coaches during talent identification. *Journal of Sports Sciences*. Advance online publication. https://doi.org/10.1080/02640414.2020.1823083
- **Roberts, A.H.,** Humberstone, C., Greenwood, D., Stanley, M., & Raynor, A. *The Coach's Eye:* An elite coach's decision-making process during talent identification. [Manuscript Under Review].

Roberts, A.H., Humberstone, C., Greenwood, D., & Raynor, A.J. (2020). Pilot study on the reliability of the coach's eye: Identifying talent throughout a 4-day cadet judo camp. *Frontiers in Sports and Active Living.* 2:596369 https://doi.org/10.3389/fspor.2020.596369

Conference presentations

- Roberts, A.H., Raynor, A., Greenwood, D. (2020, March). A new conceptualisation of sporting talent: The role of the coach. [Poster presentation]. Exercise and Sports Science Australia (ESSA) Conference, Perth, Australia. (Delayed due to COVID-19).
- **Roberts, A.H.**, Raynor, A., Greenwood, D., Humberstone, C., Iredale, F. (2019, November). *Proposition of a new conceptualisation of the longitudinal talent identification process.* [Conference presentation]. 3rd Scientific Conference on Motor Skill Acquisition, Lohja, Finland.
- Roberts, A. H., Greenwood, D., Humberstone, C., Iredale, F., & Raynor, A. (2018, November). *A case study in talent identification*. [Conference presentation].
 Australasian Skill Acquisition Network (ASAN) Conference, Sydney, Australia.
- **Roberts, A.H.**, Greenwood, D., Humberstone, C., Iredale, F., & Raynor, A. (2018, March). *A constraints-based approach to talent identification of future athletes*. [Poster presentation]. Exercise and Sports Science Australia (ESSA) Conference – Research to Practice, Brisbane, Australia.
- Roberts, A. H., Greenwood, D., Humberstone, C., Iredale, F., & Raynor, A. (2017, November). 'Talent doesn't make a champion – but it helps': The coaches' eye in talent identification. [Conference presentation]. Australasian Skill Acquisition Network (ASAN) Conference, Brisbane, Australia.
- Roberts, A. H., Greenwood, D., Humberstone, C., Iredale, F., & Raynor, A. (2017, November). How do coaches identify talent? An examination of international perspectives in combat sports. [Conference Presentation]. Scientific Conference on Motor Skill Acquisition, Lohja, Finland.

Statement of Author Contributions to Publications

This thesis, and the publications arising from the work, would not have been possible without contributions from many people.

Chapter Three: AHR led the development and writing of this review and provided an overall contribution greater than that of all co-authors. AHR designed and executed the search criteria; performed the screening, appraisal and analysis of included articles, and drafted the initial manuscript. DG independently screened articles for inclusion. MS independently appraised a subset of articles and consulted during the analysis stages. All listed authors contributed to the conceptualisation of the study and revisions of the manuscript.

Chapter Four: AHR led the development and writing of this paper and provided an overall contribution greater than that of all co-authors. AHR designed the study, collected and analysed the data, and drafted the initial manuscript. AJR and DG contributed to creation of the interview guide. MS supported AHR during data analysis. All listed authors contributed to the conceptualisation of the study and revisions of the manuscript.

Chapter Five: AHR led the development and writing of this paper and provided an overall contribution greater than that of all co-authors. AHR designed the study, collected and analysed the data, and drafted the initial manuscript. MS and AJR assisted with data analysis. All listed authors contributed to the conceptualisation of the study and revisions of the manuscript.

Chapter Six: AHR led the development and writing of this paper and provided an overall contribution greater than that of all co-authors. AHR designed the study, collected and analysed the data, and drafted the initial manuscript. AJR assisted with data collection; AJR and DG assisted with data analysis. All listed authors contributed to the conceptualisation of the study and revisions of the manuscript.

Abbreviated Terminology

AIS	Australian Institute of Sport
НРС	High-Performance Centre
NFL	National Football League
NSO	National Sporting Organisation
PRISMA	Preferred Reporting Items for Systematic Reviews and Meta-Analyses
TID	Talent Identification

It is our choices that show what we truly are, far more than our abilities

- Albus Dumbledore

JK Rowling, Harry Potter and the Chamber of Secrets

THIS PAGE INTENTIONALLY LEFT BLANK

CHAPTER ONE

Introduction

Using the theoretical framework of ecological dynamics, this program of work will explore the *coaches' eye* – the decision-making process of elite sports coaches during the athlete talent identification process. This thesis was completed within the context of three Australian national sporting organisations (NSOs) and uses combat sports as a task vehicle. This introductory chapter will present the background, research problem, context, and provide an overview of this thesis. This research will add to the body of knowledge in talent identification by exploring the role of the coach in this process.

Background of Talent Identification

Many domains use the word 'talent', yet the term does not have a standard definition. It has been used to describe innate abilities, biological predispositions, current skill level and perceived capacity to improve, among other definitions of the term (Till & Baker, 2020). Talent identification occurs when participants within a given domain or activity are recognised as having the potential to become successful elite performers (Abbott, Button, Pepping, & Collins, 2005; Bompa, 1985; Till & Baker, 2020). The opportunity to increase success while reducing resource expenditure is appealing in many settings including business, education, leadership and medicine (see Mcdonnell 2011; Putallaz, Baldwin & Selph 2005; Rhodes, Brundrett & Nevill 2008; Bell et al. 2011 respectively for examples). As such, talent identification has been a focus of research in these, and other domains for decades. Suppose those with the potential to be champions can be recognised early in sport. In that case, stakeholders such as coaches and national governing bodies can ensure that they invest appropriate resources into those athletes. Appropriate resource investment and development opportunities will ideally produce a larger number of elite performers by reducing time to peak performance and inappropriate financial investments, increasing

coach effectiveness, and increasing chances of international success (Abbott, Collins, Martindale, & Sowerby, 2002; Bompa, 1985; Ford et al., 2020; Johnston & Baker, 2020). Ostensibly, talent identification aims to recognise those who will succeed in the future. However, an existing limitation in the field is the propensity to focus on anthropometric and physiological measures (Bompa, 1985; Pearson, Naughton, & Torode, 2006; Vaeyens, Lenoir, Williams, & Philippaerts, 2008) or results at junior competition levels (Boxing Canada, 2016; McCarthy & Collins, 2014), rather than encompassing a holistic view of the athlete (Unnithan, White, Georgiou, Iga, & Drust, 2012). This focus on 'measurable' traits has led to a tendency within current sporting contexts to identify *current ability* to perform, rather than the *potential* to perform (Abbott et al., 2005; Till & Baker, 2020).

Many existing talent identification approaches rely on short term 'snapshots' of an athlete's current ability. These methods provide practitioners and coaches with a "static, one-dimensional concept of talent" (Mahon, 2004, p. 17) which frequently fails to predict performance due to the dynamic complexities of talent (Davids & Araújo, 2010; Lloyd et al., 2015). A growing body of scientific work is demonstrating that talent is multidimensional and dynamic, explaining the difficulties in accurately identifying talented athletes before they reach high-performance levels. Talent identification is inextricably linked with talent development (i.e. a recognised, structured environment or system designed to enhance athlete development; see Martindale et al. [2005] for a full description). Neither identification nor development can succeed to their fullest potential without the other component.

Current conceptualisations of talent identification contain many stages (Figure 1). Briefly, detection is the process through which potentially talented athletes are directed toward a specific sport, typically based on their existing qualities (height, aerobic capacity, strength). Once detected into a sport, the athlete begins on their developmental pathway within a deliberate talent development environment. Identification is the stage when current

2

participants are deemed to have the potential to be elite. During confirmation, athletes spend a defined period of in a high-performance environment, after which they are (de)selected for a specific competition or squad. The cyclical, multidimensional model represents the non-linearity of talent. Once athletes are involved in sport, they will move through these stages at many times during their career.

Figure 1

Key Stages in Talent Identification and Development (from Vaeyens et al., 2008)





The talent lexicon, particularly within sport, has many different terms. The usage of these terms can be incredibly nuanced in some cases and used interchangeably in others. For example, there are many different definitions and conceptualisations of the word 'talent', even within sport. Gray and Plucker (2010) argue that athletic talent ought to be defined as the "exceptional natural ability of an individual to perform a sports-related task or activity" (p. 364); while Cobley, Schorer and Baker (2012) define talent as "the quality (or qualities) identified at an earlier time that promotes (or predicts) exceptionality at a future time" (p. 3). One significant difference between these two definitions is the distinction between *current* and *future* ability or skill levels.

For this thesis, the operational definition of the term 'talent' has been developed by incorporating definitions from previous research in this field, specifically the works of Cobley and colleagues (2012), Issurin (2017) and Till and Baker (2020). Thus, talent will be defined as the capacity for future high-level performance, specifically the presence (or absence) or particular skills, traits or qualities (and combinations thereof) that may predict future elite-level performance. Differentiation in the nuances of terms and their usage both in research and practice has led to debates and disagreements among scholars about semantics in word choices (Coutinho, Mesquita, Fonesca, & Fonseca, 2016; Till & Baker, 2020) which can arguably delay the progression of research on the topic as a whole. Additionally, our current understanding of in situ language use in this field may be insufficient to tease apart many of the subtleties of the talent identification process. To avoid confusion, relevant terms have been operationally defined in Table 1.

Two broad theoretical approaches have been used to describe talent identification: natural and scientific. 'Natural' talent identification occurs when an athlete chooses to participate in a sport (typically due to social factors such as parents, peers or school programs), and coincidentally demonstrates talent in said sport, subsequently achieving high-performance outcomes (Bompa, 1985; Malina, 2003). The success of this approach relies on large numbers of participants at a junior level, the most successful of whom rise to the top through natural selection processes. Conversely, 'scientific' identification (sometimes referred to as detection) is a more structured process through which individuals are funnelled into a given sport based on discrete physiological, anthropometric, and/or psychological traits (Bompa, 1985; Crespo & McInery, 2006; Tranckle & Cushion, 2006). These methods can succeed with a smaller participation base but rely heavily on the predictive validity of the characteristics measured. Historically, the two systems have been

4

viewed as orthogonal, or not aligned, however more recent work is beginning to appreciate the potential value of combining the two models into a more modern and holistic understanding of the talent identification process. In practice, talent identification in sport has had varied success rates using either the natural or scientific methods of selection, across a variety of sports (for examples see Barreiros & Fonseca, 2012; Brouwers, De Bosscher, & Sotiriadou, 2012; Kovalchik, Bane, & Reid, 2017; Koz, Fraser-Thomas, & Baker, 2012; Reid, Crespo, Santilli, Miley, & Dimmock, 2007).

In one of the few studies examining the long-term predictive validity of applied talent identification, Schorer and colleagues (2017) investigated the ten-year predictive ability of national-level coaches. The European Handball coaches in the study were able to predict athletes' future performance level in 79.3% of cases compared to the 85.2% accuracy of the post-hoc testing of motor test data. These findings indicate that coaches are similar in accuracy to empirical testing when predicting talent; however, it is essential to emphasise that this is a single study, and the only one of its kind, therefore the results cannot be considered conclusive. Similarly, Cripps, Hopper and Joyce (2019) demonstrated that coaches could correctly predict career outcomes of Australian Football players over four years in 63% of cases. These studies have begun to capture the efficacy of the *coaches' eye* during talent identification; however, neither have investigated the underpinning mechanisms enabling coaches to make these predictions.

Much of the existing research focuses on objective measurement of many athlete qualities, despite in situ talent identification being primarily subjective and performed on a day-to-day basis by coaches (Lyle & Cushion, 2017). In reality, although many organisations use scientific selection methods to direct potential athletes to their sport of 'best fit' (Güllich & Emrich, 2014; Western Australian Institute of Sport, 2019), coaches are responsible for the majority of talent identification decisions.

5

Table 1

0	perational	Definitions	of Thesis	Terminol	logy
-					- 37

Term	Definition
Confirmation	The extended period of training and assessment in which
	coaches evaluate an athlete's adaptation to talent
	development environments before making selection decisions
	(Rynne, Crudgington, Dickinson, & Mallett, 2017).
Detection	The (usually scientific) process of determining an athlete's
	suitability for a given sport. Typically refers to athletes from
	outside the target sport; also referred to as 'promotion'
	(Vaeyens et al., 2008).
Development	The recognised, structured environment that is most
	appropriate for athletes to accelerate their learning and
	performance potential (Abbott & Collins, 2004; Unnithan et
	al., 2012). Typically incorporates performance support
	services outside of coaching (e.g. strength and conditioning,
	nutrition, psychology, etc.).
Identification	Recognising participants within an activity (using natural
	and/or scientific methods) who have the <i>potential</i> to become
	elite performers (Cobley et al., 2012).
Natural identification/	When an athlete participates in a sport in which they
selection	coincidentally demonstrate talent and are subsequently
	identified as a future champion based on competition results
	at a young age (Bompa, 1985).
Potential	Latent qualities or abilities that, if developed appropriately,
	may lead to future success; having or showing the capacity to
	develop into something in the future.
Scientific	When an athlete is identified/selected based on the results of
identification/ selection	scientific testing (Bompa, 1985).

Selection	The active decision of choosing athlete/s for a
	team/squad/event, typically based on prerequisite levels of
	performance factors (Till & Baker, 2020).
Talent	The capacity for future high-level performance; or more
	specifically, the presence (or absence) of particular skills, traits
	or qualities (or a combination) that may predict future elite-
	level performance (Cobley et al., 2012; Issurin, 2017; Till &
	Baker, 2020).

Statement of the Problem: The Coach as a Critical Decision-Maker

The multifaceted nature of sport and coaching means that the coach's primary role is arguably that of a decision-maker (Abraham & Collins, 2011; Lyle & Cushion, 2017). Talent identification is one such scenario in which the coach must make decisions, determining which athletes will be (de)selected for training squads, teams and competitions. Recent work has highlighted the need to consider the complexities of talent in conjunction with non-linear long-term athletic development when forecasting future elite athletes (Baker, Wattie, & Schorer, 2019). Coaches appear to have an inherent understanding of the interactions between different components of athletic performance and how they can change over time and under different circumstances. They appear to identify talent using many information sources, drawing from both tangible (testing scores, competition results, times) and intangible ('gut', 'coachability', resilience) components (Johnston & Baker, 2020; Williams, Ford, & Drust, 2020). However, little is known about *how* coaches identify athletes and *why* they choose the athletes they do – that is, how coaches' use their *eye* to predict talent.

This thesis aims to address the current gap in knowledge related to the *coaches' eye* by exploring experiential coach knowledge of talent identification using the Olympic combat sports as a task vehicle. Specifically, this thesis addresses the following questions:

- 1) What is currently known about how coaches identify talent?
- 2) How do elite combat sport coaches identify talented athletes?
- 3) How does an elite combat sport coach make identification and selection decisions in situ?
- 4) How reliable are coach perceptions of talent?

To date, methodology in the area of coach knowledge and decision-making (particularly in talent identification) is limited. There are many difficulties in collecting data about decision-making and the underpinning (often tacit) knowledge used to make the decision. However, Lyle and Vergeer note that despite difficulties in capturing this information, even flawed methodologies are "an important stage in understanding the decision-making process" (2013, p. 127). They also note that researchers typically assume that coaches use existing knowledge structures, pattern recognition and mental models to make in situ decisions. However, research has yet to confirm or refute this theory.

Context of this Thesis

This thesis was completed in conjunction with the Australian Institute of Sport's (AIS) Combat Centre with the support of Boxing Australia, Judo Australia, and Australian Taekwondo. Combat sports represent an ideal task vehicle for exploring talent identification due to the technical and physical qualities required for success. Combat sports are those activities in which the competitive essence of the sport consists of direct physical combat between two athletes (Noh et al., 2015). They are categorised based on the technical demands of the sport into striking (boxing, taekwondo, karate, fencing), grappling (judo, wrestling, Brazilian jiu-jitsu) or combined (mixed-martial arts) (James, Haff, Kelly, & Beckman, 2016). The open skill, individual and opponent-based nature of these sports makes them distinct from many other types of sport (Lidor, Melnik, Bilkevitz, Arnon, & Falk, 2005) and increases the complexity of identifying talented athletes. Accurate talent identification is vital in these sports, as the different combat disciplines (boxing, fencing, judo, taekwondo and wrestling) represent a significant portion of the medal opportunities at the Olympic, Paralympic, and Commonwealth Games. At the 2016 Olympics, there were 65 medal sets (248 medals) to be won, representing over one-fifth of the medals available (International Olympic Committee, 2019). This number will increase at the next Olympic Games, with the inclusion of a mixed team judo event and the sport of karate for the Tokyo Games (74 medal sets from an available 339 sets) (The Tokyo Organising Committee of the Olympic and Paralympic Games, 2019). From an Australian perspective, the importance of this research becomes more apparent when one considers that in Australia's 123-year history of Olympic competition there have only been 12 combat sport medals won, representing approximately two per cent of Australia's all-time medal tally (Australian Olympic Committee, 2019).

According to the latest data from Sport Australia's AusPlay survey, 1.4% of Australian adults (age 15 or older) participate in boxing, and 0.2% participate in taekwondo, with 1.3% of Australian children participating in taekwondo. Data for adult participation in judo and children's participation across all three combat sports were unavailable as the participating population was too small to be accurately captured by the survey (Sport Australia, 2020a, 2020b). With a limited talent pool from which to draw the next generation of champions, it is essential to establish a thorough, evidence-based method to identify potential elite athletes in these sports. The national sporting organisations (NSOs) for these sports have recognised the need for improved talent identification procedures as the integral first step in improving these sports' talent development pathways (Australian Taekwondo, 2018; Boxing Australia Limited & Australian Institute of Sport, 2014; Judo Australia, 2019). Given Australia's limited international success in these sports, international coaches must be included in the investigation. Sport scientists need to support and educate combat sport coaches and their NSOs in best-practice for talent identification and provide information

9

about the current strengths and weaknesses of the process globally. The findings will assist coaches at all levels to better identify youth athletes with the potential for future success, and assist NSOs in creating equitable, evidence-based talent identification policies.

Myself as the Researcher

Having been 'identified' myself by the AIS at age 13, I came to this research as a coach and exercise physiologist with a long-standing interest in talent identification. I participated in a Talent Search program conducted by the AIS and, based on my results, I was directed into rowing. Within two years, I was winning state championships above my age group and selected for representative squads. This piqued my interest in the field – how could measuring a 13-year-old be so accurate as to know what I would be good at – and continue to excel at for the next ten years? This interest grew as I continued to be successful at rowing, more so than in other sports that I tried my hand at.

As I transitioned into a coaching role, I became more interested in the concept of identifying talent, particularly as I worked with high- athletes in the United States and Australia, aiming to get them university scholarships. How did college coaches identify talented athletes, especially those they were only able to see on video? What factors were they looking for, and why was it necessary for my 16-year-old field hockey goal-keeper to record her 100m sprint time? Having been identified myself, and now the one to identify and to help these athletes develop and demonstrate their talent, I began to reflect on my own practices as both a coach and sport scientist.

I drew on my coaching experience to drive my quantitative master's thesis research (Roberts, 2016; later published as Roberts, Walden, Carter, & Symons, 2019c), in which I answered an applied question raised by myself, other coaches, and athletes. In doing so, I realised the value of coach questions and insights, which I brought with me into the current project. During my early stages of exploring the methodological options to approach this research, it quickly became apparent that a qualitative approach would be most appropriate to answer the questions that I had. Although my background was limited to quantitative methods, I value the knowledge and experiences of the coaches at the coalface. Therefore, the majority of this thesis is based on qualitative methods.

Before I began this research, I had little to no interest in or knowledge of combat sports. While varying across many sports, my own playing and coaching experience was limited to team ball sports and closed-skill, individual sports. I entered this study as an 'insider' to high-performance sport (as a coach, athlete and physiologist), but as a cultural 'outsider' to combat sports (Thorpe & Olive, 2019; Tinker & Armstrong, 2008). I differed from the coaches who participated in this research in a variety of ways, not the least being our age, cultural backgrounds, primary language spoken, sporting backgrounds and education levels. I viewed these differences as an advantage when completing this research. My 'insider knowledge' of coaching helped to establish rapport with the coaches (Thorpe & Olive, 2019) while my naivety of the sports in question went some way towards restoring any perceived power imbalances.

I used this naivety to establish a rapport with the coaches, allowing for the cocreation of new knowledge. This enabled me to understand the participants on their terms, encouraging them to explain their experiences to me in detail rather than assuming that I had understood 'what they meant' (Tinker & Armstrong, 2008). My position as an outsider in combat sports gave me critical distance from the topics, allowing me to appropriately analyse the respondents' views while still understanding the greater context of their perspectives (Thorpe & Olive, 2019; Wheaton, 2002). Having positioned myself in relation to the research topic and approach, I will now provide an overview of what the reader will find in the coming chapters.

11

Thesis Overview

This research uses the theoretical framework of ecological dynamics to examine coach decision-making during talent identification, using combat sports as a task vehicle. The series of studies in this thesis employ a range of methods to explore facets of the phenomenon commonly referred to as the *coaches' eye*. Colloquially, the *coaches' eye* refers to sport coaches' apparent ability to 'see' talent in athletes before they have begun to perform at a high level. This thesis explores extant research and experiential coach knowledge to understand how coaches identify talented athletes within their sporting contexts; investigating how the constraints of a coach affect their perception of talent and subsequent forecasts and selection decisions. This thesis is presented in eight chapters, briefly outlined below (Figure 1-2).
Figure 2



Flow Diagram of Research Stages with Chapters and Brief Methods

Chapter One provides an introduction to and contextualisation of this thesis, detailing the aims and significance of this research. Chapter Two synthesises and critiques extant background literature to frame this work program, providing the background knowledge necessary to examine the concepts of ecological dynamics, decision-making and talent identification. This chapter highlights some of the limitations in current talent identification research and indicates the need for a greater understanding of how experiential coach knowledge can enhance the field. Chapter Three presents the results of a systematic literature review and meta-synthesis on the use of coach knowledge in talent identification research to date. The findings from this meta-synthesis provided the direction for the subsequent investigation. Chapter Four uses semi-structured interviews with twenty-four expert coaches to examine the thought processes underpinning their decision-making during talent identification. Findings showed that the key influences on coaches' decision-making are their experience, abilities, time available, and decision context.

Chapters Five and Six examine coaches during the talent identification process, with a single coach over 18 months and a group of nine coaches over a four-day camp. The expert coach who participated in the case study detailed in Chapter Five demonstrated that coach forecasts of athlete potential change over time and supported the findings of Chapter Four. Chapter Six tracked the changes in coach perceptions of athlete talent over a four-day youth camp. This quantitative study indicated that coaches require time to attune to a group of athletes; in four days, a group of nine junior national judo coaches could not agree on individual athletes' potential. Chapters Three and Four have been published in peerreviewed journals, and Chapters Five and Six are currently under review. They have been reformatted in APA (7th) style in order to maintain a consistent thesis style.

Chapter Seven presents a synthesis of the results of this thesis through the creation of two new models: the *Coach-Informed Talent Identification Process* and the novel mode of the *Coaches' Eye in Talent Identification*. These models and their practical and theoretical outcomes are discussed in detail. Finally, Chapter Eight provides a summary of each chapter and contains recommendations for future research.

Significance of the Research

Through modelling the coach as the performer, this thesis investigates how the individual experiences, knowledge and abilities of a coach interact with the specific task and environmental constraints to shape coach decisions during talent identification. This thesis is novel in its use of ecological dynamics to explore coach decision-making during the talent identification process.

The results of this thesis provide evidence that the use of subjective coach judgments is an integral part in successful talent identification, allowing Australian coaches, practitioners and NSOs to make informed decisions regarding the use, inclusion and weighting of coach judgments during the talent identification process. In addition to contributing to the long-term development and sustainability of Australian combat sports, the inclusion of international-calibre coach participants in this research ensures that the findings may be applicable globally. Combat sports are among the most popular sports in the world (Kordi, Maffulli, Wroble, & Wallance, 2009), and the international reach of the data collected for this thesis ensures that the results apply to a wide range of contexts and countries.

This research also has implications for talent identification programs, many highperformance sporting pathways, as the use of the coach to identify future talent is widespread. As coaches are present in all levels of all sports, the findings of this investigation may apply to many open-skill sports in which the coach is an integral part of talent identification procedures. The knowledge gained from this research may also be used in coach education and development, allowing younger or less experienced coaches to better understand *how* to identify talent effectively.

By better understanding the value and best-practice applications of coaches in the talent identification process, NSOs can utilise the coaches they already employ to greater effect. By increasing coach knowledge about talent identification, athletes may be able to be brought into the sport earlier or through new channels, and coaches may encourage more athletes to stay involved in the sports for longer. This thesis shows that while the *coaches' eye* may not be 'reliable' in the word's statistical sense, the processes that coaches follow and influences on their decision-making are consistent across coaches.

CHAPTER TWO

Literature Review

Foreword

This five-section literature review addresses content relevant to the coach's role during talent identification with a specific focus on combat sports. The first section reviews the theoretical framework of ecological dynamics, which underpins this thesis. This framework links ecological concepts to behaviours demonstrated in dynamic sporting environments, such as identification decisions. The second section will review the latest work on decision-making, highlighting the current gaps in this research related to coaching. The third section will discuss an ecological approach to decision-making, tying the underpinning theoretical framework with coaches' decision-making process. The fourth section provides an overview of talent identification as it currently stands, highlighting some of the current limitations in this space and how the knowledge of expert coaches can fill some of these gaps. This section will also cover the current methods of talent identification used in combat sports. Finally, the fifth section will examine expertise and experiential knowledge in sport coaches, discussing the role of experiential knowledge in empirical research.

Theoretical Framework: Ecological Dynamics

An ecological approach is particularly relevant to scaffold this discussion of talent identification, as an understanding of the nonlinear nature of athlete development and adaptive patterns of human behaviour is imperative when exploring talent identification. Incorporating the concepts of ecological psychology and dynamical systems theory, ecological dynamics presents human movement as a series of complex, interrelated systems (Davids, Button, & Bennett, 2008). Specifically, ecological dynamics emphasises the performer-environment relationship, highlighting that a performer's actions are highly contextual, and influenced by information from the environment (Araújo, Davids, & Hristovski, 2006). A key tenet of ecological dynamics is the "complementarity" (Gibson, 1979, p. 127) of the performer and their environment. Actions are interrelated with perceptual information from the environment, so neither the performer nor the performance environment should be examined in isolation.

Complex and dynamic systems are synonymous with the sport environment and problem-solving behaviour in this context. Both the athlete and the sport itself are dynamic and continually changing, with changes in one affecting the stability of the other. A key feature of dynamic systems with multiple interacting constraints is that the rate of change is highly dependent on the changes in other areas of the systems (Kauffman, 1993). In the search for stability within multiple, interacting constraints, the relationship between an individual and their environment is (relatively) stable. Therefore, this is an appropriate level at which to analyse how constantly changing and adapting systems can stabilise to afford elite performance.

If the coach is the performer during talent identification (i.e. the performance is the act of identifying talent), then the application of ecological dynamics becomes even more relevant. In this context, the athlete forms part of the environment in which the coach is performing. During talent identification, a coach perceives information about the athletes within their environment whilst moving within and interacting with the environment to facilitate information gathering, which in turn shapes further behaviours and decisions.

Ecological Psychology

When sporting talent is expressed, an individual must possess a complementary combination of innate physical and mental traits, as well as an appropriate environment to stimulate and nurture this potential. The ecological dynamics framework appreciates that these aspects shift dynamically throughout an athlete's life (Ackerman, 2014; Davids & Araújo, 2010; Davids et al., 2008), and is, therefore, a useful method for examining this phenomenon. A component of ecological dynamics, ecological psychology emphasises the importance of a performer's interaction with their environment when attempting to understand behaviour (Barker, 1968; Brunswik, 1956; Gibson, 1979). An individual performer processes information from their environment, shaping their behaviour, which shapes their future perception of the environment. This relationship creates a symbiotic and continuous relationship between perception of the environment and behaviours within it – referred to in the literature as a perception-action coupling (See Figure 3) (Bruineberg & Rietveld, 2014; Davids et al., 2008).

Despite growing evidence that sporting performance is highly contextual, sport science and talent identification research tend to disproportionately focus of the 'individual' side of the equation (Davids & Araújo, 2010). This limitation is particularly relevant in opponent-based, dynamic sports, such as combat sports, in which the environment and context of performance can change dramatically between or even within matches. In combat sports, athletes continuously adapt their actions based on the most recent information available to them about their opponent, who in turn is adapting their own behaviours, creating continual variability and a unique scenario in each match (Inui, 2018; Maloney, 2018).

Figure 3

Relationship Between the Individual and their Environment (Based on Gibson 1979)



Dynamical Systems Theory

Dynamical systems theory positions individuals as complex, nonlinear systems consisting of several interrelated but independent parts and systems (Araújo et al., 2006; Clarke & Crossland, 1985; Kelso, 1995; Newell, 1985). These independent systems, or constraints, shape behaviours and performances. Coaches work "collectively in *dynamic* and often *non-linear* ways within a *complex* adaptive system" (Bowes & Jones, 2006, p. 236) [emphasis added]. Complexity and non-linearity are key features that define a dynamic system (Davids et al., 2008; Rein, Davids, & Button, 2010; Seifert, Komar, Araújo, & Davids, 2016). Sport is a dynamic environment, and it is well established that both human behaviour and human sporting development are nonlinear (Abbott et al., 2005; Button, Seifert, Chow, Araújo, & Davids, 2020). In practice, non-linear behaviour allows a system (i.e. performer) to demonstrate both stable and unstable behaviours and explains how the sub-parts of a system (i.e. constraints) interact to influence or compensate for the other components in the

system (Chow, Davids, Hristovski, Araújo, & Passos, 2011; Davids et al., 2008; Rein et al., 2010). The lack of consistency in talent identification research is evidence of its complexity. This complexity is due primarily to the interacting constraints on both athlete and coach behaviour. These complex interactions, present on many levels, indicate that dynamical systems theory is an empirically valid method to explore both sport as a whole and the coach's role within it.

Summary. Ecological dynamics emphasises that an individual's behaviour is emergent, co-created by the interaction of the individual and their dynamic environment. Ecological dynamics has been used to analyse the interplay between perception and action in athletes (Pinder, 2012); however, the link between a coach's perceptions and their actions (i.e. behaviours or decisions) has not yet been investigated. By applying the framework of ecological dynamics to talent identification, we can structure and improve understanding of the strengths and limitations of existing methods of talent identification. Within the context of combat sports, this framework can provide a lens through which to explore the identification of talent in open-skill sports. As athletes are required to adapt and change according to the dynamic environments within which they train and compete, it is necessary to forecast these changes and their interactions if we wish to predict the long-term potential of athletes. Ecological dynamics also provides a robust framework through which we can begin to explore and explain coach behaviours (i.e. decisions) during the process of talent identification. Specifically, an ecological dynamics approach to decision-making can help develop the body of knowledge relating to coach-environment (athlete) relationships that are essential for talent identification.

Decision-Making

Making decisions is an inherent and integral part of a coach's role and is identified as one of a coach's critical function (Abraham & Collins, 2011; Lyle & Cushion, 2017; Nash & Collins, 2006). From deciding which training exercises are appropriate, to deciding whether an injured athlete play; coaches spend most of their time making decisions that range from automatic to routine and repetitive to decisions in which there is no clarity about the outcomes (Lyle, 2010). Despite the expectation that coaches will make the right decisions, and at the right time, it is still unknown how and why coaches make the decisions they do. By understanding the processes underlying decision-making, we can begin to "explain why people choose one option instead of another from a set of alternatives when they do not know the outcome" (Marasso, Laborde, Bardaglio, & Raab, 2014).

The Decision-Making Continuum

Decision-making is conceptualised in several ways, one of the most common being that of two cognitive processes: System 1, or intuitive; and System 2, or analytical (Dhami & Thomson, 2012; Kahneman, 1991, 2011). Intuition, a hallmark of expertise, is a term used to describe decisions made by experts when they cannot articulate how they arrived at a decision. It is the primary differentiation between expert and novice decision-makers – both the number of 'gut decisions' and the confidence to trust those instincts increase with expertise (Schempp & McCullick, 2010). Intuitive decisions are made quickly and are can often be described as 'naturalistic', occurring in dynamic, uncertain, real-world environments (Hoffman & Yates, 2005). It has been shown that deliberation can lead to a decrease in the accuracy of decisions that were otherwise made intuitively (Plessner & Haar, 2006).

Comparatively, in System 2, or analytical decisions, the decision is broken down into smaller units (possible consequences; the probability of events; pros and cons; etc.). In a purely analytical approach, these smaller units are analysed in turn, attempting to objectively measure or calculate each aspect to reach a point at which the decision can be made based on the balance of the calculations (Hoffrage & Marewski, 2015).

Another interpretation of cognition views decision-making on a continuum, with decision modes dependent on the specific context in which the decision is made. The cognitive continuum theory (Figure 4) positions naturalistic, or more intuitive decisions at one end and analytic cognitions at the other, with what Hammond has termed 'quasirationality' found between the two (Dhami & Thomson, 2012; Hammond, 2000). According to the cognitive continuum theory, decisions are made by combining intuition and analysis; depending on the nature of the decision, the time available to make it, and the information available to the decision-maker.

As coaches must make decisions in many different contexts requiring different timescales, coaches must operate on all levels of this continuum. Some decisions must be made in seconds, as during a game, others have more extended timelines that take place over months (e.g. annual planning). Coaches make decisions affecting athletes' performance/development at multiple time points, and the differing timescales, environments and consequences of the decisions will affect where on the cognitive continuum they fall. Interestingly, talent identification could arguably fall anywhere on this continuum, depending, as previously stated, on the context in which identification occurs and the time available to the coach.

Figure 4

The Cognitive Continuum (from Dhami and Thomson, 2012)



Reproduced with permission from Elsevier.

Coach Decision-Making

Decision-making in sport has been a topic of research for decades in pursuit of a better understanding of athletic excellence and, by extension, how to train future athletes to make decisions in the same way as current elite competitors. Within sport, decisions are naturalistic; that is, the decision-maker encounters the decision in context and with a degree of familiarity with the task (Johnson, 2006; Kaya, 2014). Likewise, it has been established that their knowledge and experiences shape their judgments and decisions as they recognise and interpret situations (Hoffman & Yates, 2005; Lyle & Cushion, 2017). However, there is a dearth of research into coach decision-making, with the majority of existing work in sporting contexts focused on athlete decision-making (Cropley, Thelwell, Mallett, & Dieffenbach, 2019; Lyle & Vergeer, 2013).

Investigations into coach decision-making are typically divorced from real-world situations. Using questionnaires and surveys or simulations of fictional scenarios for data collection simplicity and methodological comparison, which while increasing the comparative nature of collected data, decreased the context within which the decision is made, changing the relevance and implications for the involved coaches (Dennis & Carron, 1999; Giske, Benestad, Haraldstad, & Hoigaard, 2013; Lyle & Vergeer, 2013). More recently, in-depth interviewing methodologies have been used to better capture the information which underpins coach decision-making. These techniques have explored how experiences and values impact coaching decisions (Callary, Werthner, & Trudel, 2013; Collins, Collins, & Carson, 2016; Morris et al., 2019; Potrac, Jones, & Armour, 2002) and factors that impact selection decisions (Bradbury & Forsyth, 2012; Hill & Sotiriadou, 2016). Interview methods are coming closer to capturing the contextual factors that are inherently important in decision-making in sport.

It is apparent that coaching decisions are dynamic, involving serial decision points rather than a single decision; using information gathered and processed over time. However, the reflective nature of interviews influences coach perceptions of their processes, including their justification for prior decisions, presenting a limitation in these methods. Moving away from a reliance on recall or 'what would you do' hypothetical questioning, to capturing the decision-making in real-time is an important step forward. Capturing the intuition and analysis involved in both small and large decisions within dynamic sports contexts remains a challenge.

An Ecological Approach to Decision-Making

The ecological approach conceptualises decision-making as an ongoing process through which the behaviours of the performer are emergent based on the coordination of environmental perceptions and related actions (Raab, Bar-Eli, Plessner, & Araújo, 2019). Ecological approaches have been used to study decision-making in athletes (Araújo et al.,

2015; Hristovski, Davids, & Araújo, 2006; Maloney, Renshaw, Headrick, Martin, & Farrow, 2018) and how this process occurs (Marasso et al., 2014). The ecological approach is an appropriate lens through which to approach coach decision-making, as decision-making in sport is "a complex, temporally extended process expressed by actions at the ecological scale" (Araújo, Davids, Chow, & Passos, 2009, p. 160).

From an ecological perspective, decision-making is grounded in the interaction between the performer and their environment. If decisions are emergent in this fashion, it follows that they are probabilistic and subject to noise or interference within the performerenvironment dyad (Balagué, Hristovski, & Vazquez, 2008). Decisions (or goal-directed actions/behaviours) are made to progress towards a goal, with each choice narrowing the possible future decisions along the path to a goal (Araújo et al., 2009). As such, decisions are emergent – appearing based on the current context and available information, rather than being pre-determined. If decisions are emergent, this implies that there is no such thing as a 'correct' decision (Balagué et al., 2008). Instead, there is only the most appropriate decision based on the available task, environmental and individual constraints at the time, and the performer's perception of these constraints. It has been proposed that individuals perceive environmental information differently based on their goals, and their ability to act on these goals (Cañal-Bruland & van der Kamp, 2009; Dicks, Araújo, & van der Kamp, 2019; Hristovski, Davids, Araújo, & Button, 2006). Dynamical systems theory uses the concept of constraints to conceptualise behaviour as the outcome of interactions between bounded system components (Davids et al., 2008; Newell, 1986). Within dynamical systems theory, the term 'constraint' is not used as in traditional English terminology, as a limitation or restriction; rather, it is scientific terminology that describes unique environmental properties (i.e. constraints) that guide participants' behaviours during the performance of a task. By understanding constraints as they exist for an individual in a given context and how they interact with one another, we can better understand and explain an individual's behaviour,

especially in dynamic sporting contexts. There are three categories of constraints that guide movement through invitation, discouragement, and movements or behaviours (Kelso, 1995; Newell, 1986; Newell & Valvano, 1998; Newell & Jordan, 2007). The behaviour, decisions and/or movements of a performer are directly influenced by the interactions of the individual, task and environmental constraints.

Individual constraints are characteristic of the person (Davids et al., 2008; Newell, 1986). These include physical factors, psychological traits and cognitive features which are unique to each person. Some individual constraints are inherent and cannot be changed, such as height, whereas others can be improved through training or experience, such as strength or the ability to read a play (Davids & Baker, 2007; Davids et al., 2008). Individual constraints are also flexible and as a result, can be temporary, such as fatigue or stress. Individual constraints shape how a performer will find solutions to complete a given task, as when individual constraints change, so too do the movement solutions to solve practical challenges. Within the talent identification literature, a relevant individual constraint is that of experience. As each person's knowledge and experiences are unique, their response to a given situation will be similarly unique. For coaches, their playing and coaching experiences combine with their education and knowledge to create individual constraints to their coaching abilities.

Environmental constraints are those that are external to the performer and timedependent (Glazier, 2015). These include global influences (temperature, altitude, ambient light, playing surface); local factors (sociocultural factors, access to coaching) and, more recently, any physical constraint external to the individual (implements or tools used) (Davids et al., 2008; Glazier, 2015; Newell, 1986; Renshaw & Chappel, 2010; Renshaw & Davids, 2014). Coaches must work with the environmental constraints present in a given training or competition environment, such as the weather. However, rather than adapting their *actions*, as an athlete must, a coach must adapt their decision-making process and strategies to

enable athletes to achieve the same goal for the session. For example, on a day with high temperatures, coaches may adjust training task durations to account for adjusting physical demands imposed on the group by the environmental constraints.

Task constraints are specific to the immediate performance context, particularly the goal of the activity context (Davids et al., 2008). They are influenced by the rules or conditions of a given activity, including changes in field size, game-specific conditions such as referee decisions, or instructions issued by a coach (Al-Abood, Bennett, Hernandez, Ashford, & Davids, 2002; Mellalieu, Neil, Hanton, & Fletcher, 2009; Newell & Ranganathan, 2010). Within individual, opponent-based sports such as combat sports, the opponent themselves are considered a task constraint as they differ in fighting styles and change between rounds of competition. Thus, the performer has to adapt to new task demands with each bout. In the context of coach-based decision-making, a task constraint may be the importance of performance at a major championship event. In selecting the athletes on the team, increased pressure to perform in the short term, or a focus on the long-term development of athletes may change the coach's decisions of which athletes are selected for a given team.

Constraints interact and change over time as a result of experience, maturation and shifting goals, acting in concert to create self-organised, emergent behaviours (Brymer & Davids, 2013; Davids, Araujo, & Shuttleworth, 2005; Davids et al., 2008). Each factor's relative contribution is highly contextual, based on the interaction between the individual and the specific performance (Oppici, Panchuk, Serpiello, & Farrow, 2017). Long-term success in sport is characterised by consistent adaptation to the dynamic constraints presented by athlete maturation and changes within the sport. Successful elite level performers can consistently solve the performance challenges that arise from the interactions of the dynamic constraints. According to Newell and Jordan (2007), constraints are present at multiple levels of the dynamic system, with many different "time scales of influence" (p14).

This is particularly relevant for discussions of decision-making in sport coaching, as coaches make short-, medium and long-term decisions as part of their daily responsibilities. An understanding of constraints, their interactions and their dynamic nature highlights the importance of adaptability in sport performance and, by extension, sport coaching.

Affordances

Information about the environment perceived by the performer creates an opportunity for action, known as an affordance (Fajen, 2005; Gibson, 1979). The term 'affordance' describes the relationship between a performer and their environment, in terms of the actions or behaviours available in a given context. These opportunities for action are closely linked to constraints and perception-action coupling, as a performer's actions are driven by their perception of the constraints present in their immediate environment. Skilled performances result from a perceived affordance and the selection of an action or behaviour available to the performer to achieve their task goal (Craig & Watson, 2011; Dicks et al., 2019; Fajen, 2005).

Affordances are not only features of the environment, but rather the integration of the performer's capabilities, their perceptions of environmental features, and the opportunity for actions that they allow. This includes the ability to attune and calibrate perception to key informational sources from the environment that may (dis)allow their preferred course of action (Vilar, Araújo, Davids, & Renshaw, 2012). When calibrating perception, the intended outcome must be considered as different intentions will require different specifying variables to make the most appropriate decision. In sport, performers first intend to act, and then determine whether that action is possible (afforded) or not (Craig & Watson, 2011). Performers may detect many variables that are not relevant to their intended action, and with experience, performers learn to attend to more useful information sources of the performance environment (Araújo et al., 2009; Gibson, 1966; Gibson, 1979). This process, known as perceptual attunement, allows skilled performers to perceive different affordances than novices due to superior technical and perceptual skills. For instance, Milazzo et al. (2016) demonstrated that experienced karate athletes exhibit faster response times to sport-specific information. That is, they reacted faster than novices to simulated attack scenarios - a consequence of superior physical abilities in conjunction with being attuned to relevant higher-order environmental information. Hristovski et al. (2006) demonstrated that boxers select different punches based on their perceived efficiency for the current interpersonal distance. This finding was also demonstrated in experienced taekwondo athletes (Maloney et al., 2018). As skill increases, so too does the number of appropriate actions (affordances) available in a given situation.

Selection of the contextually optimal affordance is a hallmark of expertise, demonstrating the 'grip' that experts have on their current environment (Bruineberg & Rietveld, 2014). An experts' grip allows them to achieve more consistent outcomes regardless of potential factors constraining the task (Davids, Glazier, Araujo, & Bartlett, 2003; Fajen & Warren, 2003). Experts develop optimal grip through repeated exposures to environments similar or identical to those present during performance, allowing for the opportunity to practice attuning to and selecting appropriate specifying affordances (Headrick, Renshaw, Davids, Pinder, & Araújo, 2015). This calibration process is a necessary aspect of attuning perception and developing grip, as calibration allows performers to perceive accurately in different circumstances, such as following a growth spurt (i.e. a change in body dimensions and therefore action capabilities) (Araújo et al., 2009). Successful calibration results in appropriate decisions within changing constraints and available affordances.

Talent Identification: A Research Summary

Each year, sporting organisations and governments worldwide spend millions of dollars on talent identification (Vaeyens et al., 2008) in attempts to predict their future world champions. It is proposed that if those with talent are recognised and integrated into programs that provide proper support and coaching, a larger number of elite performers will be produced (Abbott et al., 2002; Buekers, Borry, & Rowe, 2015). In one of the earliest English-language articles on sporting talent identification, Tudor Bompa (1985) provided a list of advantages of talent identification in sport. This list included substantial reductions in the time required to reach peak performance, increases in coach effectiveness, increases in numbers of high-performance athletes, and greater chances of international success; in addition to reducing waste of time, energy and money. While the benefits of early identification and subsequent development have not changed, our inability to accurately identify talent, or its components, limits its effectiveness.

From being viewed primarily as a 'Soviet tool' (Washburn, 1956) to becoming a standard and integral everyday part of sport, the process of identifying talent has evolved significantly throughout the past seventy years. Talent identification began with evaluating athletes' somatotypes to determine whether physique was a selective factor in athletic performance (Carter, 1970; Medved, 1966). This was a natural evolution from the knowledge that athletes in different sports had different somatotypes both from each other and significantly different from that of the normative population (Kohlrausch, 1929; Sheldon, Stevens, & Tucker, 1940; Tanner, 1964). From here, the transition to using anthropometric and physical performance characteristics to guide athletes into the sport to which they are best physically suited (Aule & Loko, 1982) was a natural next step.

Over time, it has been recognised that talent identification is multidimensional, and biophysical measurements alone are insufficient to predict talent accurately, or even measure current ability levels (Johnston, Wattie, Schorer, & Baker, 2018; Vaeyens et al.,

2008). Despite this, the focus in literature remains on evaluating current abilities and capacities to predict future performance (See Bennett et al., 2018; Chiwaridzo et al., 2018; Li, De Bosscher, Pion, Weissensteiner, & Vertonghen, 2018 for examples). This limitation is likely due to psychosocial factors being notoriously difficult to measure (Anshel & Lidor, 2012; Rynne et al., 2017), rather than a lack of understanding of their importance (e.g. Johnston & Baker, 2020; Mann, Dehghansai, & Baker, 2017; Till & Baker, 2020).

Talent Identification in Combat Sports

Combat sports are an ideal task vehicle for studying the role of the coach during talent identification. The unique task constraints of combat sports, such as their dyadic interpersonal nature, offensive and defensive requirements, small distances between competitors, and weight requirements (Chen et al., 2017; Franchini, 2014; James, Robertson, Haff, Beckman, & Kelly, 2017; Krabben, Orth, & van der Kamp, 2019) emphasise the need to understand how coaches identify talent in these sports.

There is limited empirical understanding of athlete selection and identification processes within the combat sports literature. Three major studies have attempted to describe predictors of talent development in combat sports. Lidor and colleagues (2005) attempted to create a judo-specific ability test to measure talent among young judokas. This test was found not to correlate with athlete rankings either one- or eight-years post-testing. The test was declared to lack the sensitivity necessary to measure talent, suggesting that a more open-skill environment (more like that found in competition) would be able to better indicate talent in young athletes. The second and third studies related to talent identification in combat sports explored the relationship between junior and senior competition results. In youth judo, competition results accounted for only seven and five per cent of senior results in males and females, respectively (Julio et al., 2011).

In contrast, Li and colleagues (2018) found that junior performances were better predictors of success in senior boxing, taekwondo and wrestling. Sixty-one per cent of junior medallists (those that medalled at international competitions) went on to win medals at senior-level international competitions, while 90% of early achievers (those that were successful in senior-level competition while underage) could be "predicted" to win a medal at senior-level international competition. These studies highlight the limited research and inconclusiveness in existing talent identification in combat sports. By including both striking (boxing and taekwondo) and grappling (judo) sports and investigating a new area of interest within combat sports (i.e. the coach), this research has the potential to highlight similarities and differences in talent identification within these sports.

Limitations in Talent Identification

One of the major limitations in current talent identification research is that existing testing focuses on using measurements to differentiate between a-priori selected and non-selected athletes, which in turn are extrapolated into measurable factors that can identify talented athletes. For example, Keller et al. (2016) examined the technical ability of a group of youth soccer players and found that technical performance on passing drills could discriminate between players at the national elite, state elite and sub-elite levels. In a similar vein, Bennett and colleagues (2018) investigated using small-sided games to assess soccerspecific proficiency, finding that passes, touches and other skill-related measures were greater in higher-level players. What these and other similar studies do not account for is that they have drawn these conclusions from a-priori groupings – athletes were already categorised into their levels, and these tests were able to discriminate between these levels. The a-prior grouping is a significant limitation as these athletes may have had skill differences due to their training in said groups, rather than these factors being what put them on the team.

Another inconsistency in existing talent identification literature is the definition of 'success'. There is no consistency in how many athletes an identification protocol must identify to be considered a success, nor is there consistency in what level an athlete must reach to have been 'successfully' identified. For example, "only" 60% of female players that made the Women's Tennis Association Top 200 won major tournaments during their youth (Brouwers et al., 2012); while an equation for predicting Australian Football League draft status accurately classified 64% of cases was labelled a 'moderate' success (Robertson, Woods, & Gastin, 2015). Even within the same sport (in this case, volleyball), prediction success ranges widely from 38 to 98% (Stamm, Stamm, & Thomson, 2005). Conversely, research into the American National Football League (NFL) draft system has multiple definitions of success. Success has been determined through draft pick order, salary, career length, number of games played, length of time with one team, or number of championships won (Boulier, Stekler, Coburn, & Rankins, 2010; Hartman, 2011; Koz et al., 2012; Lyons, Hoffman, Michel, & Williams, 2011; Teramoto, Cross, & Willick, 2016). Inconsistencies in outcome measures and the physical-based nature of the tests have resulted in the NFL's predictive ability to combine (among other combine testing protocols), being poor.

Latest Work in Talent Identification

In recent years, there has been a resurgence of interest in identifying talent within sport. However, there have yet to be any 'breakthroughs' or significant transfer of concepts between research and practice (Collins, Macnamara, & Cruickshank, 2019). The lack of significant advances in the field may be due to the "complex interaction between genetic endowment, environmental influences and learning, technology and the vagaries of specific performances [which] make prediction of achievement very difficult" (Lyle & Cushion, 2017, p. 137). A relatively modern approach to talent identification in sport has been to measure technical and/or perceptual-motor skills (Keller et al., 2016; Woods, Keller, McKeown, & Robertson, 2016; Norjali Wazir et al., 2017) in an attempt to integrate multidimensional measures into the process. Investigations into the multidimensional aspects of performance have determined the importance of psychological factors and 'hidden attributes' (coachability, versatility, and flair) when determining an athlete's potential (Musculus & Lobinger, 2018; Roberts, McRobert, Lewis, & Reeves, 2019). There is also a move towards attempts to measure more multifaceted aspects of performance such as coordination, as in Faber et al. (2017) – also see O'Brien-Smith et al. (2019) for a recent review.

To address the complexity of talent identification, many sporting organisations now include 'confirmation' phases in their talent identification. The confirmation period is an extended period following an initial selection into a talent development environment, during which athletes are observed for progress within the new environment and for signs of less-tangible factors (e.g. 'coachability', work ethic, etc.) which are perceived to contribute to success within the sport. The confirmation period allows coaches to observe multidimensional and more initially imperceptible factors which affect long-term potential (Rynne et al., 2017).

The Coaches' Eye in Talent Identification

The key gap in much of the existing talent identification research is in *who* is determining 'talented' and 'non-talented' athletes. Christensen (2009) indicated that the idea of talent in sport is a matter of taste and that coaches identify talented athletes based on subjective 'feelings' rather than a specific, quantitative checklist. As a result, it is hypothesised that two different coaches will identify a different subset of athletes as talented. If this is the case, *who* is performing the identification is just as important as *what* is identified. As each person identifies talent slightly differently, it is important to explore both *who* is (and should) be identifying talented athletes (Bailey & MacMahon, 2018), and how they do it. The subjectivity of coach skill ratings has been investigated (e.g. McIntosh, Kovalchik, & Robertson, 2019; Cripps, Joyce, Woods, & Hopper, 2017), but only one study

has explored coaches' subjectivity during talent identification. In Jokuschies, Gut and Conzelmann's 2017 study, they investigated coach perceptions of talent, having coaches rate players and then comparing the coaches' subjective talent criteria to the athletes they labelled as 'talented'. They found a high correlation between the two, indicating some level of reliability between coaches stated preferences and their selection decisions. Coaches make 'choice decisions' when selecting players (Lyle & Vergeer, 2013); however, these choices and how they are made have not been examined.

Despite the depth of research into talent identification, investigations into coach decisions and their formation are limited in the literature (Christensen, 2009; Gulbin, Croser, Morley, & Weissensteiner, 2013). The traditional assumption is that the combination of the coach's 'professional eye', when combined with testing, provides the most effective talent identification (Lidor, Côté, & Hackfort, 2009). Several studies use coaches for corroboration of their results, particularly when assessing the skills of an athlete (see O'Connor, Larkin, & Mark Williams, 2016; Van Yperen, 2009) or attempting to validate testing batteries (i.e. as an external criterion for determining the predictive validity of a proposed talent forecasting method) (see Falk, Lidor, Lander, & Lang, 2004; Pienaar, Spamer, & Steyn Jr, 1998 for examples). However, none have directly investigated nor applied coaches' knowledge to inform the talent identification process and overall, there is limited research available on coaches' ability to predict talent over time.

While coaches are the 'gold standard' in many talent identification studies, there is a dearth of research into *how* and *why* coaches make the decisions they do. Previous talent identification research has examined the differences between coach predictions of talent and mathematical or computerised regression models and found that while expert coaches were better than novices at predicting success, they were only slightly better than chance (Schorer et al., 2017). This may be due to the subjectivity inherent in coach identification and selection, and differences in opinion and perception of what constitutes 'skill' in a given sport

or level (Woods, Raynor, Bruce, & McDonald, 2015). Humans can only process a limited amount of information, which means that their decisions are influenced by informational stimuli they can perceive and deem most important (Plessner & Haar, 2006). Their experiential knowledge influences the information to which coaches attune their attention. It is known that visual search strategies differ between experts and novices, as experts can anticipate which information will be most relevant, and where that information is likely to come from (Plessner & Haar, 2006). By applying a framework to understand coach decisionmaking, we can begin to break down some of the complexity around these processes and understand how coaches identify and select the athletes they do.

Expertise and Experiential Knowledge

Talent identification decisions are dynamic, complex and occur over time, thus can be understood from an ecological decision-making perspective. During talent identification, coaches do not have a pre-selected list of talented athletes – rather, they make decisions based on the current context and the information available. Coaches need to be able to attune to the appropriate specifying variables at the appropriate time to identify talented athletes.

A growing body of literature emphasises the valuable role that the experiential knowledge of elite coaches may play in scientific research (Greenwood, Davids, & Renshaw, 2012b; Lyle & Cushion, 2017; Waters, Phillips, Panchuk, & Dawson, 2019). Expert coaches can provide information and insights into aspects of their job which are lacking in the empirical literature (e.g. Eccles, Ward, & Woodman, 2009; Waters et al., 2019). They have also been used to underpin, guide and complement research (e.g. Collins & Collins, 2013; Willmott & Collins, 2017).

Experts are required to support research in fields in which there are many paths to the 'right' answer, such as sport. Expertise is the domain-specific skill set built from experience and knowledge, where is an expert is one who can efficiently and effectively access appropriate cognitions, thought processes and informational cues, while remaining flexible and adaptable to new situations (Lyle & Cushion, 2017; Nash & Collins, 2006; Schempp & McCullick, 2010). There are eight traditional methods of identifying experts and expertise, used individually or, more commonly, in a combination of two or more (See Table 2).

Another supportive hallmark of an expert is that they have a high degree of accuracy and precision when it comes to predicting events within their field of expertise (Mccullick et al., 2006; Schempp & McCullick, 2010). When considering the talent identification literature, coaches' domain-specific experiential knowledge and discriminative ability are particularly useful as a collaborative information source to increase understanding. Interestingly, within research exploring coaching in sport, experience and certification are typically used as the hallmark of expertise and presented in methodologies to justify the classification of the coach as 'expert'.

Table 2

Expertise Criteria	Explanation of Criteria
Experience	Experts have a large number of years of job experience
Certification	Experts are certified, through a specific rank achieved (e.g. head coach) and/or accreditation (e.g. achieved through formal study)
Social acclamation	Experts are identified by those who work in the relevant field
Consistency	Expert judgments should be consistent over time. That is, they should be internally consistent
Consensus	Expert judgments should be consistent between experts
Discriminative ability	Experts should be able to discriminate between nuanced or subtle differences that non-experts often overlook
Behavioural characteristics	Experts share common behavioural traits (e.g. self-confidence, adaptability, decisiveness, perceptiveness, communication) – See Abdolmohammadi & Shanteau (1992) for a full list
Knowledge	Experts have a high level of knowledge within their domain

Traditional Approaches to Identifying Expertise (Adapted from Shanteau et al., 2002)

Expertise in Coaching

In sport, the typical criteria for defining an expert coach include being a coach of a national team (certification), the number of years coaching (experience), the development of elite performers (consistency), or nomination or selection by others (social acclamation) as being an expert (Nash, Martindale, Collins, & Martindale, 2012; Shanteau, Weiss, Thomas, & Pounds, 2002). Previous investigations into coach knowledge have demonstrated that despite a lack of 'scientific' training and vocabulary, coaches possess practical, contextualised knowledge gained through experience (Côté, Salmela, Trudel, & Baria, 1995; Greenwood, Davids, & Renshaw, 2012a; Lyle & Cushion, 2017). Coaches' intuitive understanding of sports science concepts (Christensen, 2009; Gigerenzer, 2007; Simon, 1992) is based on years of individual experience, reflection and analysis (Baars, 2011; Christensen, 2009; Greenwood, Davids, & Renshaw, 2014) and can help explore the intangible interactions of constraints that underlie successful talent identification. We must understanding of how coaches perform the necessary tasks of their role, ultimately allowing detailed and sensitive measurements of coaching skill.

Despite the benefits of utilising coach knowledge in empirical sports science research, there have been relatively few investigations which take advantage of coaches' expertise. One potential reason is the inherent subjectivity found in knowledge formed through individual experiences (Lyle & Cushion, 2017). These individual experiences tend to provide insight into the differences rather than similarities in behaviour and are often limited in their transferability across sporting contexts. This may go some way to explaining the inability of existing coaching research to capture the complexities within the field (Barnson, 2014; Bowes & Jones, 2006; Cushion, Armour, & Jones, 2006).

However, these limitations are far outweighed by the potential contributions that coach knowledge can make to the scientific literature. Coaches' first-hand, practical experiences with different athletes shape their understanding of performance within their discipline. As such, they have a unique perspective on the underlying complexities of creating a champion; practitioners and researchers may be able to harness this unique perspective to drive empirical research to areas of practical significance for practitioners.

When discussing 'the eye of the coach', the term often refers to the subjective, intangible 'art' of coaching, and how coaches can perceive aspects of performance that others cannot. Coaches are continuously presented with a myriad of choices when working with their athletes, from day-to-day decisions around training sessions, game plans and interactions with athletes to more long-term choices such as how to periodise the season and team selections. It is apparent that "coaching is just as much of an inexact science as an exact science" (Pope, Penney, & Smith, 2018, p. 146) relying on opinions in equal measure with empirical data. This is particularly true when it comes to the field of talent identification.

Summary

Using the theoretical framework ecological dynamics to examine coach decisionmaking during talent identification will provide a holistic lens through which to examine this complex and dynamic process. Existing research utilises subjective experiential coach knowledge without investigating the basis and validity of this knowledge. Although previous work has demonstrated the value of coach knowledge in empirical research, experiential coach knowledge has yet to be explored in talent identification. As such, this thesis will focus on the application of experiential coach knowledge to supplement our understanding of talent identification in combat sports.

THIS PAGE INTENTIONALLY LEFT BLANK

CHAPTER THREE

Expert Knowledge in Talent Identification: A Systematic Review and Meta-Synthesis

This chapter has been published in full as the following peer-reviewed journal article:

Roberts, A., Greenwood, D., Stanley, M., Humberstone, C., Iredale, F., Raynor, A. (2019a). Coach knowledge in talent identification: A systematic review and meta-synthesis. *Journal of Science and Medicine in Sport, 22*(10):1163-1172

Foreword

The following chapter presents a systematic review and meta-synthesis, which aimed to synthesise the available literature on coach knowledge as it relates to talent identification. There is an apparent dissonance between reliance on coach judgments, opinions or decisions during talent identification in both research and practice. There also appears to be a lack of empirical understanding in this area, and what does exist is disparate across sports, athlete populations and coach demographics. For this reason, it is difficult for practitioners or researchers to access this information and apply it to their own needs.

This review provides the most recent, accurate and comprehensive summary of the literature related to coach knowledge in talent identification, while critically appraising the quality of the literature. The review followed the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guidelines and was first published online in May 2019.

See the Publication Update at the end of this chapter for updated results up to April 2020.

Publication Update

An updated literature search conducted in May 2020 for articles published up to April 30th 2020 revealed 183 new articles that met the search criteria (57 duplicates removed). Title and abstract screening left 32 articles for a full-text review, after which two articles remained for inclusion in this chapter. These two articles were Rosevear and Cassidy (2019) and Winter, O'Brien and Collins (2019). Including these articles did not change the findings of this investigation, so the text of this chapter remains unchanged from the published version. The new articles are included in the tables for this chapter (Table 3, Appendix 1 and Appendix 2) and are indicated by an asterisk (*).

Table 3

Results of Critical Appraisal

CITATION	Study purpose (/1)	Literature (/1)	Study Design (/3)	Sampling (/3)	Descriptive Clarity (/4)	Procedural Rigour (/1)	Analytical Rigour (/2)	Auditability (/2)	Theoretical Connection (/1)	Trustworthiness (/4)	Conclusions /Implications (/2)	Total (/24)	Quality
Christensen	1	1	3	2	2	1	2	2	1	4	2	21	Good
Cupples & O'Connor	1	1	1	2	0	1	2	2	1	3	2	16	Good
Ellingsen & Danielsen	1	1	3	2	2	1	2	1	1	4	2	20	Good
Ferreira Celestino et al	1	1	1	1	0	0	1	1	1	3	2	12	Moderate
Goncalves et al	1	0	1	1	0	1	1	1	1	2	2	11	Poor
Holt & Dunn	1	1	2	1	2	1	2	2	1	4	2	19	Good
Johansson & Fahlén	1	1	2	1	1	1	2	1	1	2	2	15	Moderate
Johnson <i>et al</i>	1	1	1	3	4	1	2	2	1	4	2	22	Good
Jokuschies <i>et al</i>	1	1	2	2	2	1	2	2	1	4	2	20	Good
Lund & Söderström	1	1	2	2	0	1	2	1	1	1	2	14	Moderate
Milistetd <i>et al</i>	1	1	0	2	2	1	0	0	1	2	2	12	Moderate
Miller <i>et al</i>	1	1	2	2	3	1	1	1	1	4	2	19	Good
Mills et al	1	1	2	2	1	1	2	1	1	4	2	18	Good
Rosevar & Cassidy*	1	1	2	2	1	1	2	1	1	4	2	18	Good
Vrljic & Mallett	1	1	1	1	1	1	2	1	1	4	2	16	Good
Winter <i>et</i> al*	1	1	3	2	2	1	2	2	1	4	2	21	Good

*Included for thesis update

CHAPTER FOUR Understanding 'Gut Instinct'

This chapter has been published as the following peer-reviewed journal article:

Roberts, A.H., Greenwood, D., Humberstone, C., Iredale, F., Stanley, M., & Raynor, A. (2020). Understanding the 'gut instinct' of expert coaches during talent identification. *Journal of Sports Sciences*. DOI: 10.1080/02640414.2020.1823083

Foreword

In Chapter Three, it was demonstrated that there is a lack of in-depth investigations into how coaches make decisions regarding an athlete's talent. It quickly became apparent that in order to answer the main question of this thesis – namely, 'what is 'gut instinct?' – qualitative methods would be most appropriate. Given the value of experiential knowledge in co-creating empirical research questions, a constructivist approach was used in this study to allow the coaches involved to help build our understanding of the talent identification process. This chapter contains the findings from interviews with twenty-four elite coaches from fourteen different countries, across three Olympic combat sport disciplines (boxing, judo and taekwondo).

CHAPTER FIVE

A Case Study in Talent Identification

This chapter is currently under review with a peer-reviewed journal:

Roberts, A.H., Humberstone, C., Greenwood, D., Stanley, M., & Raynor, A. *The Coach's Eye:* An elite coach's decision-making process during talent identification [Manuscript Under Review].

Foreword

This thesis has thus far developed a theoretical understanding of how coaches believe they identify talent. Chapter Three demonstrated that when coaches identify talent, they are driven by implicit, instinctual factors that they struggle to articulate. Chapter Four explored this concept of 'gut instinct' by speaking directly to coaches and unpacking the factors that underlie their seemingly instinctual decisions. The finding that coaches rely on their experience and ability to identify talent, and time and context in order to do so confidently, while not appearing novel, is one that has yet to be expressed in the literature.

During Chapter Four, coaches were asked about hypothetical situations, or to reflect on their prior experience in order to answer the interview questions. While there are many ways in which talent identification occurs in situ, two very common methods are a single coach making decisions, or a group of coaches attempting to reach consensus. The following two chapters explore whether the findings from Chapter Four are applicable in both of these real-life talent identification environments. Chapter Five contains a case study of a single coach making identification decisions over an extended period and, consequently, in several different contexts. Chapter Six contains an investigation of multiple coaches (therefore different experiences and abilities) in a constrained time period (four days) in the same context (junior talent camp). Essentially, these two chapters investigated how the manipulation of constraints (time and context – task and environment; individual – experience and ability) affects coaches' ability to identify talent.
CHAPTER SIX

Reliability of Coach Ratings over a Four-Day Training Camp

This chapter has been published as the following peer-reviewed journal article:

Roberts, A.H., Greenwood, D., Humberstone, C., Raynor, A.J. (2020). Pilot study on the reliability of the coach's eye: Identifying talent throughout a 4-day cadet judo camp. *Frontiers in Sports and Active Living*, *2*: 596369. doi: 10.3389/fspor.2020.596369

The paper formatting, grammar and headings have been modified to maintain a consistent thesis style. References are included at the end of this thesis.

Foreword

The previous chapters in this thesis have demonstrated how both time and context can influence the perception of talent, and that many expert coaches believe that experience is integral to being able to identify talent appropriately. In particular, Chapter Five highlighted how a coach's opinion of an athlete's talent can change significantly over time, and how important context is when making these decisions. It follows that the next question relates to the individual differences between coaches. If a group of coaches with different amounts of experience are tasked with identifying athletic talent in a single group of athletes over a set period of time in the same context, will they identify the same athletes as being talented? Therefore, this chapter is an in-situ investigation into the inter-coach reliability of coach ratings throughout a four-day judo training camp for cadet (under age 17) athletes.

CHAPTER SEVEN

Discussion: Perceiving Talent and its Identification Through the Coaches' Eye

The two models and their associated definitions presented in this chapter have been presented and accepted for presentation at the following conferences:

- **Roberts, A.H.**, Raynor, A., Greenwood, D. (2020, March). *A new conceptualisation of sporting talent: The role of the coach*. [Poster presentation]. Exercise and Sports Science Australia (ESSA) Conference, Perth, Australia. (Delayed due to COVID-19); to be presented in 2021.
- **Roberts, A.H.**, Raynor, A., Greenwood, D., Humberstone, C., Iredale, F. (2019b, November). *Proposition of a new conceptualization of the longitudinal talent identification process.* [Conference presentation]. 3rd Scientific Conference on Motor Skill Acquisition, Lohja, Finland.

The first model, the *Coach-Informed Talent Identification Process*, and the associated definitions were first presented at the Scientific Conference on Motor Skill Acquisition in 2019 to an audience of researchers and practitioners. With a clear goal to enhance the empirical understanding of coaches' practical knowledge of the talent identification process, follow-up discussions based on this presentation resulted in the incorporation of additional content and feedback to improve the model's clarity.

The second model, a novel representation of the *Coaches' Eye During Talent Identification* was created to help interpret and explore the underpinning mechanisms that lead coaches to make judgments and decisions in practical settings, specifically, how they identify talent. This thesis aimed to explore the *coaches' eye* in talent identification, examining the decision-making process of elite coaches during this process through the use of combat sports as a task vehicle. Through the findings of this program of research, it has become evident that existing models and terminology are insufficient at capturing the nuances of how coaches perceive and perform talent identification. It also became apparent that although coaches and researchers use the term coaches' eye to describe how coaches make subjective decisions, there have been no attempts in the literature to define nor model this term and the associated processes.

This chapter draws on the findings previously presented in this thesis to introduce two new models. The first model, the *Coach-Informed Talent Identification Process*, is a diagrammatical representation of *what* coaches do as part of this process. The second model is a novel representation of the *Coaches' Eye during Talent Identification*, describing *how* coaches make forecasts and selections as part of this process. Together these unique contributions to the field will address the misalignment in terminology and provide an evidence-based perspective of the talent identification process as viewed through the coaches' eye.

A Model of Talent Identification in High-Performance Combat Sports

The literature reviewed in Chapters Two and Three indicated that talent identification is often conceptualised as a singular event, capturing a 'snapshot' of an athlete's *current* abilities. However, this interpretation is more representative of the term (talent) selection; the decision made to include or exclude an athlete from a given team, squad or competition is often based on their performance in a single event. Using 'identification' and 'selection' interchangeably in the literature has often incorrectly positioned talent identification as a singular point in an athlete's career, resulting in an over-reliance on current performance and/or discrete physical and physiological capacities when considering long-term athlete potential. A lack of clear definitions and inconsistent

terminology makes comparing and building on existing literature difficult, complicating the process of transferring this knowledge to external stakeholders.

Within existing literature, the coach has rarely been a focus during talent identification research. Instead, research tends to focus on athlete qualities, comparing identified/selected athletes with those who were not. This athlete-centric focus is not restricted to talent identification research and is a limitation in many areas of sport science (Araújo & Davids, 2011). Position talent identification as a dynamic system in which athletes are identified (rather than focusing on the athlete alone), the coach's role and impact within the process is emphasised. By better understanding how coaches perceive and perform talent identification, we can then begin developing this ability in current and future coaches while influencing organisational policies and practices to best support coaches in their judgment and decision-making process.

Recent conceptualisations of talent identification include detection, identification, confirmation, development and (de)selection as discrete stages within a larger process, often also referred to as 'identification' (see Figure 1, also Williams & Reilly, 2000). However, few, if any of these terms have been defined or used based on the understanding, knowledge and usage by coaches (see Table 1 for existing definitions). This, combined with a lack of models based on how coaches perform talent identification in situ, creates the need for a new model of talent identification to be developed.

The *Coach-Informed Talent Identification Process* (Figure 9) has been developed by drawing on the experiential knowledge of expert, high-level combat sport coaches and has been refined based on their feedback. This is in comparison to previous models, which have typically been developed for use in youth team sports and have been created without direct coach input or feedback. The new model characterises how athletes move through the talent identification process, and the role coaches play in that process, as perceived by expert high-performance coaches. A key difference between this and previous models is that the *Coach*-

Informed Talent Identification Process positions talent identification as an overarching process that continues throughout an athlete's development. The following sections describe each aspect of this new model and their key differences from previous conceptualisations of talent identification.

Figure 5



The Coach-Informed Talent Identification Process (first presented in Roberts et al., 2019b)

The dashed box represents the talent identification process. Directional arrows represent athlete movement through the process. The loop formed between forecasting and confirmation represents the coach judgment process, occurring in combination with athlete development. Selections are made at multiple levels and times, and both selection and detection are active decision points. Participation athletes (those who engage in the sport recreationally) can move into the talent identification process through several mechanisms. Additional detail is included in the accompanying text.

Talent, Identification, Judgment, and Decision-Making

Chapters Four and Five demonstrated that coaches perceive talent identification as the process which allows them to 'see' the skills, qualities, attributes or traits of an athlete (and relevant combinations) that indicate (or contraindicate) future high-level performance – specifically, predictors of Olympic medallist potential. This understanding aligns with the recent definition of talent put forward by Till and Baker (2020), which proposed that the purpose of talent identification is to identify the presence (or absence) of the skills, qualities, or attributes predictive of high-level performance.

Coaches view the identification of talent as an ongoing phenomenon, one which can be obscured by day-to-day events (noise) impacting on athletes such as injury, fatigue, and current performance levels. They described talent identification as a process – a series of steps repeated at different times throughout an athlete's development. Coaches make ongoing forecasts of an athlete's potential talent and confirm the accuracy of these judgments as athletes achieve, or do not achieve, the forecast outcomes. Unlike many coaching decisions in which the outcome of a decision or judgment is known very quickly, it can take years for a coach to confirm the accuracy of their forecasts and resulting selections, thus verifying or refuting their identifications.

The differentiation between 'identification', 'forecasting' and 'selection' is in response to the findings of Chapters Four and Five of this thesis. The interviews conducted in these chapters highlight the importance of terminology that is consistent in use and understanding across coaches, with all of the interviewed coaches indicating a distinction between these terms. The coach from Chapter Five articulated a significant difference in his thought processes between these three stages. He viewed forecasting an athlete's talent as "like the weather" and distinctly different from selection decisions; with the broader talent identification process drawing on the information gathered through forecasting, confirmation, and where relevant, selections (which are sometimes made by others) and

competition results. Differentiating between forecasting and selection can be articulated by describing forecasting as a *judgment* and selection as a *decision*.

The distinction between judgment and decision is important, as it captures how coaches conceptualise the difference between *forecasting* and *selection* during talent identification. Empirical literature describes judgments as the prediction of the likelihood of a given occurrence and the subjective value of said occurrence, formed from the combination of personal beliefs and values with one's knowledge and experiences (i.e. the coaches' individual constraints). Judgments guide decisions made in the absence of complete data or an obvious decision path. In contrast, decisions are deliberate actions or behaviours (with associated consequences) which are often guided by, but not beholden to, judgments (Hastie, 2001; Jacklin, Sevdalis, Darzi, & Vincent, 2009; Likierman, 2020; Tichy & Bennis, 2007). A key differentiator between a judgment and a decision, outside of 'making a choice', is context. Judgments can be, and often are, made without (much) context, particularly as they are often based on a belief or value system (i.e. that certain traits are 'good' or 'bad') (Hastie, 2001). On the other hand, decisions are (or should be) made within context, as this can provide additional information to allow scaling, or calibration, of the decision to the environment in which it is being made. Essentially, "good decisions are those that effectively choose means that are available in the given circumstances to achieve the decision-maker's goals" (Hastie, 2001). In the Coach-Informed Talent Identification Process, talent forecasting is the process of judging an athlete's long-term potential; in comparison to the singular act (decision) occurring during selection. In this way, selection and forecasting are both steps within the talent identification rather than outcomes.

Forecasting and Confirmation

The findings from the case study (Chapter Five) describe how an expert coach forecasts, or judges, each athlete's potential future ability, with limited information. Early in the talent identification process, the coach formed opinions of the athletes, but emphasised that their initial judgments "don't mean much yet".

The term 'forecast' implies that the prediction is informed by a limited amount of available information, with an inherent level of uncertainty as to the outcome. In fact, "uncertainty is an essential and non-negotiable part of a forecast" (Silver, 2015). Therefore, talent forecasting is the ongoing, subjective judgment of an athlete's future performance capabilities based on a coach's observations of an athlete, irrespective of the time between forecast and predicted outcome. As in weather forecasting, the closer to an event that a forecast is made, the more likely it is to be correct. Yet, as emphasised by the coach from Chapter Five, just because "all the conditions are right; it [still] may not storm".

When it comes to forecasting talent for long-term performance (e.g. four or eight years in the future), there are many uncertainties associated with the forecast which can be related to the interactions between the different aspects of the dynamic system. The further away from an event a prediction is made (e.g. performance at the Olympic Games in eight years), the greater the uncertainty; as the number of assumptions about future interacting constraints and their influence on athlete performance increases; as a result forecast accuracy decreases (Silver, 2015).

Coaches must be aware of changes to task and environmental constraints over time and the more traditional focus on changes in the individual athlete. For example, changes in demands within a sport have a relatively slow and predictable evolution as athletes and coaches adapt to competition constraints. There has been a steady increase in the number of distance strikes attempted (and landed) per minute in Ultimate Fighting Championship (professional mixed martial arts) bouts between 2000 and 2014 is a simple example (James et al., 2019). However, sports can also exhibit more sudden changes to either their rules or scoring system, which in turn creates rapid changes in gameplay. These changes can affect the performance capacity of athletes already involved at a high level, as well as those forecast to do so. For instance, a recent rule-change in taekwondo scoring criteria removed the incentive to 'fence', a defensive manoeuvre using legs and encouraged harder kicks by competitors. This change resulted in many athletes no longer being able to compete to the same level as previously under the old rules, as both their physical attributes and skill set were no longer suited to the new competition context. As indicated by each taekwondo coach interviewed as part of Chapter Four, this rule change significantly impacted the athletes they had identified. Previously, coaches had identified the body type of 'tall, longlimbed and flexible' as the most ideal for taekwondo; however, the rule change meant that those athletes were now likely to be beaten by shorter, stronger athletes with lower centres of gravity. This kind of uncertainty is inherent in long-term forecasts and can be described as scenario uncertainty (Silver, 2015), with the accuracy of the coach's forecast decreasing over time. The closer to the event that a change occurs in the task demands or the environment, the greater the chance that these changes will significantly impact athlete performance and, subsequently, the accuracy of early forecasts.

Confirmation is the period between a forecast being made and the forecast event occurring. According to the coaches in Chapters Four and Five, this is typically an informal, ad-hoc exercise, providing coaches with the opportunity to evaluate their forecast against the outcome. The coaches who participated in this research indicated that forecasting and confirmation cycle occurs continuously throughout an athlete's career. Confirmation is an integral part of both short-term (e.g. how long it will take an athlete to master a new skill) and long-term (e.g. an athlete's medal potential several Olympic cycles into the future) forecasts. Regardless of the length of the forecast, confirmation is an essential step in the process as it allows coaches to reflect upon the information used to make the forecast and refine their *coaches' eye* for the future. Confirmation can also be a formalised by having a set period of time used to verify selection decisions. An example is the 'apprenticeship phase' employed by UK sport in many of their talent identification programs (Vaeyens et al., 2008); where athletes detected through 'Sporting Giants' (and other similar 'talent search' programs) trained in a high-performance environment before undergoing national squad selection. In these three months, athletes were exposed to high-performance training demands and fast-tracked their skill development, while coaches were able to get to know the athletes and observe their progression before offering them a firm place on the team.

Selection

Evidence and insight gained from the coach in Chapter 5 indicated that he used his international coaching experiences to forecast the future potential of the Australian athlete. When doing this within the broader international context that he was very familiar with, he stated that "I do not believe that many of the Australian athletes belong in a high-performance training squad". However, this forecast did not mean that he did not select any athletes to join the training squad. When he was required to select athletes for the training hub and subsequent travel teams for competitions, his decisions were based not solely on his judgment of their talent. The coach emphasised that contextual factors relevant to the specific selection impacted his decision-making. Factors such as the athletes' fit within the broader squad, their existing training environment, NSO policy, and his ability to develop the athletes further all played significant roles in his selection decisions. Thus, the context forced him to select athletes, sometimes against his judgment of their talent and potential future success.

One of the key differences between the proposed model and prior iterations is the inclusion of (talent) selection as a part of, but not synonymous with, talent identification. In this context, selection can be defined as per Williams and Reilly: the "process of identifying players at various stages who demonstrate the prerequisite levels of performance for

inclusion in a given squad or team" (Williams & Reilly, 2000, p. 658; see also Table 1). It is a singular point of time when an athlete is selected (or de-selected) for involvement in a specific team, squad or competition, be it a weekly team selection or an annual selection event for a national squad. Once (de)selection has occurred, athletes continue on their developmental pathway at the appropriate level.

Development, Participation and Detection

The *Coach-Informed Talent Identification Process* includes development, participation and detection as separate but essential elements in any encompassing model of talent identification. As this thesis gathered information from high-performance coaches related explicitly to the forecasting and selecting of talented athletes, these additional stages were beyond the scope of the current research. They are described briefly in Chapter 1 (see Table 1) and are reiterated below with a rationale for their positioning within the current model.

In the proposed model, athletes enter a recognised, structured talent development environment through natural progression (continued participation and the associated improvement in the target sport) or detection (non-participants are invited into a structured development environment through testing or screening). Here, athletes develop in deliberately designed environments while coaches continually forecast and confirm predictions of their talent. Talent development is inextricably linked to talent identification, as the overall success of each process requires both the right athletes and the correct implementation of a talent development program. The distinction between talent identification and talent development was starkly highlighted by several coaches who could identify athletes despite the lack of organised developmental pathways in their respective countries. This sentiment informed the placement of talent development within this model as not a step, but a separate process that runs in parallel to talent identification, punctuated by selections of varying importance and consequence. Within the *Coach-Informed Talent*

Identification Process, the specific structured talent development environment will be different depending on the system and sport to which the model is being applied.

It could be argued that entry into a development system requires some kind of identification to occur (c.f. Williams and Reilly, 2000; Romann et al., 2017; Sieghartsleitner et al., 2019), yet there is significant anecdotal evidence of athletes 'qualifying' or otherwise entering development programs based on results (i.e. policy) rather than through selection made by a coach (i.e. a decision). For example, combat coaches, particularly in Australia, must operate within the policies of their NSO selection processes. Several of the coaches interviewed in Chapter Four lamented existing results-based selection policies which are used to place current high performing athletes into squads, camps or teams. Due to limited squad sizes in addition to the need to select training partners for the most promising athletes, these selections are often made at the expense of alternative athletes whom the coaches have forecast to be more successful. This disconnect between results and talent forecasts was evidenced in Chapter Five when the coach spoke of being 'required' to select certain athletes despite their perceived lack of talent.

In this model, participation refers to those athletes who participate in the sport at a community level. These athletes may remain in the participation space if they lack the desire, ability or opportunity to move into more formal athletic development pathways, such as high-level club teams, academies or state/regional development programs. Participation deliberately sits outside of talent identification in the *Coach-Informed Talent Identification Process*; according to the coaches who reviewed this model, many combat sport participants are 'non-competitive'. These 'non-competitive' athletes participate in the sport for fun and fitness with no desire to ever compete at a higher level; thus it is "not worth even considering identification with them" (Coach 16). However, it was deemed important to include participation in this model because it is essential to understand the various backgrounds that athletes may have when attempting to understand the entire athlete pathway. The

forecasting and subsequent selection of athletes can be influenced by the mechanism of entry to the talent identification process; therefore coaches believe it is important to know whether athletes have progressed through participation pathways (natural selection), or were detected into the sport. Talent detection refers to the mechanism through which athletes, often non-participants, are 'detected' as having the (usually physical) qualities that are advantageous within the target sport. Athletes may be detected through an official program (such as the AIS's Sports Draft program [Gul, 2016]) or unofficially, such as when a coach sees an athlete participating in another sport and invites them to join their team or squad.

Summary

The *Coach-Informed Talent Identification Process* is an updated conceptualisation of talent identification with the steps involved in the process based explicitly on the perceptions of elite coaches. This new model was developed from the results of this thesis and refined by feedback from applied practitioners, researchers and coaches. It has refined terminology, particularly around the concepts of 'forecasting' being continuous while 'selection' is a discrete event, both of which occur within the broader talent identification process. This model also captures the non-linearity of athlete development, with athletes continuously moving throughout the process at different rates as coaches repeatedly forecast and confirm their talent. An athlete's development is punctuated by selections, the point at which a coach makes an active decision to include or exclude the athlete from a specific team, squad or competition. This model can be used by researchers and practitioners alike to understand the role and impact of subjective coach judgments and decisions within talent identification.

The Coaches' Eye during Talent Identification

There has been little investigation into the decision-making process of coaches while identifying talent. Within the literature, this process and its subjectivity have been referred to as the *coaches' eye* (Romann et al., 2017; Schorer et al., 2020) and coaches are often used as the 'gold standard' within talent identification research. Despite this, there has been little investigation into coach judgment and decision-making, or whether the *coaches' eye* is reliable between coaches. Chapter Four of this thesis showed that international-calibre combat sport coaches follow the same process when identifying talent (as shown in Figure 9). However, Chapter Six demonstrated that coaches do not necessarily identify the same athlete(s) as being talented. Within a dynamic sporting context, the coach, the athlete, and their complex interactions must be understood in order to attempt to understand the organisation of the system as a whole (Rothwell et al., 2020); i.e., both athlete and coach interact with and contribute to the outcomes of talent identification.

The second part of this chapter introduces an original model and understanding of the *coaches' eye*, specifically during talent identification (Figure 10). This practical representation of coach judgment and decision-making is based on the data presented in Chapters Four, Five and Six of this thesis in conjunction with appropriate literature. The model demonstrates how a coach's constraints influence their talent forecasting judgments and selection decisions by exploring talent identification through the lens of ecological dynamics. This model's key principle is the influence of each individual coaches experience and ability, and how these shape perception of athlete talent. The process of talent identification is then viewed through this lens of coach traits in their role as the 'arbiter' of talent. It was highlighted in Chapter Five that practically this results in a matching-type process, where individual athlete capabilities and the coach capacities influence the process.

A major finding of this thesis was that coaches rely on their experience and expertise to decide which athletes they "want to see more of" when considering the process of

identifying performance potential. When probing this concept, it was determined that the *coaches' eye* is how coaches make their initial assessment of athletes ("catches my eye" – Coach 4); the lens through which the coach interpreted the available information given the context of the judgment or decision made. A coach's attunement to information is dependent on their experience, the time available, the context of the identification, and their coaching abilities when identifying talent, with each coach appearing to attune to different specifying information.

The *Coaches' Eye* is the lens through which the coach integrates their knowledge and experience to form a forecast (judgment) of an athlete's talent. When appropriate, this forecast becomes a selection decision, at which time the coach actively attunes to the context, their coaching abilities, and their knowledge of the athlete. As evidenced in the case study (Chapter Five) and earlier in this chapter, this differentiation is important as coaches do not always select the athletes they have forecast to be talented. Coaches will make their selections based on the current context, choosing the athlete that affords them the best opportunity to achieve their goal or key performance indicator for the context of the selection, for example, short-term performance outcomes, or long-term development goals.

Figure 6

A Model of the Coaches' Eye during Talent Identification



The solid arrows represent the judgment and decision-making process of coaches during talent identification. The circles indicate outcomes, with the dotted lines indicating the sources of information used to inform the *Coaches' Eye*.

Coach-Athlete Dyad

When viewing talent identification through an ecological lens, the framework highlights how the coach and the athlete interact as part of a broader dynamic system. The coach-athlete sub-system represents a critical component of the larger performance system. The constraints which affect each sub-system also affect the broader context (i.e. talent identification). The individual constraints of each coach and athlete are important as, highlighted by Chapter Five, by changing either the coach or athlete involved, the outcomes of the identification (forecasts and/or selections made) will change. The coach and athlete are represented in the model of the *Coaches' Eye during Talent Identification* as triangles to indicate the inherent interactions between the individual, environmental and task constraints and their influence on behaviours and decisions.

The double-headed arrow between the coach and the athlete indicates the symbiotic relationship between the two sub-systems within talent identification. Each is continually perceiving the other and adjusting their actions or behaviours based on these perceptions. The findings of Chapter Five demonstrate that the interrelationship between coach and athlete impacts the ongoing forecasting process. Through his continual interactions with the athletes in a variety of contexts, the coach's perception of the athletes' potential changed over time. During this time, the coach observed the athletes and their behaviour, but his interactions influenced the behaviour of the athletes, which in turn further influenced his observations and interpretations.

Coaches draw on their experience with athletes of varying ability to understand the athlete and form judgments about the athlete's current and future performance capacity. These forecasts (judgments) are heavily influenced by the coach's constraints, particularly their knowledge and experience. Their experience influences how they interpret the information they perceive about an athlete. This was evident in the findings of Chapter Six, as the coaches in this study interpreted the athlete information differently, making very different forecasts of athlete potential. As these coaches were interacting with the same athletes under the same task and environmental constraints, the apparent difference in their interpretation of the athletes' talent lies in the coaches' individual constraints. While there are many contributions to a coach's individual constraints, the coaches interviewed in Chapters Four and Five indicated that they interpret athlete information in different ways based on their experience, as coaches who have seen a wider variety of athletes tended to be more open in their beliefs of "what it takes" (Coach 22) to be successful in the long term. Experienced coaches can see "different paths" (Coach 4) to success, viewing elite performance as a 'threshold' that athletes need to reach, regardless of the combination of traits led to their achievements.

Forecasting, Selection and Attunement

The model of *The Coaches' Eye during Talent Identification* positions forecasting and selection as outcomes of the talent identification process. The key difference between the two outcomes is the coach's attunement to specific information used to inform their context-specific selection. As the *coaches' eye* is the lens through which coaches make forecasting judgments and selection decisions, this lens is focused by the purpose of the judgment or decision. As shown in the *Coach-Informed Talent Identification Process*, forecasting occurs when a judgment is made about an athlete's future performance potential; however, there is no active (de)selection as part of this judgment. Accordingly, in the model of the *Coaches' Eye in Talent Development*, forecasts are made based on the individual coach's interpretation of the athlete's future potential with reference to the coach's experiential knowledge of athlete progression within their sport.

Aligned with the *Coach-Informed Talent Identification Process*, selection is the active decision point at which an athlete is (de)selected for a given opportunity (squad or competition). When a coach makes a selection, their initial forecast is attuned to the specifying variables that influence their decision, specifically the context of the selection,

how well they know the athlete, and the coach's perceived ability to develop the athlete within the timelines available. Attunement creates an inherent bias when identifying talent, as coaches subconsciously rate an athlete's future ability in terms of whether the coach in question can help them progress. This was demonstrated throughout the current thesis, for example when Coach 14 (Chapter 4) spoke about his prowess in developing technique and stated that he often selects athletes that are lacking technical proficiency because he "can give them that". There was an apparent connection between the traits that coaches believe are (un)important and those they are (in)capable of developing in their athletes. Coaches do not necessarily attune to information that is more or less important, or in more or less effective ways, but rather attune to different information sources based on their experience and coaching ability. The more expert coaches within the cohort interviewed in Chapter Four, and the coach in Chapter Five, appeared to be aware of their strengths and weaknesses as a coach and how this could limit and bias their identification process, and impact athlete development trajectories. The coach's expertise (rather than experience) allows them to account for these biases as and when it is appropriate during the talent identification process. For example, it was shown during Chapter Five that if the coach's task was to select athletes to win at a competition next month, he would select the current best-performing athletes, regardless of his capacity to develop them over the long-term. Conversely, when he is selecting athletes to join a development centre, his focus was on selecting those athletes he would improve in the timeframe available.

The reliability of coach judgments or forecasts was shown to be very low during the four-day camp detailed in Chapter Six. Traditionally, the disparity in perception may have been explained due to the differing levels of experience among the coaches; however, the evidence and model presented in this section provide an alternative view. The novel model of the *Coaches' Eye in Talent Identification* (Figure 10) suggests that the *coaches' eye* differs between coaches due to the different constraints of each coach-athlete dyad. The *coaches'*

eye allows for the rapid integration of various complex, interacting pieces of information. If used correctly, this skill enables coaches to identify and select athletes by matching their expertise to the athlete's capabilities, thus optimising outcomes for the athletes they can develop.

Both the coach and the athlete exist within the dynamic talent identification system. As soon as a critical component of any system is changed, the whole system changes. Therefore, the coach needs to be a critical consideration when planning and executing talent identification strategies within sport.

Practical Implications

The models and terminology discussed in this chapter and other findings in this research program will provide coaches and national sporting organisations with a greater insight into the underpinning factors that enable coaches to identify talent. The evidence indicates that creating a standardised talent identification procedure or testing battery without input from coaches, particularly those working with the identified athletes, will not deliver the desired results either empirically or practically. The following section details practical implications for NSOs in their design of talent identification programs and policies, and recommendations for coaches and coach educators to enhance the effectiveness of the *coaches' eye*.

Practical Implications for National Sporting Organisations

A key limitation identified by coaches in existing talent identification processes is the absence of coach input on athlete selection policies. The lack of coach input on athlete selection policies has, according to the coaches, resulted in negative consequences for both short- and long-term athlete development. For example, placing athletes who forecast poorly in development squads based on their current results often occurs at the expense of athletes who forecast strongly but are deselected due to their current performance level. While current performance must be a practical consideration when making selections and

cannot (nor should it be) completely discounted, NSOs should work with coaches to ensure an appropriate blend between consideration of current performance and future potential. National sporting organisations should include coaches from all levels of the pathway when developing new identification protocols. This thesis has demonstrated that different coaches will identify different athletes, so choosing the right coaches to identify talent and ensuring a variety of coaches are involved in the process is essential. The inclusion of multiple coaches is important, to limit the bias inherent in experiential knowledge and to ensure expertise from across the pathway. It is also important that there is a balance between coach and NSO input as coach contracts are typically short-term; thus, NSOs need to ensure long-term consistency for their athletes.

National sporting organisations need to ensure that they provide coaches with the necessary information to make the best possible decision. Specifically, NSOs need to ensure that the coach is explicitly aware of their key performance indicators related to the selection – is their goal to get results immediately, at the end of the next Olympic cycle or both? This will provide contextual information to inform the attunement of the decision-making process based on short- or long-term forecasts. This contextual understanding facilitates the selection balance required between short- and long-term performance goals, the time the athletes have to improve, the different skills deemed improvable by the coach, and the athlete's current capabilities. In addition, NSOs should strive to ensure that the coaches performing the selection are those who will be responsible for the athletes' development, and (where practicable) to provide the coach with sufficient time to determine the athlete's potential relative to their constraints. Talent identification camps need to run for as long as practicable (several days at least) in order for coaches to gain the fullest possible understanding of the athletes. Where possible, talent identification should occur for several months or years, keeping athletes in the system to enable the best possible forecasts to be

A final recommendation for NSOs is to consider the coach who is performing the identification with the same level of scrutiny as they do athletes they identify. This research has demonstrated that coaches' forecasts and selections are individual and based on their inherent traits, but that is not to say that coaches should not identify, forecast, and select talent. These differences are what afford the coaches the opportunity to develop specific athletes. When the career of both the coach and the athlete depends on athlete outcomes, coaches should be able to select the athletes for whom they have forecasted the highest level of success. National sporting organisations should involve coaches when planning talent identification to ensure that the coach can work with the athletes they can develop. In turn, athletes should be placed with a coach who will best develop their potential.

Practical Implications for Coaches and Coach Educators

This research has indicated that expert coaches are more aware of their own coaching limitations than their less experienced counterparts. Specifically, this allows them to compensate for their weaknesses and, as a result, they are afforded to select and subsequently develop a wider range of athletes. Coach education related to talent identification should include information about which athlete attributes are trainable, the timelines required to develop these attributes, and what resources (both professional development and sport science practitioners) can assist in this development. This will ensure that coaches are not (de)selecting athletes based on erroneous assumptions (e.g. 'mental toughness cannot be improved') or a lack of understanding of what is available within the broader development system (e.g. access to psychologists). At the same time, coaches need to feel confident in the future performance of the athletes that they select. By developing coach education that focuses on all facets of athlete development, coaches will be better able to develop and support athletes in their squads, regardless of the mechanism through which an athlete was placed in their team.

CHAPTER EIGHT Summary and Future Directions

Traditionally, talent identification research has focused on snapshot physical and/or psychological testing to objectively measure predictors of athletic performance. Often, coaches' subjective opinions are used as the yardstick against which the efficacy of these objective measures are evaluated (Roberts et al., 2019a). When coaches contribute to talent evaluations, for example through technical and/or tactical assessment during match play, the coaches' eye (the lens through which athlete performance is evaluated) has been applied with little consideration of the validity or reliability of coaches' perceptions of talent (Schorer et al., 2020). Recently, Schorer and colleagues have published several articles that have longitudinally tracked coaches' predictions of athlete success in European handball, demonstrating that the coaches' eye may be a valid mechanism for talent identification (Schorer et al., 2020, 2017). However, there is still a limited understanding of the reliability or the underpinning processes of this useful tool. Fundamental questions exist as to how coaches identify talent, particularly in the combat sport disciplines in which the subjective nature of competition performance increases the uncertainty in talent evaluation processes. Basic questions include: How do coaches perceive talent identification? What factors underpin their decision-making process? How reliable are coaches in their predictions of future performance? Specifically, what is the *coaches' eye*, and how does it contribute to talent identification?

This thesis's overarching aim was to address the current gap in knowledge related to the *coaches' eye* by exploring the experiential coach knowledge of talent identification, using the Olympic combat sport disciplines of boxing, judo and taekwondo as a task vehicle. A series of four investigations explored the following questions:

- 1) What is currently known about how coaches identify talent? (Chapter Three)
- 2) How do elite combat sport coaches identify talented athletes? (Chapter Four)
- How does an elite combat sport coach make identification and selection decisions in situ? (Chapter Five)
- 4) How reliable are coach ratings of perceived talent? (Chapter Six)

This chapter provides an overview of this thesis, summarizing the theoretical and practical outcomes with an emphasis on how this research can inform coaching practice, NSO strategies and future research.

Summary of Findings

This research has applied the theoretical framework of ecological dynamics in examining coach decision-making during talent identification in combat sports. Beginning with an in-depth examination of the existing empirical research related to talent identification, ecological dynamics, and decision-making, Chapters Two and Three provide an understanding for the reader of the current state of understanding of the coaches' eye during talent identification. As the first exploratory chapter of this thesis, Chapter Three presented a systematic review and meta-synthesis which examined the existing empirical knowledge of coach decision-making during talent identification. This chapter emphasised a trend for utilising experiential coach knowledge in both research and practice while highlighting the dearth of empirical understanding of this knowledge. The consensus among existing research was that coaches appear to make decisions about an athlete's talent primarily based on intuition or 'feel'. The themes of drive and ambition, physical and technical skills, and game intelligence emerged as important underlying factors contributing to the coach's instinctual decisions. The results from this study indicated that coaches rely on their 'gut instinct' to identify athletes, but struggle to articulate the process through which they arrive at these instinctual opinions and decisions.

Chapter Four used semi-structured interviews with elite-level, expert coaches from boxing, judo and taekwondo to explore how these coaches identify talent. This chapter provided one of the first definitions of 'talent' from coaches working in high-performance sport; it also empirically captured the coaches' 'gut instinct' related to talent identification. *Experiential knowledge* (know-how built through years of experience and reflection), *temporal factors* (e.g. the time available to get to know the athlete, the time available for the coach to decide, how far away the target competition is), *athlete context* (athlete performing under differing individual, environmental and task constraints), and *what can be worked with* (i.e. the alignment between a coach's strengths/weaknesses and those of the athlete) were found to be the key considerations for coaches during their decision-making process.

This chapter also demonstrated that different coaches are likely to identify different athletes as being talented. Talent identification is affected by four variables: experience, ability, time and context. These findings supported recent claims from Baker and colleagues that "beliefs about talent matter" (Baker et al., 2018, p. 3), and that the lived experiences of expert coaches affect their selection of athletes (Johnston & Baker, 2020).

Chapter Five detailed an 18-month case study with a single expert, elite-level coach during a longitudinal talent identification and selection process. This allowed for exploration of the impact of two key considerations of the *coaches' eye – time* and *context*. The most significant finding of this chapter was that the context of talent identification is vital, to the point that 'identification' and 'selection' are conceptualised as two different processes with two different intended outcomes. This is an important differentiation as much of the existing talent identification literature uses identification and selection interchangeably or as a single entity ('talent identification and selection'). A lack of differentiation in the literature between 'identified' and 'selected' athletes contributes to misunderstandings among coaches, practitioners, and researchers regarding how decisions are made. The themes identified in Chapter Four (experiential knowledge, time, context and ability) were reinforced through the findings of Chapter Five. The coach examined in the case study considered these four aspects throughout his decision-making process; however, each component's relative importance varied depending on whether the coach was *forecasting* or *selecting* athletes – that is, it was dependent on the context.

Chapter Six highlighted the differences in individual coach perceptions of athlete talent throughout a four-day elite youth judo camp. This novel chapter demonstrated that coaches are not necessarily reliable judges of talent, even when observing the same group of athletes, for the same period, under the same circumstances. Despite the athletes all being high performers for their age group (or perhaps because they were), coaches could not distinguish between those with high levels of 'potential' and those without. By the final day, all coaches agreed on the placement of only two out of the 24 athletes. This finding supports earlier findings in this thesis, which align with coach comments that a significant amount of time is required to 'get to know' athletes and get a 'true' sense of their talent. In the context of real-life talent identification, particularly in youth age groups, it is generally assumed that coaches for the same sport at the same level (i.e. judo cadets) would identify the same groups of athletes as talented. This research has shown that this assumption is incorrect. There is a need for future word investigating the inter-coach reliability of coaches, and the contributing factors to coach reliability during talent identification.

Chapter Seven synthesised the findings of this thesis by presenting two new models to interpret talent identification: *The Coach-Informed Talent Identification Process* and the model of *The Coaches' Eye during Talent Identification*. Both models were developed from elite coach experiential knowledge in high-performance combat sport settings. The *Coach-Informed Talent Identification Process* positions talent identification as an ongoing process instead of as a singular event. It demonstrates the role of the coach at different stages and the cyclical relationship with athlete development. Drawing on and synthesising the findings

of this thesis, the new model of talent identification has been developed based on the experiences and expertise of coaches involved in high-performance combat sports. This chapter also defined a new term ('forecasting') while re-defining other terms (selection; confirmation) to align with the perceptions and word usage of the coaches who operate in this space daily.

The novel description and model of *The Coaches' Eye during Talent Identification* encapsulates the difference between forecasting and selection in terms of the coach's judgment and decision-making. This model accounts for, and highlights to practitioners, the influence within talent identification of individual coach and athletes and how the coach's expertise and ability can impact the results. Coaches forecast athlete talent based on their own experience, and their own constraints influence their selections. This model explains why different coaches are afforded to select different athletes and how context influences these decisions.

The models and related terms will help guide understanding of how coaches both perceive and perform talent identification, driving coach development of their ability to identify talent and ensure that national governing bodies and coaches can collaborate to ensure that the most appropriate athletes are identified and developed. This empirical research will also benefit future studies, providing a framework to understand the coach's role and their position as variables, not constants, during talent identification processes.

Key takeaway messages from these models are that NSOs should ensure that each selection's goals are clearly stated. Where practicable, coaches should be allowed to contribute to squad selections based on their forecasts of athlete talent. Coaches also need to be aware of their own biases, related to their experience and abilities, inherent in their judgments and decisions.

Limitations of this study

This research's primary delimitation was the specificity of the target population – national level or higher coaches of three Olympic combat sports – boxing, judo and taekwondo. This delimitation potentially limits the applicability of the findings to settings other than high-performance combat sports. However, it is anticipated that these results could be applied to talent identification in lower levels of the combat athlete pathway, and it is likely that they will also be generalizable to talent identification in other individual sports. Future research is needed to explore the transferability of these findings to team sport settings.

An additional delimitation was the focus on the identification of Australian athletes in Chapters Five and Six. As mentioned in the introduction to this thesis, combat sports have very low participation rates within Australia, and Australians have not historically been successful in international competitions in these sports. Therefore, the process followed by the coach in Chapter Five may have been different had he been working with a group of athletes whom he perceived as being more talented.

A significant limitation of this program of work is the focus on coaches from Australia and Western Europe. Due to the English-language requirements placed on participation in the interview study (Chapter Four), the research failed to capture coaches' perceptions from other regions, particularly Asia. Given the traditional dominance of Asian countries in many combat sports, the inclusion of Asian coaches may alter the findings of this body of work. The geographical limitations inherent in research being performed by Australian researchers may have also limited the findings, as interviews with coaches were only able to be conducted during a short period of time when the student was physically present in Europe.

Future Directions

This research has provided a greater understanding of the *coaches' eye* in talent identification and the factors that influence the decision-making process. However, there is still significant research needed to fully understand the influence that coaches have on the talent identification process.

An overarching recommendation for future research is for researchers to provide better descriptions of the coach's demographic information, particularly concerning the experience and/or ability of the coach/es in question. This research has made it clear that the term 'expert' is not yet sufficiently defined within the coaching literature, creating research that is difficult and often unhelpful to compare or apply. This thesis has demonstrated that coach decision-making, particularly during talent identification, is subjective and heavily influenced by a coach's experience and expertise. Thus, future researchers should ensure that sufficient information about the coach is provided to contextualise the coach's decision-making.

This research has demonstrated that coaches are an integral part of the talent identification system, and future research should continue to explore the relationship between the coach and the athlete they are identifying. Based on the results of this work, it is suggested that providing coaches with some level of autonomy in athlete selection, along with more explicit goals for selections, will result in improved athlete performance outcomes throughout the next Olympic cycle. Future research should investigate this idea and examine how subjective coach decisions can best be integrated with objective policies to create the best possible outcomes for all concerned.

Many existing studies use the *coach's eye* as the standard for discriminating between athlete potential, without critical consideration of these opinions. This work has clearly shown that coaches cannot be considered as dependent variables in research due to the subjective differences in perceptions of talent. Thus, future research (and interpretation of

existing research) involving coaches in any capacity should be treated with caution, particularly when a single coach is positioned as representative of a larger group or as the standard for comparison of objective testing.

Researchers may also want to consider further investigating how coaches develop their 'eye'. As the coaches in this research viewed the *coaches' eye* as a skill that can be developed over time, it is vital to determine how it can be developed and how we might enhance the process. Ultimately, this research has demonstrated that while the *coaches' eye* is subjective, it appears that that individuality is what affords greater athlete performance outcomes.

Conclusion

This research used ecological dynamics to explore the decision-making that underpins the talent identification process for high-level Olympic combat sport coaches. Mixed methodologies were employed to explore the concept of the *coaches' eye* within talent identification, resulting in the creation of a new model of talent identification and related terminology, including a definition of the *coaches' eye*. This thesis found that while the *coaches' eye* may not be reliable in the statistical sense of the word, the process they follow and the type of information they draw is consistent. Coaches attune to the information they perceive as most relevant in allowing them to forecast an athlete's ability and select the athletes they will be best able to work within the available timeframe.

References

- Abbott, A., Button, C., Pepping, G.J., & Collins, D. (2005). Unnatural selection: Talent identification and development in sport. *Nonlinear Dynamics, Psychology, and Life Sciences*, *9*(1), 61–88.
- Abbott, A., & Collins, D. (2004). Eliminating the dichotomy between theory and practice in talent identification and development: Considering the role of psychology. *Journal of Sports Sciences*, *22*(5), 395–408. https://doi.org/10.1080/02640410410001675324
- Abbott, A., Collins, D., Martindale, R. J. J., & Sowerby, K. (2002). *Talent identification and development: An academic review*. Retrieved from http://researchrepository.napier.ac.uk/2493/
- Abdolmohammadi, M. J., & Shanteau, J. (1992). Personal attributes of expert auditors. Organizational Behavior and Human Decision Processes, 53, 158–172.
- Abraham, A., & Collins, D. (2011). Taking the next step: Ways forward for coaching science. *Quest*, *63*(4), 366–384. https://doi.org/10.1080/00336297.2011.10483687
- Ackerman, P. L. (2014). Nonsense, common sense, and science of expert performance: Talent and individual differences. *Intelligence*, 45(1), 6–17. https://doi.org/10.1016/j.intell.2013.04.009
- Al-Abood, S. A., Bennett, S. J., Hernandez, F. M., Ashford, D., & Davids, K. (2002). Effect of verbal instructions and image size on visual search strategies in basketball free throw shooting. *Journal of Sports Sciences*, 20(3), 271–278. https://doi.org/10.1080/026404102317284817
- Almeida, J., Sarmento, H., Kelly, S., & Travassos, B. (2019). Coach decision-making in Futsal: From preparation to competition. *International Journal of Performance Analysis in Sport*, 19(5), 711–723. https://doi.org/10.1080/24748668.2019.1648717
- Anderson, N. E., Slark, J., & Gott, M. (2019). Unlocking intuition and expertise: using interpretative phenomenological analysis to explore clinical decision making. *Journal of Research in Nursing*, 24(1–2), 88–101. https://doi.org/10.1177/1744987118809528
- Anshel, M. H., & Lidor, R. (2012). Talent detection programs in sport: The questionable use of psychological measures. *Journal of Sport Behavior*, *35*(3), 239–266.

- Araújo, D., & Davids, K. (2011). What exactly is acquired during skill acquisition? *Journal of Consciousness Studies*, *18*(3–4), 7–23.
- Araújo, D., Davids, K., Chow, J. Y., & Passos, P. (2009). The development of decision making skill in sport: An ecological dynamics perspective. In D. Araújo, H. Ripoll, & M. Raab (Eds.), *Perspectives on Cognition and Action in Sport* (pp. 157–169). Nova Science.
- Araújo, D., Davids, K., Diniz, A., Rocha, L., Santos, J. C., Dias, G., & Fernandes, O. (2015).
 Ecological dynamics of continuous and categorical decision-making: The regatta start in sailing. *European Journal of Sport Science*, 15(3), 195–202. https://doi.org/10.1080/17461391.2014.928749
- Araújo, D., Davids, K., & Hristovski, R. (2006). The ecological dynamics of decision making in sport. *Psychology of Sport and Exercise*, 7(6), 653–676. https://doi.org/10.1016/j.psychsport.2006.07.002
- Araújo, D., Davids, K., & Serpa, S. (2005). An ecological approach to expertise effects in decision-making in a simulated sailing regatta. *Psychology of Sport and Exercise*, 6(6), 671–692. https://doi.org/10.1016/j.psychsport.2004.12.003
- Araújo, D., Hristovski, R., Seifert, L., Carvalho, J., & Davids, K. (2017). Ecological cognition:
 Expert decision-making behaviour in sport. *International Review of Sport and Exercise Psychology*, 1–25. https://doi.org/10.1080/1750984X.2017.1349826

Aule, B. R., & Loko, J. (1982). Selection of young athletes. Kehakultuur, 16.

- Australian Olympic Committee. (2019). Australian Olympic Committee. Retrieved January 28, 2019, from http://olympics.com.au/
- Australian Taekwondo. (2018). High Performance Strategic Plan. Retrieved from https://austkd.com.au/uploads/docs/Australian_Taekwondo_High_Performance_Stra tegic_Plan_2018-2020.pdf
- Baars, T. (2011). Experiential science: Towards an integration of implicit and reflected practitioner-expert knowledge in the scientific development of organic farming. *Journal* of Agricultural and Environmental Ethics, 24(6), 601–628. https://doi.org/10.1007/s10806-010-9281-3
- Bailey, A., & MacMahon, C. (2018). Exploring talent identification and recruitment at circus arts training and performance organizations. *High Ability Studies*, 29(2), 213–240. https://doi.org/10.1080/13598139.2018.1518776

- Baker, J., Schorer, J., & Wattie, N. (2018). Compromising talent: Issues in identifying and selecting talent in sport. *Quest*, 70(1), 48–63. https://doi.org/10.1080/00336297.2017.1333438
- Baker, J., & Wattie, N. (2018). Innate talent in sport: Separating myth from reality. *Current Issues in Sport Science*, *3*, 006. https://doi.org/10.15203/CISS
- Baker, J., Wattie, N., & Schorer, J. (2019). A proposed conceptualization of talent in sport: The first step in a long and winding road. *Psychology of Sport and Exercise*, 43, 27–33. https://doi.org/10.1016/j.psychsport.2018.12.016
- Bakken, B. T., & Gilljam, M. (2003). Dynamic intuition in military command and control: why it is important, and how it should be developed. *Cognition, Technology & Work*, 5(3), 197–205. https://doi.org/10.1007/s10111-003-0123-1
- Balagué, N., Hristovski, R., & Vazquez, P. (2008). Ecological dynamics approach to decision making in sport: Training issues. *Ugdymas Kuno Kultura Sportas*, *71*(4), 11–22.
- Balyi, I., & Hamilton, A. (2004). Long-term athlete development: Trainability in childhood and adolescence. Windows of opportunity. Optimal trainability. Victoria, BC: National Coaching Institute British Columbia & Advanced Training and Performance.
- Barker, R. (1968). *Ecological Psychology: Concepts and Methods for Studying the Environment of Human Behaviour*. CA: Stanford University Press.
- Barnson, S. C. (2014). The Authentic Coaching Model : A Grounded Theory of Coaching. International Sport Coaching Journal, 1, 61–74.
- Barreiros, A. N., & Fonseca, A. M. (2012). A Retrospective Analysis of Portuguese Elite Athletes' Involvement in International Competitions. International Journal of Sports Science & Coaching, 7(3), 593–600. https://doi.org/10.1260/1747-9541.7.3.593
- Barsingerhorn, A. D., Zaal, F. T. J., de Poel, H. J., & Pepping, G. J. (2013). Shaping decisions in volleyball: An ecological approach to decision-making in volleyball passing.
 International Journal of Sport Psychology, 44(3), 197–214.
- Baxter, P., & Jack, S. (2008). Qualitative case study methodology: Study design and implementation for novice researchers. *The Qualitative Report*, *13*(4), 544–559.
 Retrieved from http://nsuworks.nova.edu/tqr/vol13/iss4/2

- Bell, R. M., Fann, S. A., Morrison, J. E., & Lisk, J. R. (2011). Determining personal talents and behavioral styles of applicants to surgical training: A new look at an old problem, Part
 I. Journal of Surgical Education, 68(6), 534–541. https://doi.org/10.1016/j.jsurg.2011.05.017
- Bennett, K. J. M., Fransen, J., Scott, B. R., Sanctuary, C. E., Gabbett, T. J., & Dascombe, B. J. (2016). Positional group significantly influences the offensive and defensive skill involvements of junior representative rugby league players during match play. *Journal of Sports Sciences*, 34(16), 1542–1546. https://doi.org/10.1080/02640414.2015.1122206
- Bennett, K. J. M., Novak, A. R., Pluss, M. A., Stevens, C. J., Coutts, A. J., & Fransen, J. (2018).
 The use of small-sided games to assess skill proficiency in youth soccer players: a talent identification tool. *Science and Medicine in Football*, 2(3), 231–236. https://doi.org/10.1080/24733938.2017.1413246
- Bergkamp, T. L. G., Niessen, A. S. M., Den Hartigh, R. J. R., Frencken, W. G. P., & Meijer, R. R.
 (2019). Methodological issues in soccer talent identification research. *Sports Medicine*, 49, 1317–1335. https://doi.org/10.1007/s40279-019-01113-w
- Bompa, T. O. (1985). Talent Identification. *Science Periodical on Research and Technology in Sport*. Ottowa.
- Bondas, T., Hall, E. O., & Wikberg, A. (2014). Metasynthesis in health care research. In Pranee
 Liamputtong (Ed.), *Research Methods in Health* (2nd ed., pp. 280–294). South
 Melbourne, Victoria, Australia: Oxford University Press.
- Boulier, B. L., Stekler, H. O., Coburn, J., & Rankins, T. (2010). Evaluating National Football League draft choices: The passing game. *International Journal of Forecasting*, 26(3), 589–605. https://doi.org/10.1016/j.ijforecast.2009.10.009
- Bowes, I., & Jones, R. L. (2006). Working at the edge of chaos: Understanding coaching as a complex, interpersonal system. *The Sport Psychologist*, 20(2), 235–245. https://doi.org/10.1123/tsp.20.2.235
- Boxing Australia Limited, & Australian Institute of Sport. (2014). *Boxing's Winning Edge 2014-2018*. Caberra, Australia.

- Boxing Canada. (2016). *High performance program selection protocol* (pp. 1–11). pp. 1–11. Retrieved from http://boxingcanada.org/wp-content/uploads/2015/09/HIGH-PERFORMANCE-PROGRAM-SELECTION-PROTOCOL_EnFinal.pdf
- Bradbury, T., & Forsyth, D. (2012). You're in; you're out: selection practices of coaches. Sport, Business and Management: An International Journal, 2(1), 7–20. https://doi.org/10.1108/20426781211207638
- Braun, V., Clarke, V., Hayfield, N., & Terry, G. (2019). Thematic Analysis. In P Liamputtong (Ed.), Handbook of research methods in health social sciences (pp. 843–860). Singapore: Springer.
- Brouwers, J., De Bosscher, V., & Sotiriadou, P. (2012). An examination of the importance of performances in youth and junior competition as an indicator of later success in tennis.
 Sport Management Review, 15(4), 461–475. https://doi.org/10.1016/j.smr.2012.05.002
- Brown, J. (2001). Sports Talent: How to identify and develop outstanding athletes. Human Kinetics.
- Brown, S. G., & Daus, C. S. (2015). The influence of police officers' decision-making style and anger control on responses to work scenarios. *Journal of Applied Research in Memory* and Cognition, 4(3), 294–302. https://doi.org/10.1016/j.jarmac.2015.04.001
- Bruce, L., Farrow, D., Raynor, A., & Mann, D. (2012). But I can't pass that far! The influence of motor skill on decision making. *Psychology of Sport and Exercise*, 13(2), 152–161. https://doi.org/10.1016/j.psychsport.2011.10.005
- Bruineberg, J., & Rietveld, E. (2014). Self-organization, free energy minimization, and optimal grip on a field of affordances. *Frontiers in Human Neuroscience*, *8*(August), 1–14. https://doi.org/10.3389/fnhum.2014.00599
- Brunswik, E. (1956). *Perception and the representative design of psychological experiments*. University of California Press.
- Brymer, E., & Davids, K. (2013). Ecological dynamics as a theoretical framework for development of sustainable behaviours towards the environment. *Environmental Education Research*, 19(1), 45–63. https://doi.org/10.1080/13504622.2012.677416
- Buekers, M., Borry, P., & Rowe, P. (2015). Talent in sports. Some reflections about the search for future champions. *Movement & Sport Sciences*, 88, 3–12. https://doi.org/10.1051/sm/2014002
- Button, C., Seifert, L., Chow, J. Y., Araújo, D., & Davids, K. (2020). *Dynamics in skill acquisition: An ecological dynamics approach*. Champaign, IL: Human Kinetics.
- Cahill, M., Robinson, K., Pettigrew, J., Galvin, R., & Stanley, M. (2018). Qualitative synthesis: A guide to conducting a meta-ethnography. *British Journal of Occupational Therapy*, *81*(3), 129–137. https://doi.org/10.1177/0308022617745016
- Callary, B., Werthner, P., & Trudel, P. (2013). Exploring coaching actions based on developed values: A case study of a female hockey coach. *International Journal of Lifelong Education*, 32(2), 209–229. https://doi.org/10.1080/02601370.2012.733974
- Cañal-Bruland, R., & van der Kamp, J. (2009). Action goals influence action-specific perception. *Psychonomic Bulletin and Review*, 16(6), 1100–1105. https://doi.org/10.3758/PBR.16.6.1100
- Carter, J. E. L. (1970). The somatotypes of athletes A review. *Human Biology*, *42*(4), 535–569.
- Castillo, D., Pérez-González, B., Raya-González, J., Fernández-Luna, Á., Burillo, P., & Lago-Rodríguez, Á. (2019). Selection and promotion processes are not associated by the relative age effect in an elite Spanish soccer academy. *PLoS ONE*, 14(7), 1–15. https://doi.org/10.1371/journal.pone.0219945
- Cejuela, R., Ferriz-Valero, A., & Selles-Pérez, S. (2020). Science-based criteria to identify talent among triathlon athletes. In *Triathlon Medicine* (pp. 317–328). https://doi.org/10.1007/978-3-030-22357-1_19

Charlesworth, R. (2017). World's Best. RC Sports (WA) Pty Ltd.

Chen, W.Y., Wu, S. K., Song, T.F., Chou, K.-M., Wang, K.Y., Chang, Y.C., & Goodbourn, P. T. (2017). Perceptual and Motor Performance of Combat-Sport Athletes Differs According to Specific Demands of the Discipline. *Perceptual and Motor Skills*, 124(1), 293–313. https://doi.org/10.1177/0031512516681342

- Cheung, J. M. Y., Bartlett, D. J., Armour, C. L., Laba, T. L., & Saini, B. (2018). To drug or not to drug: A qualitative study of patients' decision-making processes for managing insomnia.
 Behavioral Sleep Medicine, 16(1), 1–26. https://doi.org/10.1080/15402002.2016.1163702
- Chi, M. T. H. (2006). Two approaches to the study of experts' characteristics. In K. A. Ericsson,
 N. Charness, P. J. Feltovich, & R. R. Hoffman (Eds.), *The Cambridge Handbook of Expertise and Expert Performance* (pp. 21–30). Cambridge University Press.
- Chidley, J. B., MacGregor, A. L., Martin, C., Arthur, C. A., & Macdonald, J. H. (2015). Characteristics explaining performance in downhill mountain biking. *International Journal of Sports Physiology and Performance*, 10(2), 183–190. https://doi.org/10.1123/ijspp.2014-0135
- Chiwaridzo, M., Chandahwa, D., Oorschot, S., Tadyanemhandu, C., Dambi, J. M., Ferguson, G., & Smits-Engelsman, B. C. M. (2018). Logical validation and evaluation of practical feasibility for the SCRuM (School Clinical Rugby Measure) test battery developed for young adolescent rugby players in a resource-constrained environment. *PLoS ONE*, *13*(11), 1–20. https://doi.org/10.1371/journal.pone.0207307
- Chow, J. Y., Davids, K., Hristovski, R., Araújo, D., & Passos, P. (2011). Nonlinear pedagogy: Learning for self-organizing neurobiological systems. *New Ideas in Psychology*, *29*(2), 189–200.
- Christensen, M. K. (2009). "An eye for talent": Talent identification and the "practical sense" of top-level soccer coaches. *Sociology of Sport Journal*, *26*(3), 365–382. https://doi.org/10.1123/ssj.26.3.365
- Chyung, S. Y., Roberts, K., Swanson, I., & Hankinson, A. (2017). Evidence-based survey design: The use of a midpoint on the Likert scale. *Performance Improvement*, *56*(10), 15–23. https://doi.org/10.1002/pfi
- Clarke, D., & Crossland, J. (1985). *Action systems: An introduction to the analysis of complex behaviour*. London: Methuen.
- Cobley, Stephen, Baker, J., Wattie, N., & Mckenna, J. (2009). Annual age-grouping and athlete development: A meta-analytical review of relative age effects in sport. *Sports Med*, *39*(39), 235–256. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/19290678

- Cobley, Steve, Schorer, J., & Baker, J. (2012). Identifiation and development of sport talent:
 A brief introduction to a growing field of research and practice. In J. Baker, S. Cobley, &
 J. Schorer (Eds.), *Talent identification and development in sport: International perspectives* (pp. 1–10). New York, NY: Routledge.
- Collins, D., Collins, L., & Carson, H. J. (2016). "If it feels right, do it": Intuitive decision making in a sample of high-level sport coaches. *Frontiers in Psychology*, 7, 504. https://doi.org/10.3389/fpsyg.2016.00504
- Collins, D., Macnamara, Á., & Cruickshank, A. (2019). Research and practice in talent identification and development some thoughts on the state of play. *Journal of Applied Sport Psychology*, *31*(3), 340–351. https://doi.org/10.1080/10413200.2018.1475430
- Collins, L., & Collins, D. (2013). Decision Making and Risk Management in Adventure Sports Coaching. *Quest*, *65*(1), 72–82. https://doi.org/10.1080/00336297.2012.727373
- Collins, L., & Collins, D. (2016). Professional judgement and decision-making in adventure sports coaching: the role of interaction. *Journal of Sports Sciences*, 34(13), 1231–1239. https://doi.org/10.1080/02640414.2015.1105379
- Connor, J. D., Renshaw, I., & Farrow, D. (2020). Defining cricket batting expertise from the perspective of elite coaches. *PloS One*, *15*(6), e0234802. https://doi.org/10.1371/journal.pone.0234802
- Côté, J., & Gilbert, W. (2009). An integrative definition of coaching effectiveness and expertise. International Journal of Sports Science & Coaching, 4(3), 307–323. https://doi.org/10.1260/174795409789623892
- Côté, J., Salmela, J., Trudel, P., & Baria, A. (1995). The coaching model: A grounded assessment of expert gymnastic coaches' knowledge. *Journal of Sport & Exercise Psychology*, *17*(1992), 1–17. https://doi.org/10.1080/02640410500189173
- Coutinho, P., Mesquita, I., Fonesca, A., & Fonseca, A. M. (2016). Talent development in sport: A critical review of pathways to expert performance. *International Journal of Sports Science & Coaching*, *11*(2), 279–293. https://doi.org/10.1177/1747954116637499
- Craig, C., & Watson, G. (2011). An affordance based approach to decision making in sport:
 Discussing a novel methodological framework. *Revista de Psicologia Del Deporte*, 20(2), 689–708.

- Crespo, M., & McInery, P. (2006). Talent identification and development in tennis. *Coaching* & Sport Science Review, 14(39), 2–3.
- Creswell, J. W., & Poth, C. M. (2018). *Qualitative inquiry & research design: Choosing among five approaches* (4th ed.). https://doi.org/10.1017/CBO9781107415324.004
- Cripps, A. J., Hopper, L. S., & Joyce, C. (2019). Can coaches predict long term career attainment outcomes in adolescent athletes? *International Journal of Sports Science & Coaching*, 14(3), 324-328.
- Cripps, A. J., Joyce, C., Woods, C. T., & Hopper, L. S. (2017). Biological maturity and the anthropometric, physical and technical assessment of talent identified U16 Australian footballers. *International Journal of Sports Science & Coaching*, *12*(3), 344–350. https://doi.org/10.1177/1747954117710507
- Crisp, V. (2010). Towards a model of the judgement processes involved in examination marking. *Oxford Review of Education*, *36*(1), 1–21. https://doi.org/10.1080/03054980903454181
- Cropley, B., Thelwell, R., Mallett, C. J., & Dieffenbach, K. (2019). Guest editorial: Exploring sport psychology in the discipline of sports coaching. *Journal of Applied Sport Psychology*. https://doi.org/10.1080/02626667.2015.1029482
- Cupples, B., & O'Connor, D. (2011). The development of position-specific performance indicators in elite youth rugby league: A coach's perspective. *International Journal of Sports Science & Coaching*, 6(1), 125–142. https://doi.org/10.1260/1747-9541.6.1.125
- Cushion, C. J., Armour, K. M., & Jones, R. L. (2006). Locating the coaching process in practice: Models "for" and "of" coaching. *Physical Education and Sport Pedagogy*, *11*(01), 83–99. https://doi.org/10.1080/17408980500466995
- Datson, N., Weston, M., Drust, B., Gregson, W., & Lolli, L. (2019). High-intensity endurance capacity assessment as a tool for talent identification in elite youth female soccer. *Journal of Sports Sciences, 00*(00), 1–7. https://doi.org/10.1080/02640414.2019.1656323
- Davids, K., & Araújo, D. (2010). The concept of "Organismic Asymmetry" in sport science. *Journal of Science and Medicine in Sport*, 13(6), 633–640. https://doi.org/10.1016/j.jsams.2010.05.002

- Davids, K., Araújo, D., Hristovski, R., Passos, P., & Chow, J. Y. (2012). Ecological dynamics and motor learning design in sport. *Skill Acquisition in Sport: Research, Theory & Practice*, (March), 112–130. https://doi.org/10.13140/RG.2.1.2297.0089
- Davids, K., Araujo, D., & Shuttleworth, R. (2005). Applications of Dynamical Systems Theory to Football. *Science and Football V: The Proceedings of the Fifth World Congress on Science and Football*, (i), 537–550.
- Davids, K., & Baker, J. (2007). Genes, environment and sport performance: Why the naturenurture dualism is no longer relevant. *Sports Medicine*, *37*(11), 961–980. https://doi.org/10.2165/00007256-200737110-00004
- Davids, K., Button, C., & Bennett, S. (2008). *Dynamics of skill acquisition*. Champaign: Human Kinetics.
- Davids, K., Glazier, P., Araujo, D., & Bartlett, R. (2003). Movement systems as dynamical systems. *Sports Medicine*, *33*(4), 245–260.
- Davids, K., Renshaw, I., & Glazier, P. (2005). Movement models from sports reveal fundamental insights into coordination processes. *Exercise and Sport Sciences Reviews*, *33*(1), 36–42.
- Dawson, K., Salmon, P. M., Read, G. J. M., Neville, T., Goode, N., & Clacy, A. (2017). Removing concussed players from the field: The factors influencing decision making around concussion identification and management in Australian Rules Football. 13th International Conference on Naturalistic Decision Making, 48–55.
- Day, D. (2011). Craft coaching and the "discerning eye" of the coach. *International Journal of Sports Science & Coaching*, 6(1), 179–195. https://doi.org/10.1260/1747-9541.6.1.179
- Dennis, P. W., & Carron, A. V. (1999). Strategic decisions of ice hockey coaches as a function of game location. *Journal of Sports Sciences*, 17(4), 263–268. https://doi.org/10.1080/026404199365984
- Dhami, M. K., & Thomson, M. E. (2012). On the relevance of Cognitive Continuum Theory and quasirationality for understanding management judgment and decision making.
 European Management Journal, 30(4), 316–326.
 https://doi.org/10.1016/j.emj.2012.02.002

- Dicks, M., Araújo, D., & van der Kamp, J. (2019). Perception-action for the study of anticipation and decision-making. In A. M. Williams & R. C. Jackson (Eds.), *Anticipation and decision making in sport* (pp. 181–199). New York, NY: Routledge.
- Eccles, D. W., Ward, P., & Woodman, T. (2009). Competition-specific preparation and expert performance. *Psychology of Sport and Exercise*, 10(1), 96–107. https://doi.org/10.1016/j.psychsport.2008.01.006
- Edwards, W. (1954). The theory of decision making. *Psychological Bulletin*, *51*(4), 380–417. https://doi.org/10.1037/h0053870
- Elbe, A., & Wikman, J. M. (2017). Psychological factors in developing high performance athletes. In J. Baker, S. Cobley, J. Schorer, & N. Wattie (Eds.), *Routledge Handbook of Talent Identification and Development in Sport* (pp. 169–180). New York, NY: Routledge.
- Ellingsen, J. E., & Danielsen, A. G. (2017). Norwegian children's rights in sport and coaches' understanding of talent. *International Journal of Children's Rights*, *25*(2), 412–437. https://doi.org/10.1163/15718182-02502006
- Emmonds, S., Till, K., Jones, B., Mellis, M., & Pears, M. (2016). Anthropometric, speed and endurance characteristics of English academy soccer players: Do they influence obtaining a professional contract at 18 years of age? *International Journal of Sports Science & Coaching*, *11*(2), 212–218. https://doi.org/10.1177/1747954116637154
- Faber, I. R., Pion, J., Munivrana, G., Faber, N. R., & Nijhuis-Van der Sanden, M. W. G. (2017).
 Does a perceptuomotor skills assessment have added value to detect talent for table tennis in primary school children? *Journal of Sports Sciences*, 00(00), 1–8. https://doi.org/10.1080/02640414.2017.1316865
- Fajen, B. R. (2005). Perceiving possibilities for action: On the necessity of calibration and perceptual learning for the visual guidance of action. *Perception*, 34(6), 717–740. https://doi.org/10.1068/p5405
- Fajen, B. R., & Warren, W. H. (2003). Behavioural dynamics of steering, obstacle avoidance, and route selection. *Journal of Experimental Psychology: Human Perception and Performance*, 29(2), 343–362. https://doi.org/10.1037/0096-1523.29.2.343
- Falk, B., Lidor, R., Lander, Y., & Lang, B. (2004). Talent identification and early development of elite water-polo players: a 2-year follow-up study. *Journal of Sports Sciences*, 22(4), 347–355. https://doi.org/10.1080/02640410310001641566

- Ferreira Celestino, T., Gomes Leitão, J. C., Borges Sarmento, H., Routen, A., & Almeida Pereira, A. (2015). Elite coaches views on factors contributing to excellence in orienteering. *Cultura, Ciencia y Deporte, 10*(28), 77–86.
- Fiander, M. F., Jones, M. I., & Parker, J. K. (2013). Coaches' perceptions of the use of chronological and biological age in the identification and development of talented athletes. *Athletic Insight*, 5(2), 181–196. https://doi.org/10.2966/scrip.100113.57
- Ford, P. R., Luis, J., Bordonau, D., Bonanno, D., Tavares, J., Groenendijk, C., ... Salvo, A. (2020).
 A survey of talent identification and development processes in the youth academies of professional soccer clubs from around the world. *Journal of Sports Sciences*, 38:11-12, 1269-1278. https://doi.org/10.1080/02640414.2020.1752440
- Franchini, E. (2014). Born to fight? Genetics and combat sports. *Revista de Artes Marciales Asiaticas*, *9*(1), 1–8. https://doi.org/10.18002/rama.v10i1.1635
- Fraser-Thomas, J., Côté, J., & Deakin, J. (2008). Understanding dropout and prolonged engagement in adolescent competitive sport. *Psychology of Sport and Exercise*, 9, 645– 662.
- Gibson, J. J. (1966). The senses considered as perceptual systems. Houghton Mifflin.
- Gibson, J. J. (1979). The ecological approach to visual perception. Houghton Mifflin.
- Gigerenzer, G. (2007). Gut feelings: The intelligence of the unconscious. Penguin.
- Gines, S. (2017). Tastes for true talent: How professional baseball scouts define talent and decide who gets to play. [Doctorcal Thesis, University of St. Thomas]. University of St. Thomas Repository. https://ir.stthomas.edu/caps_ed_lead_docdiss/90/
- Giske, R., Benestad, B., Haraldstad, K., & Hoigaard, R. (2013). Decision-making styles among Norwegian soccer coaches: An analysis of decision-making style in relationt to elite and non-elite coaching and level of playing history. *International Journal of Sports Science* & Coaching, 8(4), 689–701. https://doi.org/10.1260/1747-9541.8.4.689
- Glazier, P. S. (2017). Towards a Grand Unified Theory of sports performance. *Human Movement Science*, *56*, 139-156. <u>https://doi.org/10.1016/j.humov.2015.08.001</u>

- Gonçalves, L., Santos, A., Tavares, F., & Janeira, M. (2017). From talent to high performance: The view of coaches, players and club coordinators on the relevant factors in the development of a basketball player. *Cuademos de Psicología Del Deporte, 7*(3), 129– 135.
- Gray, H. J., & Plucker, J. A. (2010). "She's a natural": Identifying and developing athletic talent. *Journal for the Education of the Gifted*, 33(3), 361–380. https://doi.org/10.1177/016235321003300304
- Greenwood, D. A., Davids, K., & Renshaw, I. (2012a). How elite coaches' experiential knowledge might enhance empirical research on sport performance: A response to commentary. *International Journal of Sports Science & Coaching*, 7(2), 427–429. https://journals.sagepub.com/doi/10.1260/1747-9541.7.2.427
- Greenwood, D. A., Davids, K., & Renshaw, I. (2012b). How elite coaches' experiential knowledge might enhance empirical research on sport performance. *International Journal of Sports Science & Coaching*, 7(2), 411–422. https://doi.org/10.1260/1747-9541.7.2.427
- Greenwood, D., Davids, K., & Renshaw, I. (2014). Experiential knowledge of expert coaches can help identify informational constraints on performance of dynamic interceptive actions. *Journal of Sports Sciences*, 32(4), 328–335. https://doi.org/10.1080/02640414.2013.824599
- Guenter, R. W., Dunn, J. G. H., & Holt, N. L. (2019). Talent identification in youth ice hockey:
 Exploring "intangible" player characteristics. *The Sport Psychologist*, *33*(4), 323–333.
 https://doi.org/10.1123/tsp.2018-0155
- Gul, J. (2016). Combat sport draftees aim to impress at Australian Institute of Sport in Canberra. Retrieved September 2, 2020, from https://www.abc.net.au/news/2016-01-30/combat-sport-draftees-aim-to-impress-in-canberra/7127024
- Gulbin, J. P., Croser, M. J., Morley, E. J., & Weissensteiner, J. R. (2013). An integrated framework for the optimisation of sport and athlete development: A practitioner approach. *Journal of Sports Sciences*, 31(12), 1319–1331. https://doi.org/10.1080/02640414.2013.781661

- Güllich, A. (2014). Selection, de-selection and progression in German football talent promotion. *European Journal of Sport Science*, 14(6), 530–537. https://doi.org/10.1080/17461391.2013.858371
- Güllich, A. (2017). International medallists' and non-medallists' developmental sport activities – A matched-pairs analysis. *Journal of Sports Sciences*, 35(23), 2281–2288. https://doi.org/10.1080/02640414.2016.1265662
- Güllich, A., & Cobley, S. (2017). On the efficacy of talent identification and talent development programmes. In J. Baker, S. Cobley, J. Schorer, & N. Wattie (Eds.), *Routledge Handbook of Talent Identification and Development in Sport* (pp. 80–98). New York, NY: Routledge.
- Güllich, A., & Emrich, E. (2006). Evaluation of the support of yount athletes in the elite sports system. *European Journal for Sport and Society*, *3*(2), 85–108.
- Güllich, A., & Emrich, E. (2014). Considering long-term sustainability in the development of world class success. *European Journal of Sport Science*, 14(sup1), S383–S397. https://doi.org/10.1080/17461391.2012.706320
- Halperin, I. (2018). Case-studies in exercise science and sport sciences: A powerful tool to bridge the science-practice gap. *International Journal of Sports Physiology and Performance*, 13(6): 824-825. https://doi.org/10.1123/ijspp.2018-0185
- Hammond, K. R. (2000). Judgments under stress. Oxford University Press.
- Hartman, M. (2011). Competitive performance compared to combine performance as a valid predictor of NFL status. *Journal of Strength & Conditioning Research*, 25, 105–106. https://doi.org/10.1097/01.JSC.0000395746.03546.e8
- Hastie, R. (2001). Problems for judgment and decision making. *Annual Review of Psychology*, *52*, 653–683.
- Headrick, J., Renshaw, I., Davids, K., Pinder, R. A., & Araújo, D. (2015). The dynamics of expertise acquisition in sport: The role of affective learning design. *Psychology of Sport* and Exercise, 16(P1), 83–90. https://doi.org/10.1016/j.psychsport.2014.08.006
- Hendry, D. T., Williams, A. M., & Hodges, N. J. (2018). Coach ratings of skills and their relations to practice, play and successful transitions from youth-elite to adult-professional status in soccer. *Journal of Sports Sciences*, 36(17), 2009–2017. https://doi.org/10.1080/02640414.2018.1432236

- Hertwig, R., Barron, G., Weber, E. U., & Erev, I. (2004). Decisions from experience and the effect of rare events in risky choice. *Psychological Science*, *15*(8), 534–539. https://doi.org/10.1111/j.0956-7976.2004.00715.x
- Highhouse, S. (2008). Stubborn reliance on intuition and subjectivity in employee selection. Industrial and Organizational Psychology, 1, 333–342.
- Hill, B., & Sotiriadou, P. (2016). Coach decision-making and the relative age effect on talent selection in football. *European Sport Management Quarterly*, 16(3), 292–315. https://doi.org/10.1080/16184742.2015.1131730
- Hoare, D. G., & Warr, C. R. (2000). Talent identification and women's soccer: An Australian experience. *Journal of Sports Sciences*, 18(9), 751–758. https://doi.org/10.1080/02640410050120122
- Hoffman, R. R., & Yates, J. F. (2005). Decision (?) Making (?). *IEEE Intelligent Systems*, 20(4), 76–83.
- Hoffrage, U., & Marewski, J. N. (2015). Unveiling the Lady in Black: Modeling and aiding intuition. Journal of Applied Research in Memory and Cognition, 4(3), 145–163. https://doi.org/10.1016/j.jarmac.2015.08.001
- Holt, N. L., & Dunn, J. G. H. (2004). Toward a grounded theory of the psychosocial competencies and environmental conditions associated with soccer success. *Journal of Applied Sport Psychology*, 16(3), 199–219. https://doi.org/10.1080/10413200490437949
- Höner, O., & Feichtinger, P. (2016). Psychological talent predictors in early adolescence and their empirical relationship with current and future performance in soccer. *Psychology* of Sport and Exercise, 25, 17–26. https://doi.org/10.1016/j.psychsport.2016.03.004
- Hristovski, R., Davids, K., & Araújo, D. (2006). Affordance-controlled bifurcations of action patterns in martial arts. *Nonlinear Dynamics, Psychology, and Life Sciences*, *10*(4), 409–444. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/16884651
- Hristovski, R., Davids, K., Araújo, D., & Button, C. (2006). How boxers decide to punch a target: Emergent behaviour in nonlinear dynamical movement systems. *Journal of Sports Science and Medicine*, 60–73. Retrieved from http://www.jssm.org
- Hyndman, D., & Athanasopoulos, G. (2018). *Forecasting: Principles and practice* (2nd ed.). O Texts.

IBM Corporation. (2017). SPSS Statistics for Windows. IBM Corp.

- International Olympic Committee. (2019). International Olympic Committee. Retrieved January 28, 2019, from https://www.olympic.org/
- Inui, N. (2018). Interpersonal coordination: A social neuroscience approach. https://doi.org/10.1007/978-981-1765-1
- Issurin, V. B. (2017). Evidence-based prerequisites and precursors of athletic talent: A review. *Sports Medicine, 47* (10), 1993-2010. https://doi.org/10.1007/s40279-017-0740-0
- Jacklin, R., Sevdalis, N., Darzi, A., & Vincent, C. A. (2009). Efficacy of cognitive feedback in improving operative risk estimation. *American Journal of Surgery*, 197(1), 76–81. https://doi.org/10.1016/j.amjsurg.2008.07.049
- Jako, R. A., & Murphy, K. R. (1990). Distributional ratings, judgment decomposition, and their impact on interrater agreement and rating accuracy. *Journal of Applied Psychology*, 75(5), 500–505. <u>https://doi.org/10.1037/0021-9010.75.5.500</u>
- James, L. P., Haff, G. G., Kelly, V. G., & Beckman, E. M. (2016). Towards a Determination of the Physiological Characteristics Distinguishing Successful Mixed Martial Arts Athletes: A Systematic Review of Combat Sport Literature. *Sports Medicine*, 46(10), 1525–1551. https://doi.org/10.1007/s40279-016-0493-1
- James, L. P., Robertson, S., Haff, G. G., Beckman, E. M., & Kelly, V. G. (2017). Identifying the performance characteristics of a winning outcome in elite mixed martial arts competition. *Journal of Science and Medicine in Sport*, 20(3), 296–301. https://doi.org/10.1016/j.jsams.2016.08.001
- James, L.P., Sweeting, A.J., Kelly, V.G., & Robertson, S. (2019). Longitudinal analysis of tactical strategy in the men's division of the Ultimate Fighting Championship. *Frontiers in Artificial Intelligence*, 2, 29. https://doi.org/10.3389/frai.2019.00029
- Johansson, A., & Fahlén, J. (2017). Simply the best , better than all the rest? Validity issues in selections in elite sport. *International Journal of Sports Science & Coaching*, 12(4), 470– 480. https://doi.org/10.1177/1747954117718020
- Johnson, J. G. (2006). Cognitive modeling of decision making in sports. *Psychology of Sport* and Exercise, 7(6), 631–652. https://doi.org/10.1016/j.psychsport.2006.03.009

- Johnson, M. B., Castillo, Y., Sacks, D. N., Cavazos, Jr, J., Edmonds, W. A., & Tenenbaum, G. (2008). "Hard work beats talent until talent decides to work hard": Coaches' perspectives regarding differentiating elite and non-elite swimmers. *International Journal of Sports Science & Coaching*, 3(3), 417–430. https://doi.org/10.1260/174795408786238579
- Johnston, K., & Baker, J. (2020). Waste reduction strategies: Factors affecting talent wastage and the efficacy of talent selection in sport. *Frontiers in Psychology*, *10*, 2925. https://doi.org/10.3389/fpsyg.2019.02925
- Johnston, K., Wattie, N., Schorer, J., & Baker, J. (2018). Talent Identification in Sport: A Systematic Review. *Sports Medicine*, *48*(1), 97–109. https://doi.org/10.1007/s40279-017-0803-2
- Jokuschies, N., Gut, V., & Conzelmann, A. (2017). Systematizing coaches' "eye for talent": Player assessments based on expert coaches' subjective talent criteria in top-level youth soccer. *International Journal of Sports Science & Coaching*, *12*(5), 565–576. https://doi.org/10.1177/1747954117727646
- Jones, R. L., Armour, K. M., & Potrac, P. (2003). Constructing expert knowledge: A case study of a top-level professional soccer coach. *Sport, Education and Society*, 8(2), 213–229. https://doi.org/10.1080/13573320309254
- Judo Australia. (2019). *Strategic plan 2019-2022*. Retrieved from https://docs.wixstatic.com/ugd/570d63_6d5c774e17dc4f5b94b720d3ac7f720a.pdf
- Julio, U. F., Takito, M. Y., Mazzei, L., Miarka, B., Sterkowicz, S., & Franchini, E. (2011). Tracking
 10-year competitive winning performance of judo athletes across age groups.
 Perceptual and Motor Skills, 113(1), 139–149.
 https://doi.org/10.2466/05.10.11.PMS.113.4.139-149
- Kahneman, D. (1991). Judgment and decision making: A personal view. *Psychological Science*, 2(3), 142–145. https://doi.org/10.1111/j.1467-9280.1991.tb00121.x
- Kahneman, D. (2011). Thinking, fast and slow. Penguin.
- Kauffman, S. A. (1993). The origins of order: Self-organization and selection in evolution. Oxford University Press.
- Kaya, A. (2014). Decision making by coaches and athletes in sport. *Procedia Social and Behavioral Sciences*, *152*, 333–338. https://doi.org/10.1016/j.sbspro.2014.09.205

- Kearney, P. E., Carson, H. J., & Collins, D. (2018). Implementing technical refinement in highlevel athletics: Exploring the knowledge schemas of coaches. *Journal of Sports Sciences*, 36(10), 1118–1126. https://doi.org/10.1080/02640414.2017.1358339
- Keller, B. S., Raynor, A. J., Bruce, L., & Iredale, F. (2016). Technical attributes of Australian youth soccer players: Implications for talent identification. *International Journal of Sports Science & Coaching*, 11(6), 819–824. https://doi.org/10.1177/1747954116676108
- Kelso, J. A. S. (1995). *Dynamic patterns: The self-organization of brain and behaviour* (3rd ed.). Cambridge, MA: The MIT Press.
- Kerr, R. (2019). The role of science in the practice of talent identification: A case study from gymnastics in New Zealand. Sport in Society, 22(9), 1589–1603. https://doi.org/10.1080/17430437.2018.1435032

Klein, G. A. (2003). *The Power of Intuition*. New York, NY: Currency-Doubleday.

- Kohlrausch, W. (1929). Zusammenhänge von Körperform und Leistung: Ergebnisse de anthropometrischen Messungen an den Athleten der Amsterdamer Olypiade. Ergibnisse Der Sportarzlichen Untersuchungen Bei Den IX. Olympischen Spielen in Amsterdam 1928, 30–47. Berlin, Heidelberg: Springer.
- Kordi, R., Maffulli, N., Wroble, R. R., & Wallance, A. (Eds.). (2009). *Combat sports medicine*. Springer Science & Business Media. https://doi.org/10.1007/978-1-84800-354-5_19
- Kovalchik, S. A., Bane, M. K., & Reid, M. (2017). Getting to the top: An analysis of 25 years of career rankings trajectories for professional women's tennis. *Journal of Sports Sciences*, 35(19), 1904-1910. https://doi.org/10.1080/02640414.2016.1241419
- Koz, D., Fraser-Thomas, J., & Baker, J. (2012). Accuracy of professional sports drafts in predicting career potential. *Scandinavian Journal of Medicine and Science in Sports*, 22(4), 64–69. https://doi.org/10.1111/j.1600-0838.2011.01408.x
- Krabben, K., Orth, D., & van der Kamp, J. (2019). Combat as an interpersonal synergy: An ecological dynamics approach to combat sports. *Sports Medicine*, 1-12. https://doi.org/10.1007/s40279-019-01173-y
- Kugler, P. N., & Turvey, M. T. (1987). Information, natural law and the assembly of rhythmic movement. London: Lawrence Erbaum.

- Larkin, P., & O'Connor, D. (2017). Talent identification and recruitment in youth soccer: Recruiter's perceptions of the key attributes for player recruitment. *PLoS ONE*, *12*(4), e0175716. https://doi.org/https://doi.org/10.1371/journal.pone.0175716
- Larkin, P., & Reeves, M. J. (2018). Junior-elite football: Time to re-position talent identification? *Soccer & Society*, *19*(8), 1183–1192. https://doi.org/10.1080/14660970.2018.1432389
- Letts, L., Wilkins, S., Law, M., Stewart, D., Bosch, J., & Westmorland, M. (2007). *Guidelines* for critical review form: Qualitative studies (Version 2.0). Retrieved from https://srsmcmaster.ca/wp-content/uploads/2015/04/Guidelines-for-Critical-Review-Form-Qualitative-Studies-English.pdf
- Lewis, M. (2004). *Moneyball: The art of winning an unfair game*. W. W. Norton & Company, Inc.
- Li, P., De Bosscher, V., Pion, J., Weissensteiner, J. R., & Vertonghen, J. (2018). Is international junior success a reliable predictor for international senior success in elite combat sports? *European Journal of Sport Science*, 18(4), 550–559. https://doi.org/10.1080/17461391.2018.1439104
- Lidor, R., Côté, J., & Hackfort, D. (2009). ISSP position stand: To test or not to test? The use of physical skill tests in talent detection and in early phases of sport development. *International Journal of Sport and Exercise Psychology*, 7(2), 131–146. https://doi.org/10.1080/1612197X.2009.9671896
- Lidor, R., Melnik, Y., Bilkevitz, A., Arnon, M., & Falk, B. (2005). Measurement of talent in judo using a unique, judo-specific ability test. *Journal of Sports Medicine and Physical Fitness*, 45(1), 32–37.
- Likierman, A. (2020). The elements of good judgment. Retrieved August 30, 2020, from Harvard Business Review website: https://hbr.org/2020/01/the-elements-of-goodjudgment
- Lloyd, R. S., Oliver, J. L., Faigenbaum, A. D., Howard, R., De Ste Croix, M. B. A., Williams, C. A.,
 ... Myer, G. D. (2015). Long-term athletic development, part 2: Barriers to success and
 potential solutions. *Journal of Strength and Conditioning Research*, 29(5), 1451–1464.

- Lund, S., & Söderström, T. (2017). To see or not to see: Talent identification in the Swedish Football Association. Sociology of Sport Journal, 34(3), 248–258. https://doi.org/https://doi.org/10.1123/ssj.2016-0144
- Lyle, J. (2010). Coaches' decision making: A naturalistic decision making analysis. In J. Lyle &C. Cushion (Eds.), Sports coaching: Professionalisation and practice (pp. 27–42).Elsevier.
- Lyle, J., & Cushion, C. (2017). *Sport coaching concepts: A framework fo coaching practice* (2nd ed.). Routledge.
- Lyle, J., & Vergeer, I. (2013). Recommendations on the methods used to investigate coaches' decision making. In P. Potrac, W. Gilbert, & J. Denison (Eds.), *Routledge Handbook of Sports Coaching* (pp. 121–132). Routledge.
- Lyons, B. D., Hoffman, B. J., Michel, J. W., & Williams, K. J. (2011). On the Predictive Efficiency of Past Performance and Physical Ability: The Case of the National Football League. *Human Performance*, *24*(APRIL), 158–172. https://doi.org/10.1080/08959285.2011.555218
- MacMahon, C., Bailey, A., Croser, M., & Weissensteiner, J. (2019). Exploring the skill of recruiting in the Australian Football League. *International Journal of Sports Science & Coaching*, 14(1), 72–81. https://doi.org/10.1177/1747954118809775
- MacMahon, C., & McPherson, S. L. (2009). Knowledge base as a mechanism for perceptualcognitive tasks: Skill is in the details! *International Journal of Sport Psychology*, *40*(4), 565–579.
- Mahon, T. (2004). Linking promise to the podium: Talent identification and development (TID) in New Zealand. A report to SPARC's board from the TID taskforce. Retrieved from https://docplayer.net/27753787-Linking-promise-to-the-podium.html
- Malina, R. M. (2003). Youth sports: Readiness, selection and trainability. In J. A. P. Day & J.W. Duguet (Eds.), *Kinanthropometry IV*. Routledge.
- Maloney, M. A. (2018). Enhancing representative practice design through consideration of affective and situational constraints. [Doctoral thesis, Victoria University]. Victoria University Research Repository. http://vuir.vu.edu.au/38652/

- Maloney, M. A., Renshaw, I., Headrick, J., Martin, D. T., & Farrow, D. (2018). Taekwondo fighting in training does not simulate the affective and cognitive demands of competition: Implications for behavior and transfer. *Frontiers in Psychology*, *9*, 1–13. https://doi.org/10.3389/fpsyg.2018.00025
- Mann, D. L., Dehghansai, N., & Baker, J. (2017). Searching for the elusive gift: Advances in talent identification in sport. *Current Opinion in Psychology*, 16(17), 128–133. https://doi.org/10.1016/j.copsyc.2017.04.016
- Marasso, D., Laborde, S., Bardaglio, G., & Raab, M. (2014). A developmental perspective on decision making in sports. *International Review of Sport and Exercise Psychology*, 7(1), 251–273. https://doi.org/10.1080/1750984X.2014.932424
- Martindale, R.J.J., Collins, D., & Daubney, J. (2005) Talent development: A guide for practice and research within sport. *Quest, 57*(4), 353-375. https://doi.org/10.1080/00336297.2005.10491862
- Martindale, R. J. J., Collins, D., & Abraham, A. (2007). Effective talent development: The elite coach perspective in UK sport. *Journal of Applied Sport Psychology*, *19*(2), 187–206. https://doi.org/10.1080/10413200701188944
- Matthys, S. P. J., Vaeyens, R., Vandendriessche, J., Vandorpe, B., Pion, J., Coutts, A. J., ...
 Philippaerts, R. M. (2011). A multidisciplinary identification model for youth handball. *European Journal of Sport Science*, 11(5), 355–363.
 https://doi.org/10.1080/17461391.2010.523850
- McCarthy, N., & Collins, D. (2014). Individual identification & selection bias versus the eventual confirmation of talent: Evidence for the benefits of a rocky road? *Journal of Sports Sciences*, *32*(17), 1604–1610. https://doi.org/10.1080/02640414.2014.908322
- McCullick, B., Schempp, P., Hsu, S.H., Jung, J. H., Vickers, B., & Schuknecht, G. (2006). An analysis of the working memories of expert sport instructors. *Journal of Teaching in Physical Education*, *25*(2), 149–165. https://doi.org/10.1123/jtpe.25.2.149
- McDonnell, A. (2011). Still fighting the "War for Talent"? Bridging the science versus practice gap. Journal of Business Psychology, 26, 169–173. https://doi.org/10.1007/s10869-011-9220-y

- McIntosh, S., Kovalchik, S., & Robertson, S. (2019). Comparing subjective and objective evaluations of player performance in Australian Rules football. *PLOS ONE*, *14*(8), e0220901. https://doi.org/10.1371/journal.pone.0220901
- Medved, R. (1966). Body height and predisposition for certain sports. *Journal of Sports Medicine and Physical Fitness*, 6(2), 89–91.
- Mellalieu, S. D., Neil, R., Hanton, S., & Fletcher, D. (2009). Competition stress in sport performers: Stressors experienced in the competition environment. *Journal of Sports Sciences*, 27(7), 729–744. https://doi.org/10.1080/02640410902889834
- Milazzo, N., Farrow, D., Ruffault, A., & Fournier, J. F. (2016). Do karate fighters use situational probability information to improve decision-making performance during on-mat tasks?
 Journal of Sports Sciences, 34(16), 1547–1556. https://doi.org/10.1080/02640414.2015.1122824
- Milistetd, M., Mesquita, I., Sobrinho, A. S., Carrara, P., & Nascimento, J. (2013). Coaches representation about detection and selection of talents on the Brazillian volleyball. *International Journal of Sports Science, 3*(5), 157–162. https://doi.org/10.5923/j.sports.20130305.03
- Miller, P. K., Cronin, C., & Baker, G. (2015). Nurture, nature and some very dubious social skills: An interpretative phenomenological analysis of talent identification practices in elite English youth soccer. *Qualitative Research in Sport, Exercise and Health*, 7(5), 642– 662. https://doi.org/10.1080/2159676X.2015.1012544
- Mills, A., Butt, J., Maynard, I., & Harwood, C. G. (2012). Identifying factors perceived to influence the development of elite youth football academy players. *Journal of Sports Sciences*, 30(15), 1593–1604. https://doi.org/10.1080/02640414.2012.710753
- Moher, D., Shamseer, L., Clarke, M., Ghersi, D., Liberati, A., Petticrew, M., ... PRISMA-P Group.
 (2015). Preferred reporting items for systematic review and meta-analysis protocols
 (PRISMA-P) 2015 statement. *Systematic Reviews*, 4(1), 1.
 https://doi.org/10.1186/2046-4053-4-1

- Morris, K. S., Jenkins, D. G., Osborne, M. A., Rynne, S. B., Shephard, M. E., & Skinner, T. L. (2019). The role of the upper and lower limbs in front crawl swimming: The thoughts and practices of expert high-performance swimming coaches. *International Journal of Sports Science & Coaching*, 14(5), 629–638. https://doi.org/10.1177/1747954119866358
- Moskowitz, T. J., & Wetheim, L. J. (2011). *Scorecasting: The hidden influences behind how sports and played and games are won*. New York, NY: Three Rivers Press.
- Mumford, M. D., McIntosh, T., & Mulhearn, T. (2018). Using cases to understand expert performance: Method and methodological triangulation. In K. A. Ericsson, R. R. Hoffman, A. Kozbelt, & A. M. Williams (Eds.), *The Cambridge Handbook of Expertise and Expert Performance* (2nd ed., pp. 291–309). New York, NY: Cambridge University Press.
- Murray, S., James, N., Perš, J., Mandeljc, R., & Vučković, G. (2018). Using a situation awareness approach to determine decision-making behaviour in squash. *Journal of Sports Sciences*, *36*(12), 1415–1422. https://doi.org/10.1080/02640414.2017.1389485
- Musculus, L., & Lobinger, B. H. (2018). Psychological characteristics in talented soccer players
 Recommendations on how to improve coaches' assessment. *Frontiers in Psychology*, 9:41. https://doi.org/10.3389/fpsyg.2018.00041
- Nash, C. S., & Collins, D. (2006). Tacit knowledge in expert coaching: Science or art? *Quest*, 58(4), 465–477. https://doi.org/10.1080/00336297.2006.10491894
- Nash, C. S., Martindale, R., Collins, D., & Martindale, A. (2012). Parameterising expertise in coaching: Past, present and future. *Journal of Sports Sciences*, *30*(10), 985–994. https://doi.org/10.1080/02640414.2012.682079
- Newell, K.M. (1985). Coordination, control and skill. In D. Goodman, R. B. Wilberg, & I. . Franks (Eds.), *Differing Perspectives in Motor Leraning, Memory and Control*. Elsevier Science Publishers.
- Newell, K.M. (1986). Constraints on the development of coordination. In M. G. Wade & H. T.
 A. Whiting (Eds.), *Motor development in children: Aspects of coordination and control* (pp. 341–360). Martinus Nijhoff Publishers.
- Newell, K.M., & Valvano, J. (1998). Movement science: Therapeutic intervention as a constraint in learning and relearning movement skills. *Scandinavian Journal of Occupational Therapy*, 5(2), 51–57. https://doi.org/10.3109/11038129809035730

- Newell, K.M, & Jordan, K. (2007). Task constraints and movement organisation: A common language. In W.E. Davis & G.D. Broadhead (Eds.), *Ecological task analysis and movement* (pp. 5–23). Human Kinetics.
- Newell, K.M, & Ranganathan, R. (2010). Instructions as constraints in motor skill acquisition.
 In I. Renshaw, K. Davids, & G. J. P. Savelsbergh (Eds.), *Motor learning in practice: A constraints led approach* (pp. 17–31). New York, NY: Routledge.
- Noh, J.W., Park, B.S., Kim, M.Y., Lee, L.K., Yang, S.M., Lee, W.D., ... Kim, J. (2015). Analysis of combat sports players' injuries according to playing style for sports physiotherapy research. *Journal of Physical Therapy Science*, 27(8), 2425–2430. https://doi.org/10.1589/jpts.27.2425
- Norjali Wazir, M.R.W., Torfs, M., Mostaert, M., Pion, J., & Lenoir, M. (2017). Predicting judo champions and medallists using statistical modelling. *Archives of Budo, 13* 161-168.
- Nowell, L. S., Norris, J. M., White, D. E., & Moules, N. J. (2017). Thematic analysis: Striving to meet the trustworthiness criteria. *International Journal of Qualitative Methods*, 16(1), 1–13. https://doi.org/10.1177/1609406917733847
- O'Brien-Smith, J., Tribolet, R., Smith, M. R., Bennett, K. J. M., Fransen, J., Pion, J., & Lenior, M. (2019). The Use of the Körperkoordinationstest für Kinder in the Talent Pathway in Youth Athletes: A Systematic Review. *Journal of Science and Medicine in Sport*. 22(9): 1021-1029. https://doi.org/10.1016/j.jsams.2019.05.014
- O'Connor, D., Larkin, P., & Mark Williams, A. (2016). Talent identification and selection in elite youth football: An Australian context. *European Journal of Sport Science*, *16*(7), 837–844. https://doi.org/10.1080/17461391.2016.1151945
- Olusoga, P., Butt, J., Maynard, I., & Hays, K. (2010). Stress and coping: A study of world class coaches. Journal of Applied Sport Psychology, 22(3), 274–293. https://doi.org/10.1080/10413201003760968
- Oppici, L., Panchuk, D., Serpiello, F. R., & Farrow, D. (2017). Long-term Practice with Domain-Specific Task Constraints Influences Perceptual Skills. *Frontiers in Psychology*, 8:1387. https://doi.org/10.3389/fpsyg.2017.01387
- Patton, M. Q. (2015). *Qualitative research & evaluation methods: Integrating theory and practice* (4th ed.). SAGE.

- Pearson, D. T., Naughton, G. A., & Torode, M. (2006). Predictability of physiological testing and the role of maturation in talent identification for adolescent team sports. *Journal* of Science and Medicine in Sport, 9(4), 277–287. https://doi.org/10.1016/j.jsams.2006.05.020
- Pienaar, A. E., Spamer, M. J., & Steyn Jr, H. S. (1998). Identifying and developing rugby talent among 10-year-old boys: A practical model. *Journal of Sports Sciences*, 16(8), 691–699. https://doi.org/10.1080/026404198366326
- Pinder, R. A. (2012). Representative learning design in dynamic interceptive actions. [Doctorial thesis, Queensland University of Technology]. Queensland University of Technology ePrints. https://eprints.qut.edu.au/59803/
- Pion, J., Hohmann, A., Liu, T., Lenoir, M., & Segers, V. (2017). Predictive models reduce talent development costs in female gymnastics. *Journal of Sports Sciences*, 35(8), 806–811. https://doi.org/10.1080/02640414.2016.1192669
- Platanov, V. (2016). My profession The game. Lulu Press, Inc.
- Plessner, H., & Haar, T. (2006). Sports performance judgments from a social cognitive perspective. *Psychology of Sport and Exercise*, 7(6), 555–575. https://doi.org/10.1016/j.psychsport.2006.03.007
- Pope, C. C., Penney, D., & Smith, T. B. (2018). Overtraining and the complexities of coaches' decision-making: Managing elite athletes on the training cusp. *Reflective Practice*, 19(2), 145–166. https://doi.org/10.1080/14623943.2017.1361923
- Potrac, P., Jones, R. L., & Armour, K. (2002). "It's all about getting respect": The coaching behaviors of an expert English soccer coach. *Sport Education and Society*, 7(2), 183–202.
- Putallaz, M., Baldwin, J., & Selph, H. (2005). The Duke University Talent Identification Program. *High Ability Studies, 16*(1), 41–54. https://doi.org/10.1080/13598130500115221
- Raab, M., Bar-Eli, M., Plessner, H., & Araújo, D. (2019). The past, present and future of research on judgment and decision making in sport. *Psychology of Sport and Exercise*, 42(November 2019), 25–32. <u>https://doi.org/10.1016/j.psychsport.2018.10.004</u>

- Re, A. H. N., Correa, U. C., & Bohme, M. T. S. (2010). Anthropometric characteristics and motor skills in talent selection and development in indoor soccer. *Perceptual and Motor Skills*, *110*(3), 916–930. https://doi.org/10.2466/PMS.110.3.916-930
- Reid, M., Crespo, M., Santilli, L., Miley, D., & Dimmock, J. (2007). The importance of the International Tennis Federation's junior boys' circuit in the development of professional tennis players. *Journal of Sports Sciences*, 25(6), 667–672. https://doi.org/10.1080/02640410600811932
- Rein, R., Davids, K., & Button, C. (2010). Adaptive and phase transition behavior in performance of discrete multi-articular actions by degenerate neurobiological systems. *Experimental Brain Research*, 201(2), 307–322. https://doi.org/10.1007/s00221-009-2040-x
- Renshaw, I., Davids, K., Shuttleworth, R., & Chow, J. Y. (2009). Insights from ecological psychology and dynamical systems theory can underpin a philosophy of coaching. *International Journal of Sport Psychology*, *40*(4), 540–602.
- Renshaw, I., & Chappel, G. (2010). A constraints-led approach to talent development in cricket. In L. Kidman & B. J. Lombardo (Eds.), *Athlete-centred coaching: Developing decision makers* (pp. 151–172). IPC Print Resources.
- Renshaw, I., Davids, K., & Savelsbergh, G. J. P. (2010). *Motor learning in practice: A constraints-led approach*. London: Routledge.
- Renshaw, I., & Davids, K. W. (2014). Task constraints. In R. Eklund & G. Tenenbaum (Eds.), Encyclopedia of Sport and Exercise Psychology (pp. 734–737). SAGE Publications Inc.
- Rhodes, C., Brundrett, M., & Nevill, A. (2008). Leadership talent identification and development: Perceptions of headds, middle leaders and classroom teachers in 70 contextually different primary and secondary schools in England. *Educational Management Administration & Leadership*, 36(3), 311–335. https://doi.org/10.1177/1741143208090592
- Ritchie, L., Wright-St Clair, V. A., Keogh, J., & Gray, M. (2014). Community integration after traumatic brain injury: A systematic review of the clinical implications of measurement and service provision for older adults. *Archives of Physical Medicine and Rehabilitation*, 95(1), 163–174. <u>https://doi.org/10.1016/j.apmr.2013.08.237</u>

- Roberts, Alexandra. H, Greenwood, D., Humberstone, C., Iredale, F., Stanley, M., & Raynor,A. J. (2020). Understanding coaches' "gut instinct" in talent identification. Journal ofSportSciences,OnlineAheadAhttps://doi.org/10.1080/02640414.2020.1823083
- Roberts, Alexandra H., Greenwood, D. A., Stanley, M., Humberstone, C., Iredale, F., & Raynor,
 A. (2019a). Coach knowledge in talent identification: A systematic review and metasynthesis. *Journal of Science and Medicine in Sport*, 22(10), 1163–1172. https://doi.org/10.1016/j.jsams.2019.05.008
- Roberts, Alexandra H, Raynor, A., Greenwood, D., Humberstone, C., & Iredale, F. (2019b). Proposition of a new conceptualisation of the longitudinal talent identification process. *The 3rd Scientific Conference on Motor Skill Acquisition*. Lohja, FI.
- Roberts, Alexandra H. (2016). *Effects of sport-specific training conditions on performance in high school field hockey players*. University of Louisville.
- Roberts, Alexandra H, Walden, A. J., Carter, K. A., & Symons, T. B. (2019c). Effect of Sport-
Specific Constraints on Aerobic Capacity in High School Field Hockey Players. Journal of
Strength and Conditioning Research.
https://www.doi.org/10.1519/jsc.00000000003471
- Roberts, S. J., McRobert, A. P., Lewis, C. J., & Reeves, M. J. (2019). Establishing consensus of position-specific predictors for elite youth soccer in England. *Science and Medicine in Football*, 3(3), 205–213. <u>https://doi.org/10.1080/24733938.2019.1581369</u>
- Robertson, S., Woods, C., & Gastin, P. (2015). Predicting higher selection in elite junior Australian Rules football: The influence of physical performance and anthropometric attributes. *Journal of Science and Medicine in Sport*, *18*(5), 601–606. https://doi.org/10.1016/j.jsams.2014.07.019
- Rollin, J., Alegi, P. C., Weil, E., Guilianotti, R. C., & Joy, B. (2018). Football Soccer. Retrieved
 November 10, 2018, from Encyclopaedia Brittanica website: https://www.britannica.com/sports/football-soccer
- Romann, M., Javet, M., & Fuchslocher, J. (2017). Coaches' eye as a valid method to assess biological maturation in youth elite soccer. *Talent Development & Excellence*, 9(1), 3– 13.

- Romeas, T., Guldner, A., & Faubert, J. (2016). 3D-Multiple Object Tracking training task improves passing decision-making accuracy in soccer players. *Psychology of Sport and Exercise*, 22, 1–9. https://doi.org/10.1016/j.psychsport.2015.06.002
- Rosciano, A., Lindell, D., Bryer, J., & DiMarco, M. (2016). Nurse practitioners' use of intuition. *Journal for Nurse Practitioners, 12*(8), 560–565. https://doi.org/10.1016/j.nurpra.2016.06.007
- Rosevear, R., & Cassidy, T. (2019). The role of character in talent identification and development in New Zealand rugby union. *International Journal of Sports Science & Coaching*, 14(3), 406–418. https://doi.org/10.1177/1747954119847172
- Rothwell, M., Davids, K., Stone, J. A., & Sullivan, M. O. (2020). A department of methodology can coordinate transdisciplinary sport science support. *Journal of Expertise*, *3*(1), 55– 65.
- Rowat, O., Fenner, J., & Unnithan, V. (2017). Technical and physical determinants of soccer match-play performance in elite youth soccer players. *The Journal of Sports Medicine* and Physical Fitness, 57(4), 369–379. https://doi.org/10.23736/S0022-4707.16.06093-X
- Russell, S., Renshaw, I., & Davids, K. (2019). How interacting constraints shape emergent decision-making of national-level football referees. *Qualitative Research in Sport, Exercise and Health*, 11(4), 579–588. https://doi.org/10.1080/2159676X.2018.1493525
- Rynne, S. B., Crudgington, B., Dickinson, R. K., & Mallett, C. J. (2017). On the (potential) value of coaching. In J. Baker, S. Cobley, J. Schorer, & N. Wattie (Eds.), *Routledge Handbook of Talent Identification and Development in Sport* (pp. 285–300). Routledge.
- Sandelowski, M. (2000). Whatever happened to qualitative description? *Research in Nursing* & *Health*, *23*(4), 334–340. https://doi.org/10.1002/1098-240X(200008)23:4<334::AID-NUR9>3.0.CO;2-G
- Sarmento, H., Anguera, M. T., Pereira, A., & Araújo, D. (2018). Talent identification and development in male football: A systematic review. *Sports Medicine*, 48(4), 907–931. https://doi.org/10.1007/s40279-017-0851-7
- Schempp, P. G., & McCullick, B. (2010). Coaches' expertise. In J. Lyle & C. Cushion (Eds.), Sports Coaching: Professionalisation and Practice (pp. 221–232). Elsevier.

- Schorer, J., Busch, D., Fischer, L., Pabst, J., Rienhoff, R., Sichelschmidt, P., & Strauß, B. (2012).
 Back to the future: A case of the ongoing evaluation of the German handball talent selection and development system. In J. Baker, S. Cobley, & J. Schorer (Eds.), *Talent identification and development in sport: International perspectives*. Routledge.
- Schorer, J., Faber, I., Koopmann, T., Büsch, D., & Baker, J. (2020). Predictive value of coaches' early technical and tactical notational analyses on long-term success of female handball players. *Journal of Sports Sciences*, *00*(00), 1–7. https://doi.org/10.1080/02640414.2020.1776923
- Schorer, J., Rienhoff, R., Fischer, L., & Baker, J. (2017). Long-term prognostic validity of talent selections: Comparing national and regional coaches, laypersons and novices. *Frontiers in Psychology*, 8: 1146. https://doi.org/10.3389/fpsyg.2017.01146
- Seifert, L., Komar, J., Araújo, D., & Davids, K. (2016). Neurobiological degeneracy: A key property for functional adaptations of perception and action to constraints. *Neuroscience and Biobehavioral Reviews, 69,* 159–165. https://doi.org/10.1016/j.neubiorev.2016.08.006
- Shanteau, J., Weiss, D. D. J., Thomas, R. R. P., & Pounds, J. C. (2002). Performance-based assessment of expertise: How to decide if someone is an expert or not. *European Journal of Operational Research*, 136(2), 253–263. https://doi.org/10.1016/S0377-2217(01)00113-8
- Sheldon, W., Stevens, S. S., & Tucker, W. B. (1940). *The varieties of human physique*. Harper Bros.
- Sieghartsleitner, R., Zuber, C., Zibung, M., & Conzelmann, A. (2019). Science or coaches' eye?
 Both! Beneficial collaboration of multidimensional measurements and coach assessments for efficient talent selection in elite youth football. *Journal of Sports Science and Medicine*, 18(1), 32–43. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/30787649
- Silver, N. (2015). *The Signal and the Noise*. New York, NY: Penguin Random House.
- Simon, H. A. (1992). What Is an Explanation of Behavior? *Psychological Science*, *3*(3), 150–161. <u>https://doi.org/10.1111/j.1467-9280.1992.tb00017.x</u>

- Smith, B., & McGannon, K. R. (2018). Developing rigor in qualitative research: problems and opportunities within sport and exercise psychology. *International Review of Sport and Exercise Psychology*, 11(1), 101–121. https://doi.org/10.1080/1750984X.2017.1317357
- Smith, B., & Sparkes, A. C. (2006). Narrative inquiry in psychology: Exploring the tensions within. Qualitative Research in Psychology, 3(3), 169–192. https://doi.org/10.1191/1478088706qrp068oa
- Sport Australia. (2020a). *AusPlay survey results January 2019 December 2019: Table 14: Participation by activity (adults)*. Retrieved from https://www.clearinghouseforsport.gov.au/__data/assets/excel_doc/0004/842791/A usPlay_National_data_tables_30_April_2020.xlsx
- Sport Australia. (2020b). *AusPlay survey results January 2019 December 2019: Table 15: Organised participation by activity (children)*. Retrieved from https://www.clearinghouseforsport.gov.au/__data/assets/excel_doc/0004/842791/A usPlay_National_data_tables_30_April_2020.xlsx
- Stake, R. E. (2003). Case studies. In N. K. Denzin & Y. S. Lincoln (Eds.), *Strategies of Qualitative Enquiry* (2nd ed., pp. 134–164). Thousand Oaks, CA: Sage.
- Stambulova, N., Alfermann, D., Statler, T., & Côté, J. (2009). ISSP Position stand: Career development and transitions of athletes. *International Journal of Sport and Exercise Psychology*, 7(4), 395–412. https://doi.org/10.1080/1612197X.2009.9671916
- Stamm, R., Stamm, M., & Thomson, K. (2005). Role of adolescent female volleyball players' psychophysiological properties and body build in performance of different elements of the game. *Perceptual and Motor Skills*, 101(1), 108–120.
- Stanley, M. (2014). Qualitative descriptive: A very good place to start. In S. Nayar & M. Stanley (Eds.), *Qualitative research methodologies for occupational science and therapy* (pp. 37–52). Routledge.
- Stewart, T. R., Roebber, P. J., & Bosart, L. F. (1997). The importance of the task in analyzing expert judgment. Organizational Behavior and Human Decision Processes, 69(3), 205– 219. https://doi.org/10.1006/obhd.1997.2682

Tanner, J. M. (1964). The physique of the Olympic athlete. London: Allen and Unwin.

- Teramoto, M., Cross, C. L., & Willick, S. E. (2016). Predictive Value of National Football League Scouting Combine on Future Performance of Running Backs and Wide Receivers. *Journal of Strength and Conditioning Research*, 30(5), 1379–1390. https://doi.org/10.1519/JSC.000000000001202
- The Tokyo Organising Committee of the Olympic and Paralympic Games. (2019). Tokyo 2020. Retrieved January 28, 2019, from https://tokyo2020.org/en/games/sport/
- Thelwell, R. C., Weston, N. J. V. V., Greenlees, I. A., & Hutchings, N. V. (2008). Stressors in elite sport: A coach perspective. *Journal of Sports Sciences*, 26(9), 905–918. https://doi.org/10.1080/02640410801885933
- Thorpe, H., & Olive, R. (2019). Conducting observations in sport and exercise. In *Routledge* Handbook of Qualitative Research in Sport and Exercise (pp. 124–138). Routledge.
- Tichy, N., & Bennis, W. (2007). *Judgment: How winning leaders make great calls*. Penguin Random House.
- Till, K., & Baker, J. (2020). Challenges and [Possible] Solutions to Optimizing Talent Identification and Development in Sport. *Frontiers in Psychology*, 11(April), 1–14. https://doi.org/10.3389/fpsyg.2020.00664
- Till, K., Muir, B., Abraham, A., Piggott, D., & Tee, J. (2019). A framework for decision-making within strength and conditioning coaching. *Strength and Conditioning Journal*, 41(1), 14–26. https://doi.org/10.1519/SSC.0000000000000408
- Tinker, C., & Armstrong, N. (2008). From the outside looking in: how an awareness of difference can benefit the qualitative research process. *The Qualitative Report*, 13(1), 53–60. Retrieved from http://www.nova.edu/ssss/QR/QR13-1/
- Tranckle, P., & Cushion, C. J. (2006). Rethinking giftedness and talent in sport. *Quest*, *58*(2), 265–282. https://doi.org/10.1080/00336297.2006.10491883
- Tromp, E. J. Y., Pepping, G.-J., Lyons, J., Elferink-Gemser, M. T., & Visscher, C. (2013). "Let's pick him!": Ratings of skill level on the basis of in-game playing behaviour in bantam league junior ice hockey. *International Journal of Sports Science & Coaching*, 8(4), 641–660. https://doi.org/10.1260/1747-9541.8.4.641
- Trottier, K. (2017). Leading under pressure: Evaluating the decision-making style of NHL coaches. *Sport, Business and Management: An International Journal, 7*(1), 97–112. https://doi.org/10.1108/SBM-11-2014-0046

- Unnithan, V., White, J., Georgiou, A., Iga, J., & Drust, B. (2012). Talent identification in youth soccer. *Journal of Sports Sciences*, *30*(15), 1719–1726. https://doi.org/10.1080/02640414.2012.731515
- Vaeyens, R., Lenoir, M., Williams, A. M., & Philippaerts, R. M. (2008). Talent identification and development programmes in sport: Current models and future directions. *Sports Medicine*, 38(9), 703–714. https://doi.org/10.2165/00007256-200838090-00001
- Vaismoradi, M., Turunen, H., & Bondas, T. (2013). Content analysis and thematic analysis: Implications for conducting a qualitative descriptive study. *Nursing and Health Sciences*, 15(3), 398–405. https://doi.org/10.1111/nhs.12048
- Van Yperen, N. W. N. W. (2009). Why some make it and others do not: Identifying psychological factors that predict career success in professional adult soccer. *The Sport Psychologist*, 23(3), 317–329. https://doi.org/10.1123/tsp.23.3.317
- Vilar, L., Araújo, D., Davids, K., Correia, V., & Esteves, P. T. (2013). Spatial-temporal constraints on decision-making during shooting performance in the team sport of futsal. *Journal of Sports Sciences*, 31(8), 840–846. https://doi.org/10.1080/02640414.2012.753155
- Vilar, L., Araújo, D., Davids, K., & Renshaw, I. (2012). The need for 'representative task design' in evaluating efficacy of skills tests in sport: A comment on Russell , Benton and Kingsley (2010). Journal of Sports Sciences, 30(16), 1727–1730. https://doi.org/10.1080/02640414.2012.679674
- Vrljic, K., & Mallett, C. J. (2008). Coaching knowledge in identifying football talent. International Journal of Coaching Science, 2(January), 1–34.
- Washburn, J. N. (1956). Sport as a Soviet tool. *Foreign Affairs*, 34(3), 490–499. https://doi.org/10.1007/978-1-349-95321-9_146
- Waters, A., Phillips, E., Panchuk, D., & Dawson, A. (2019). Coach and biomechanist experiential knowledge of maximum velocity sprinting technique. *International Sport Coaching Journal*.
- Weber, E. U., Ames, D. R., & Blais, A. R. (2005). "How do I choose thee? Let me count the ways": A textual analysis of similarities and differences in modes of decision-making in China and the United States. *Management and Organization Review*, 1(1), 87–118. https://doi.org/10.1111/j.1740-8784.2004.00005.x

- Weber, E. U., & Johnson, E. J. (2009). Mindful judgment and decision making. *Annual Review of Psychology*, *60*, 53–85. https://doi.org/10.1146/annurev.psych.60.110707.163633
- Western Australian Institute of Sport. (2019). Test your potential at the WAIS Talent Searchday.RetrievedDecember19,2019,fromhttp://wais.org.au/other/news_detail.php?id=10110
- Wharton, L., & Rossi, T. (2015). How would you recognise an expert coach if you saw one? *International Journal of Sports Science & Coaching*, 10(2–3), 577–588. https://doi.org/10.1260/1747-9541.10.2-3.577
- Wheaton, B. (2002). Babes on the beach, women in the surf: Researching gender, power and differences in the windsurfing culture. In J. Sugden & A. Tomlinson (Eds.), *Power games:* A critical sociology of sport (pp. 240–266). London and New York: Routledge.
- Williams, A. M., Ford, P. R., & Drust, B. (2020). Talent identification and development in soccer since the millennium. *Journal of Sports Sciences*, 00(00), 1–12. https://doi.org/10.1080/02640414.2020.1766647
- Williams, A. M., & Reilly, T. (2000). Talent identification and development in soccer. *Journal of Sports Sciences*, *18*(9), 657–667. https://doi.org/10.1080/02640410050120041
- Willmott, T., & Collins, D. (2017). A retrospective analysis of trick progression in elite freeskiing and snowboarding. *International Sport Coaching Journal*, 4(2014), 1–12.
- Winter, S., Brien, F. O., & Collins, D. (2019). Things ain't what they used to be? Coaches perceptions of commitment in developing athletes. *Journal of Applied Sport Psychology*. https://doi.org/10.1080/10413200.2019.1646839
- Wiseman, A. C., Bracken, N., Horton, S., & Weir, P. L. (2014). The difficulty of talent identification: Inconsistency among coaches through skill-based assessment of youth hockey players. *International Journal of Sports Science & Coaching*, 9(3), 447–455. https://doi.org/10.1260/1747-9541.9.3.447
- Woods, T. C., Keller, B. S., McKeown, I., & Robertson, S. (2016). A comparison of athletic movement between talent identified juniors from different football codes in Australia: Implications for talent development. *Journal of Strength and Conditioning Research*, 30(9), 2440–2445. https://doi.org/10.1519/JSC.00000000001354

- Woods, T. E. C., Raynor, A. J., Bruce, L., & McDonald, Z. (2015). The use of skill tests to predict status in junior Australian football. *Journal of Sports Sciences*, 33(11), 1132–1140. https://doi.org/10.1080/02640414.2014.986501
- Wright, G., & Bolger, F. (Eds.). (1992). *Expertise and decision support*. Plenum Press.
- Wu, H., & Leung, S. O. (2017). Can Likert Scales be Treated as Interval Scales?—A Simulation
 Study. Journal of Social Service Research, 43(4), 527–532.
 https://doi.org/10.1080/01488376.2017.1329775
- Wylleman, P., De Knop, P., & Reints, A. (2011). Transitions in competitive sport. In N. L. Holt
 & M. Talbot (Eds.), *Lifelong engagement in sport and physical activity: Participation across the lifespan*. Routledge.

Yin, R. K. (2018). *Case study research and applications: Design and methods* (6th ed.). Sage Publications.

Appendices

Appendix A

Characteristics of Included Studies

Reference (Region)	N	Sport	Sex	Age	Coach level	Years' Experience	Coach Education	Athlete age/level	Study Design	Key Findings
Christensen (2009) (Europe)	8	Soccer	Μ	33-64 (mean=45)	National	8-28 (mean = 15)	>Bachelor's Degree	Youth	Interview	 Visual experience and pattern recognition Pre-eminence of hard work and dedication Coach as arbiter of taste
Cupples & O'Connor (2011) (Oceania)	13	Rugby League	NR	NR	State / National	>10	NR	NR	Delphi	 Cognitive indicators Game skill Physiological
Ellingsen & Danielsen (2017) (Europe)	8 2 2 2 2	Multi-Sport Soccer Gymnastics Swimming Skiing	7M 1F	NR	District or higher	15-40	BA, MA or HL Diploma	5-15	Interview	 Social characteristics Personal characteristics Bodily characteristics

Reference (Region)	N	Sport	Sex	Age	Coach Level	Years' Experience	Coach Education	Athlete Age/Level	Study Design	Key Findings
Ferreira Celestino <i>et al</i> (2015) (Europe)	10	Orienteering	Μ	mean = 42	National	NR	PE & Sport (7/10)	NR	Interview	 Primary influence factors (self-regulation competencies; cognitive processes; quality of practice; ten years of practice) Secondary influence factors (family; sociocultural aspects; peers; club; sport at school; coach)
Gonçalves <i>et</i> <i>al</i> (2017) (Europe)	14	Basketball	NR	NR	Youth and Men's	>10	Level II or III	NR	Interview	 Environmental Psychological Technical tactical skills Physical attributes Anthropometry
Holt & Dunn (2004) (Europe)	6	Soccer	NR	35-64	Academy	>5	UEFA A license	NR	Interview	 Soccer development system Desired player qualities Training environment Mental aspects
Johansson & Fahlén (2017) (Europe)	14 8 6	Multi-Sport Soccer Alpine Skiing	NR	NR	International	NR	NR	NR	Interview	 Idea of selections Criteria for selections Selection process Outcomes and consequences of selections

Reference	N	Sport	Sex	Age	Coach Level	Years'	Coach	Athlete	Study	Key Findings
Johnson et al (2009) (North America)	6	swimming	М	48-57	International	>20	Level 5 American Swimming Coaches Association	NR	Interview	 Intrapersonal Interpersonal Lifestyle Training Environment Systemic interaction
Jokuschies <i>et</i> <i>al</i> (2017) (Europe)	5	Soccer	Μ	47-60 (M=55.6)	National	>10	UEFA licence; national soccer diploma	U15-U18	Interview	 Personality Cognitive-perceptual skills Motor abilities Development Technique Social environment Physical constitution Cognitive-perceptual skills/technique Motor abilities/technique Personality/technique Other
Lund & Söderström (2017) (Europe)	15	Soccer	М	NR	District / Regional	NR	NR	U15	Interview	 How districts organize TID How coaches understand and define talent and identification Significance activities for talent development

Reference (Region)	N	Sport	Sex	Age	Coach Level	Years' Experience	Coach Education	Athlete Age/Level	Study Design	Key Findings
Milistetd <i>et</i> <i>al</i> (2013) (South America)	10	Volleyball	NR	45 <u>+</u> 13.8	State / National	24.8 <u>+</u> 12.1	NR	Juniors	Interview	 Stature importance Indicators of detection Indicators of selection Use of detection Methods of selection
Miller <i>et al</i> (2015) (Europe)	6	Soccer	Μ	26 - 62 (mean = 45)	EPP category 1-3	>5	NR	NR	Interview	 Nature vs nurture Psychology Social skills
Mills <i>et al</i> (2012)	10	Soccer	NR	47.5 <u>+</u> 10.5	Premier League	14.5 <u>+</u> 6.2	UEFA Pro / UEFA A license	16-18 years	Interview	 Awareness Resilience Coal directed attributes
(Europe)										 Goal-difected attributes Intelligence Sport-specific attributes Environmental factors
Rosevear & Cassidy (2019) (Oceania)*	1	Rugby Union	Μ	NR	Provincial / National	NR	NR	17-21 years	Interview	- Character
Vrljic & Mallett (2008) (Oceania)	5	Soccer	Μ	42-51 (mean = 46.5)	State	mean = 20.1	Level II - III	Youth	Interview	 Defining 'elements' of talent Importance of identified elements of talent Capacity to evaluate the identified elements of talent Selecting talented players for the state football team

Reference (Region)	N	Sport	Sex	Age	Coach Level	Years' Experience	Coach Education	Athlete Age/Level	Study Design	Key Findings
Winter et al. (2019)	12	Multi-Sport	М	31-58 <u>+</u> 8.76	Academy	13.67 <u>+</u> 8.42	NR	15-18	Interview	- Characteristics of the 'ideal' athlete
(Europe)*	4	Soccer		(mean =						- Balance between
	1	Cricket		41.25)						performance and winning
	2	Rugby League								 Commitment and talent Maturity
	1	Rugby Union								- Drivers
	1	Badminton								- Role models
	1	Judo								- Honesty
	1	Swimming								
	1	Tennis								

NR = Not reported

* Included for thesis update

Appendix B

Specific Questions from Each Article

Citation	Question/s asked
Celestino et. al (2015)	NR
Christensen (2009)	NR
Cupples & O'Connor (2011)	NR
Ellingsen & Danielsen (2017)	NR
Gonçalves et. al (2017)	NR
Holt & Dunn (2004)	 What is the structure and aim of your youth academy? How do you try to develop players? What are the most important qualities a player needs to make it as a professional? What qualities do your current crop of players possess? How do you prepare players for professional demands? What areas do you work on? What are the mental strengths you are looking for in players? What mental qualities are most important for professional players?
Johansson & Fahlén (2017)	 What is the position and responsibility/power of the coach? What is the goal/purpose of the selection? What are the basis for selections? Are selection criteria defined? What abilities/skills are judged/measured? Who has knowledge about selection criteria? What factors are most important to consider, if you have to choose between two similar/equally good athletes? Who is involved in selection? Who has the most power during selection? How long is the selection period? Are some selection situations more difficult? Are there possibilities for appeal? Are there protests/discussions about selections? Are the 'right' athletes selected? Are selections evaluated? Are there any fairness issues? Is there the possibility for athletes to influence selections?
Johnson et. al (2008)	 What do you feel contributes to a swimmer achieving top performance (e.g. World Records) vs excellent performances (e.g., finaling [sic] at NCAAs)?
Citation	Question/s asked
----------------------------	--
Jokuschies et. al (2017)	 Thinking about all of your players, is there any player who has something that it takes to achieve peak performance in adulthood?
Lund & Söderström (2017)	NR
Milistetd et. al (2013)	 What is the importance of height factor? How should be done [sic] the detection of talents? And the selection? Which is the more relevant indicators for each?
Miller et. al (2015)	 In your experience, what is talent in soccer? How do you recognise it? What is a typical TI experience for you? What is it like to be a coach involved in TI at your club?
Mills et. al (2012)	 Can you tell me a little about your coaching background and experience in football? What things do you consider to influence player development? What personal characteristics or qualities do you believe young footballers require in order to make it to the professional level? Who do you consider to play a significant role in the overall development process?
Rosevear & Cassidy (2019)*	 Describe your role as the PDM (player development manager) at this union? Explain what groups you work with? Ages; aims and ambitions; family, club and school background? Identify character traits that you believe have more of a moral focus Of the traits you have just identified which do you deem more important to the talent identification and development process? What has been the biggest challenge in your efforts to develop character is players? How has this been overcome when using character as a criterion in talent selection? How does a player showcase desirable character?

Citation	Question/s asked
Vrljic & Mallett (2008)	 What are the 'elements' of talented football players? Think about the best youth player you have coached. Who was he? What position did he play? When did you coach him? When you think about this player, can you come up with any other elements associated with talent? In your opinion, what 'elements' are most important in a player? Outline how you evaluate these elements What is your purpose for selecting players for the state team? Please describe in as much detail the process of how you select players for the state team
Winter et. al (2019)*	 Could you describe how hard they work Can you give me some examples that demonstrate this What percentage of your squad are like this? Can you talk me through some of the methods you use to encourage commitment with your athletes?
*Included for thesis update	

Appendix C

Information Form – Semi-Structured Coach Interviews



Australian Institute of Sport

1. Introduction

We would like to invite you to take part in our research study, which will investigate the perceptions of expert combat sport coaches regarding talent identification in sport. This study will be carried out by academics from Edith Cowan University, in collaboration with colleagues from the Australian Institute of Sport. This *Information Sheet* tells you about the research project and what you will be asked to do. Knowing what is involved will help you decide if you want to take part in the research. Please take time to read this information sheet carefully. One of our team will explain anything that you do not understand and will answer any questions you may have. Please note that participation in this research is entirely voluntary – if you do not wish to take part, then you do not have to. If you decide you want to take part in the research project, you will be asked to verbally confirm your consent.

daniel.greenwood@ausport.gov.au

2. What is the purpose of the study?

The aim of this study is to investigate the perceptions of coaches regarding talent identification within combat sports and to understand the knowledge and skills required to predict future ability in young athletes. We hope that the information obtained from this study will inform other coaches, sport policymakers and sports science practitioners about talent identification within combat sports, and ways to make talent identification more effective.

3. Why have I been invited to take part in this study?

You have been invited to take part in this study as we believe you fit our inclusion criteria that are: 1) at least 10 years of coaching experience in boxing, judo or taekwondo; 2) current or recent involvement with your respective sport's national governing body; 3) working knowledge of English allowing you to answer interview questions.

4. What does the study involve?

You will be asked to participate in a digitally recorded interview with the principal investigator (Ms Alexandra Roberts). Ms Roberts will ask you to share your knowledge and experiences on the following topics: a) understanding of talent identification and the perceived importance of talent identification; b) current talent identification practices within your sport and the specific knowledge and skills required to accurately predict performance; c) key attributes necessary for long-term performance and how they change over time; and d) commonalities in talent forecasting between sports. The interview is expected to last approximately one hour.

5. Who is organising and funding the research?

This study is being organised and funded by Edith Cowan University in collaboration with the Australian Institute of Sport.

6. How will my confidentiality be protected?

All information relating to this research project will be confidential and will be stored securely. All electronic data will be stored on an external hard drive in password protected files, and all hard-copy data will be stored in a locked cabinet at the Australian Institute of Sport. If you choose to withdraw from the study all data collected up to that point in time will be destroyed.

7. What happens with the results?

The results of this study may be published in reports, journals, conference proceedings and doctoral research theses. Information collected during the project may be used in future work aimed at developing a talent prediction model for combat sports. In any publication or presentation, information will be provided in such a way that you cannot be identified.

8. What happens when the study ends?

After the study is completed, we will analyse the data to determine themes that emerge across coaches. You will be provided with a list of these themes and asked to confirm that the results match your thoughts. If you are interested, you may be provided with a summary of the full results once the research project is completed.

9. What are the potential benefits and/or risks in taking part in this study?

There are no expected benefits for you personally in taking part in this research; however, your voluntary participation would be greatly appreciated. Eventually, it is hoped that this research will improve your talent identification abilities and practices, so that you can better identify athletes with the potential to become world-class combat athletes.

10. Do I have to take part in this study?

It is up to you if you want to take part in this study – participation is voluntary. If you decide to take part and later change your mind, you are free to withdraw at any stage and you do not have to give a reason for your withdrawal. If you chose to withdraw, there will be no changes in your relationship with Edith Cowan University or the Australian Institute of Sport. If you do decide to take part, we will describe the study and review this information sheet with you. You will then be asked to provide recorded verbal consent for both participation in the interview and consent for us to record the interview. If you would like to participate but decline to consent to recording, you will be asked to sign a consent form to show that you have understood the information provided, and your responses will be recorded by hand.

11. Who should I contact if I have any questions or concerns?

If you have further questions please contact Ms Alexandra Roberts who is the chief investigator of this study at Edith Cowan University. Please find her contact details at the start of this information sheet.

12. Ethics approval

This study has been approved by the Edith Cowan University Human Research Ethics Committee. If you have any concerns or complaints about the research project and wish to talk to an independent person, you may contact the University's Senior Research Ethics Advisor (contact details below):

Ms Kim Gifkins Senior Research Ethics Advisor Edith Cowan University 270 Joondalup Drive Joondalup WA 6027 08 6304 2170 research.ethics@ecu.edu.au

Thank you for taking the time to consider this study. If you wish to take part, please sign the attached consent form. This information sheet is for you to keep.

Appendix D

Consent Form – Semi-Structured Coach Interviews

CONSENT FORM – PARTICIPANT

Understanding coaches' ideas of talent identification and forecasting

Chief Investigator:

Ms Alexandra Roberts Chief Investigator School of Medical and Health Sciences Edith Cowan University, Joondalup, Western Australia Ph: +

Associate Investigators:

Associate Professor Annette Raynor (Edith Cowan University), Ms Fiona Iredale (Edith Cowan University), Dr Clare Humberstone (Australian Institute of Sport), Dr Daniel Greenwood (University of Memphis).

Declaration by Participant

- I have been provided with a copy of the Participant Information Sheet explaining the research project.
- I have read and understood this Information Sheet and I understand the purpose and aims of the research project.
- I have been given the opportunity to ask any questions and I have had any questions answered to my satisfaction.
- I am aware that if I have any further questions then I can contact a member of the research team.
- I understand that all information provided and data collected will be strictly confidential and will be stored accordingly, with access given only to people involved in this research project.
- I agree that the research data gathered may be published provided no name or other identifying information is used.
- I freely agree to participate in this research project as described and understand that I am free to withdraw at any time during the project without explanation or prejudice.
- I understand that I will be given a signed copy of this document to keep.

Name of Participant	
Signature of Participant	
Date	

Appendix E

Interview Guide

QUESTION	PROBES	STIMULI	PURPOSE
Please tell me about your	- Which sports?	- Level of comp/results	- To place all future
coaching background –	- Length of time coaching?	- Particular subjects	responses in context
how did you get in to	- What levels/ages?	studied/electives	- Establish current
coaching?	- Formal coaching education?		performance/coaching
When did you decide you	- School/university?		level
wanted to be a coach?			
	Section 1: Understanding	g and importance of TID	
What does the term 'talent	- What are your coaching	- Concepts	- Definition of talent
identification' mean to	philosophies related to talent	- Processes	identification
you?	identification?	- Age groups	
	- What is the 'end goal' of a	- National vs international	
	talent ID program?	- Why do we want to identify	
	- Do you think of talent	talented athletes?	
	identification as an 'immediate'	- What do we do with these	
	(within a year) concept or more	talented athletes?	
	long-term? How far in advance	- Can we identify talented	
	can we realistically identify an	athletes?	
	athlete? Ages?		

QUESTION	PROBES	STIMULI	PURPOSE
	Section 1: Understandi	ing and importance of TID	
You've just talked about	- Is success only about winning?	- Improvement count as	- Definition of athlete
what it means to you can	- Does your definition of success	success?	success
you think of an athlete who	change based on age group /	- National success vs	- Probe for differing
comes to mind?	athlete / experience level?	international success	definitions based on
At what point would this	- How would you define 'success'	- Elite success vs sub elite	age group/experience
identified athlete be	for talent identification?	success	level
'successful'?			- Definition of 'successful
			TID program'
	Section 2: (Current TID Processes	
How do <i>you</i> identify	- Does this change based on	- Current TID processes	- Current use of TID
talented athletes?	age/gender/weight	within country	- Perceived effectiveness
	category/athlete experience?	- Typical athlete pathways	of current TID methods
	- Do you take an athlete's	- Has the process produced	
	progress/improvement into	'successful' athletes that	
	account?	otherwise might have been	
	- What characterizes an athlete	missed?	
	with the potential to become	- Drills	
	elite?	- Performance under pressure	

QUESTION	PROBES	STIMULI	PURPOSE
	Section 3: Attributes necessar	ry for long-term performance	
Think of an athlete that	- Do these change with	- Physical?	- Specific, preferably
you believe has the	age/gender/experience/weight?	- Psychological?	measurable, factors or
potential to succeed long-	- Can they be measured?	- Competition scores?	attributes that coaches
term – what is it about	- Can others be taught to see	- Intuition/gut feeling?	use to predict talent
them that makes you think	these things?	- Socioeconomic/upbringing?	
that? What sets them	- What makes the difference		
apart from other athletes?	between a good (general)		
	athlete and a great [sport]		
	athlete?		
	- Please provide examples		
	- Can you pinpoint what it was		
	that made you think that?		
Of the things we've talked	- Technical – repertoire	- Use examples from	- Ranking order of
about, can you please rank	- Technical - ability	conversation – use to probe	importance
them in order of	- Tactical	for further insight	- Understand fluid nature
importance for predicting	- Mental Toughness/Resilience	- Is that true for everybody all	of TID processes
future talent?	- Psychological skills	the time? Gender, age,	
	- Competition results	experience level, weight	
	- OTHER (gut feeling)	category	
	- Physical	- How flexible is the list?	
about, can you please rank them in order of importance for predicting future talent?	 Technical - ability Tactical Mental Toughness/Resilience Psychological skills Competition results OTHER (gut feeling) Physical 	 conversation – use to probe for further insight Is that true for everybody all the time? Gender, age, experience level, weight category How flexible is the list? 	importanceUnderstand fluid nature of TID processes

QUESTION	PROBES	STIMULI	PURPOSE
	Section 3: Attributes nece	ssary for long-term performance	
What are the 'non- negotiable' aspects of forecasting talent?	 Is there anything that no matter how good they are in other areas, if they don't have x they won't make it? Can these be trained? WHY is it important, WHO has it HOW do you compare it? 	 Height Strength Correct body type Mental toughness 'want to fight' 	- Non-negotiables
What are the "no-go's" of forecasting talent?	 Is there anything that is an absolute deal breaker – if they have this, then it's not worth the time/effort? Can these be trained? 	 Attitude problems (define?) Weight problems (maintaining weight) 	- 'deal breakers'
How much emphasis do you place on intuition/gut feelings/instinct?	 Please provide examples of an athlete who you 'just knew' was going to be good – how were they different from others? 		- Role of coaches' intuition
How long do you need to observe an athlete for in order to identify them as talented?		 Settings (competition vs training) Interaction (coach vs observation) 	- Application of TID

QUESTION	PROBES	STIMULI	PURPOSE
	Section 3: Attributes nec	essary for long-term performance	
Is there a difference when identifying athlete's capacity to perform short- term (within a year) or long-term (five-ten years from now)?	 Why/why not? Do you prioritize one form of identification over the other? 	 Is there a difference or do you select all athletes the same way? Is it possible to predict talent far into the future? 	 Long term vs short term selection/identification methods
	Section 4: Commonali	ties in TID across combat sports	
Are there any talent identification/forecasting procedures that you would like to implement?	 Why/why not? From other coaches or other sports/systems? 	 Other combat sports Racquet sports European Soccer academies NCAA system Similar sporting demands (eg physicality, psychological makeup) 	 Similarities between combat sports Programs that might be useful

Appendix F

Example of Interview Coding

Example meaning unit	Example code	Example category	Theme
Now that I'm older, instinct comes into it a lot more. I trust myself more. Know to take into consideration more things, like the family environment	Time spent coaching / Instinct	Experience	Experiential Knowledge
[Instinct] comes with experience. It comes with the mistakes that you make, and that you recognise the mistakes so you get better, and the more you see the more examples you have	Recognition of examples	Experience	Experiential Knowledge
They do amazing work and you rely on sparring and drills, but in the end what counts is to have the proof in the realistic situation – the competition	Observations	Different scenarios	Context
You see, gut instinct is something that's [developed] over a period of time	Instinct	Takes time	Temporal Factors
He has to fight a certain type of fight, because of his size. He's small so you have to give him the technical ability and tactics to be able to fight that distance	Athlete constraints	Compensation	Experiential Knowledge

Appendix G

Information Form – Coach Case Study

INFORMATION TO PARTICIPANTS

Research Title: Talent Identification by an Elite Taekwondo Coach: An Applied Case Study

Principal Researcher:

Alexandra Roberts; Alexandra.roberts@ausport.gov.au

We would like to invite you to participate in this original research project. You should only participate if you want to; choosing not to take part will not disadvantage you in any way. Before you decide whether you want to take part, it is important for you to understand why the research is being done and what your participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

Aim:

The aim of this research project is to understand how an elite taekwondo coach develops his opinions of athletes' athletic talent over the course of a five-day camp.

Benefits:

This study will provide a baseline for future research into the use of coach judgements in talent identification.

What is involved?

Athletes: During an upcoming five-day Australian Taekwondo camp, the principal researcher (Alex Roberts) will observe your training and testing sessions and take notes about the content of these sessions. No activity outside of that required by the camp will be requested of you.

Coach: At the end of each day, Alex will interview you about your experiences from the day and your opinions regarding the athletes' potential. Twice during the camp (day 3 and day 5) you will be asked to group the athletes according to your current level of perception of their talent. You will also be asked to participated in a brief pre- and post-camp interview on the same topics. The total time of your involvement in this project is anticipated to be 5 hours.

Who we are recruiting?

We are recruiting elite, national level taekwondo coaches within Australia

Confidentiality:

All data will be kept confidential and stored on password-protected computers. It will only be seen by members of the research team and used for academic research. If published in an article or report, or presented at a conference, all identifying information will be removed. The coach will be described as 'Head Coach', and country of origin and previous employment will not be stated. You will receive a report of the results of this study six months after completion.

Ethics Approval:

This study has been approved by the Australian Institute of Sport ethics committee. If you have any concerns, you may contact the secretary of the AIS Ethics Committee on 02 6214 1577.

Further information:

For further information on any aspect of participating in this study, please contact the principal researcher (Alex Roberts).

Appendix H

Consent Form – Coach Case Study

<u>'INFORMED CONSENT' FORM (Adult) - Coach</u>

Project Title: Talent Identification by an Elite Taekwondo Coach: An Applied Case Study

Principal Researchers: Alexandra Roberts

This is to certify that I, hereby agree to participate as a volunteer in a scientific investigation as an authorised part of the research program of the Australian Sports Commission under the supervision of Alexandra Roberts.

The investigation and my part in the investigation have been defined and fully explained to me by Alexandra Roberts and I understand the explanation. A copy of the procedures of this investigation and a description of any risks and discomforts has been provided to me and has been discussed in detail with me.

- I understand that I am consenting to the use of previously collected data and possibly identifiable data
- I understand that I will be described as "Head Coach", and that my country of origin and previous employment will not be stated in any published materials.
- I have been given an opportunity to ask whatever questions I may have had and all such questions and inquiries have been answered to my satisfaction.
- I understand that I am free to deny any answers to specific items or questions in interviews or questionnaires.
- I understand that I am free to withdraw consent and to discontinue participation in the project or activity at any time, without disadvantage to myself.
- I understand that I am free to withdraw my data from analysis without disadvantage to myself.
- I certify to the best of my knowledge and belief, I have no physical or mental illness or weakness that would increase the risk to me of participating in this investigation.
- I am participating in this project of my (his/her) own free will and I have not been coerced in any way to participate.
- I have read and understand the product and policy information provided to me on surrounding the use of supplements/medications within the study (where applicable)

Privacy Statement: The information submitted will be managed in accordance with the ASC Privacy Policy.

□ I consent to the ASC keeping my personal information.

Signature of Subject: _____

Date: ___/___/___

I, the undersigned, was present when the study was explained to the subject/s in detail and to the best of my knowledge and belief it was understood.

Signature of Researcher: _____

Date:		//	/
-------	--	----	---

Appendix I

Information Form – Coach Reliability



Participant Information Sheet

Understanding coaches' ideas of talent identification and forecasting

Contact details and affiliations of researchers

Ms Alexandra Roberts	Ph: + or	
Associate Professor Annette Raynor	Ph: +61 8 6304 2771 or <u>a.raynor@ecu.edu.au</u>	
Ms Fiona Iredale	Ph: +61 8 6304 2559 or <u>f.iredale@ecu.edu.au</u>	
School of Medical and Health Scie	nces	
Edith Cowan University, Joondalup, Western Australia		
Dr Clare Humberstone	Ph: +61 2 6214 7343 or	
Australian Institute of Sport	clare.humberstone@ausport.gov.au	
Dr Daniel Greenwood	daniel.greenwood@memphis.edu	
University of Memphis		

1. Introduction

We would like to invite you to take part in our research study, which will investigate the perceptions of expert combat sport coaches regarding talent identification in sport. This study will be carried out by academics from Edith Cowan University, in collaboration with colleagues from the Australian Institute of Sport. This *Information Sheet* tells you about the research project and what you will be asked to do. Knowing what is involved will help you decide if you want to take part in the research. Please take time to read this information sheet carefully. One of our team will explain anything that you do not understand and will answer any questions you may have. Please note that participation in this research is entirely voluntary – if you do not wish to take part, then you do not have to. If you decide you want to take part in the research project, you will be asked to verbally confirm your consent.

2. What is the purpose of the study?

The aim of this study is to investigate how coaches' perceptions of athletic talent change over time, and how well coaches agree on an athlete's potential. We hope that the information obtained from this study will inform other coaches, sport policy-makers and sports science practitioners about talent identification with combat sports, and ways to make talent identification more effective.

3. Why have I been invited to take part in this study?

You have been invited to take part in this study because you are a coach who has been invited to attend a state or national development and/or selection camp for athletes.

4. What does the study involve?

You will be asked to rate athletes involved in the camp based on your subjective opinion of their potential future in the sport. The rating will occur on a scale from 1 to 11 (1 being very little potential, 11 being potential future Olympic medallist), and will occur twice a day for the duration of the camp. It is expected that the ratings will take approximately five minutes per session. The number of athletes you will be asked to rate will be no more than 25.

5. Who is organising and funding the research?

This study is being organised and funded by Edith Cowan University in collaboration with the Australian Institute of Sport.

6. How will my confidentiality be protected?

All information relating to this research project will be confidential and will be stored securely. All electronic data will be stored on an external hard drive in password protected files, and all hard-copy data will be stored in a locked cabinet at the Australian Institute of Sport. If you choose to withdraw from the study all data collected up to that point in time will be destroyed.

7. What happens with the results?

The results of this study may be published in reports, journals, conference proceedings and doctoral research theses. Information collected during the project may be used in future work aimed at developing a talent prediction model for combat sports. In any publication or presentation, information will be provided in such a way that you and the athletes you are rating cannot be identified.

8. What happens when the study ends?

After the study is completed we will analyse the data to determine the consistency of your ratings across the camp, and to examine the agreement between coaches during the course of the camp. If you are interested, you may be provided with a summary of the full results once the research project is completed.

9. What are the potential benefits and/or risks in taking part in this study?

There are no expected benefits for you personally in taking part in this research; however your voluntary participation would be greatly appreciated. Eventually, it is hoped that this research will improve your talent identification abilities and practices, so that you can better identify athletes with the potential to become world-class athletes.

10. Do I have to take part in this study?

It is up to you if you want to take part in this study – participation is voluntary. If you decide to take part and later change your mind, you are free to withdraw at any stage and you do not have to give a reason for your withdrawal. If you chose to withdraw, there will be no changes in your relationship with Edith Cowan University or the Australian Institute of Sport. If you do decide to take part, we will describe the study and review this information sheet with you. You will be asked to sign a written consent form to show that you have understood the information provided.

11. Who should I contact if I have any questions or concerns?

If you have further questions please contact Ms Alexandra Roberts who is the chief investigator of this study at Edith Cowan University. Please find her contact details at the start of this information sheet.

12. Ethics approval

This study has been approved by the Edith Cowan University Human Research Ethics Committee. If you have any concerns or complaints about the research project and wish to talk to an independent person, you may contact the University's Senior Research Ethics Advisor (contact details below):

Ms Kim Gifkins Senior Research Ethics Advisor Edith Cowan University 270 Joondalup Drive Joondalup WA 6027 08 6304 2170 research.ethics@ecu.edu.au

Thank you for taking the time to consider this study. If you wish to take part, please sign the attached consent form. This information sheet is for you to keep.

Appendix J

Consent Form – Coach Reliability

CONSENT FORM – PARTICIPANT

Understanding coaches' ideas of talent identification and forecasting

Chief Investigator:

Ms Alexandra Roberts Chief Investigator School of Medical and Health Sciences Edith Cowan University, Joondalup, Western Australia Ph: ______ or Email: ______

Associate Investigators:

Associate Professor Annette Raynor (Edith Cowan University), Ms Fiona Iredale (Edith Cowan University), Dr Clare Humberstone (Australian Institute of Sport), Dr Daniel Greenwood (University of Memphis).

Declaration by Participant

- I have been provided with a copy of the Participant Information Sheet explaining the research project.
- I have read and understood this Information Sheet and I understand the purpose and aims of the research project.
- I have been given the opportunity to ask any questions and I have had any questions answered to my satisfaction.
- I am aware that if I have any further questions then I can contact a member of the research team.
- I understand that all information provided and data collected will be strictly confidential and will be stored accordingly, with access given only to people involved in this research project.
- I agree that the research data gathered may be published provided no name or other identifying information is used.
- I freely agree to participate in this research project as described and understand that I am free to withdraw at any time during the project without explanation or prejudice.
- I understand that I will be given a signed copy of this document to keep.

Name of Participant	
Signature of Participant	
Date	